

Picloram -

Revised Human Health and Ecological Risk Assessment – Final Report

Prepared for:

USDA, Forest Service

Forest Health Protection GSA Contract No. **GS-10F-0082F**

USDA Forest Service BPA: WO-01-3187-0150 USDA Purchase Order No.: 43-1387-3-0708 Task No. 16

Submitted to:

Dave Thomas, COTR Forest Health Protection Staff USDA Forest Service Rosslyn Plaza Building C, Room 7129C 1601 North Kent Street Arlington, VA 22209



Prepared by Patrick R. Durkin¹ and Mark Follansbee²

Submitted by: ¹ Syracuse Environmental Research Associates, Inc. 5100 Highbridge St., 42C Fayetteville, New York 13066-0950 Telephone: (315) 637-9560 Fax: (315) 637-0445 E-Mail: <u>SERA_INC@msn.com</u> Home Page: www.sera-inc.com

June 30, 2003

² Syracuse Research Corporation, 301 Plainfield Road, Suite 350, Syracuse, New York 13212



TABLE OF CONTENTS

TABLE OF CONTENTS (continued)

3.2. EXPOSURE ASSESSMENT	. 3-7
3.2.1. Overview	. 3-7
3.2.2. Workers	. 3-8
3.2.2.1. General Exposures	. 3-9
3.2.2.2. Accidental Exposures	3-11
3.2.3. General Public	3-12
3.2.3.1. General Considerations	3-12
3.2.3.2. Direct Spray	3-13
3.2.3.3. Dermal Exposure from Contaminated Vegetation	3-14
3.2.3.4. Contaminated Water	3-14
3.2.3.4.1. ACUTE EXPOSURE	3-14
3.2.3.4.2. LONGER-TERM EXPOSURE	3-14
3.2.3.5. Oral Exposure from Contaminated Fish	3-18
3.2.3.6. Oral Exposure from Contaminated Vegetation	3-18
3.2.4. Hexachlorobenzene	3-19
3.2.4.1. Dermal Absorption	3-19
3.2.4.2. Acute Exposures	3-21
3.2.4.3. General Considerations for Chronic Exposures	3-22
3.2.4.4. Chronic Exposures Involving Contaminated Vegetation	3-23
3.2.4.5. Chronic Exposures Involving Contaminated Water	3-25
3.2.4.6. Chronic Exposures Involving the Consumption of Contaminated F	`i 3 h27
3.3. DOSE-RESPONSE ASSESSMENT	3-28
3.3.1. Overview	3-28
3.3.2. Existing Guidelines for Picloram	3-28
3.3.3. Existing Guidelines for Hexachlorobenzene.	3-29
3.3.3.1. Systemic Toxicity	3-29
3.3.3.2. Carcinogenic Potency	3-29
3.4. RISK CHARACTERIZATION	3-30
3.4.1. Overview	3-30
3.4.2. Workers	3-31
3.4.3. General Public	3-32
3.4.4. Sensitive Subgroups	3-34
3.4.5. Connected Actions.	3-34
3.4.6. Cumulative Effects.	3-35
3.4.7. Hexachlorobenzene	3-35
3.4.7.1. Workers	3-35
3.4.7.2. General Public	3-36

TABLE OF CONTENTS (continued)

4.	ECOLOGICAL RISK ASSESSMENT	4-1
	4.1. HAZARD IDENTIFICATION	4-1
	4.1.1. Overview	4-1
	4.1.2. Toxicity to Terrestrial Organisms	4-2
	4.1.2.1. Mammals	4-2
	4.1.2.2. Birds	4-2
	4.1.2.3. Terrestrial Invertebrates	4-2
	4.1.2.4. Terrestrial Plants (Macrophytes)	4-3
	4.1.2.5. Terrestrial Microorganisms	4-4
	4.1.3. Aquatic Organisms.	4-4
	4.1.3.1. Fish	4-4
	4.1.3.2. Amphibians	4-6
	4.1.3.3. Aquatic Invertebrates	4-6
	4.1.3.4. Aquatic Plants	4-6
	4.2. EXPOSURE ASSESSMENT	4-7
	4.2.1. Overview	4-7
	4.2.2. Terrestrial Animals.	4-8
	4.2.2.1. Direct Spray	4-8
	4.2.2.2. Indirect Contact	4-11
	4.2.2.3. Ingestion of Contaminated Vegetation or Prey	4-11
	4.2.2.4. Ingestion of Contaminated Water	4-12
	4.2.3. Terrestrial Plants	4-12
	4.2.3.1. Direct Spray	4-13
	4.2.3.2. Off-Site Drift	4-13
	4.2.3.3. Soil Contamination	4-14
	4.2.3.4. Contaminated Irrigation Water	4-16
	4.2.3.5. Wind Erosion	4-17
	4.2.4. Aquatic Organisms.	4-18
	4.3. DOSE-RESPONSE ASSESSMENT	4-18
	4.3.1. Overview	4-18
	4.3.2. Toxicity to Terrestrial Organisms.	4-18
	4.3.2.1. Mammals	4-18
	4.3.2.2. Birds	4-19
	4.3.2.3. Terrestrial Invertebrates	4-19
	4.3.2.4. Terrestrial Plants (Macrophytes)	4-19
	4.3.2.5. Terrestrial Microorganisms	4-19
	4.3.3. Aquatic Organisms	4-19
	4.3.3.1. Animals	4-20
	4.3.3.2. Aquatic Plants	4-20
	4.4. KISK CHARACTERIZATION	4-20
	4.4.1. Overview	4-20
	4.4.2. Terrestrial Organisms	4-21

TABLE OF CONTENTS (continued)

4.4.2.1. Terrestrial Animals 4.4.2.2. Terrestrial Plants	4-21 4-23
4.4.3. Aquatic Organisms.	4-25
5. REFERENCES	4-25

LIST OF TABLES

Table 2-1: Identification and Physical/Chemical Properties of Picloram and the Potassium Salt of Picloram	Tables-1
Table 2-2:Use of picloram by USDA Forest Service in 2001 by Type of Use	Tables-2
Table 2-3: Use of picloram by USDA Forest Service in 2001 by Region	Tables-3
Table 3-1: Concentrations of picloram in streams after the application of known amounts	Tables-4
Table 3-2: Chemical and site parameters used in GLEAMS Modeling for picloram	Tables-5
Table 3-3: Summary of modeled concentrations of picloram in streams	Tables-6
Table 3-4: Summary of modeled concentrations of picloram in ponds	Tables-7
Table 3-5: Chemical and site parameters used in GLEAMS Modeling for hexachlorobenzene	Tables-8
Table 3-6: Summary of modeled concentrations of hexachlorobenzene in soil	Tables-9
Table 3-7: Summary of modeled concentrations of hexachlorobenzene in streams	Гables-10
Table 3-8: Summary of modeled concentrations of hexachlorobenzene in a pond	Гables-11
Table 4-1: Summary of key studies on the toxicity of picloram to fish	Fables-12
Table 4-2: Summary of modeled concentrations of picloram in soil	Fables-13

LIST OF FIGURES

Figure 2-1: Use of picloram by the USDA Forest Service in various regions	
of the United States	Figures-1
	-
Figure 2-2: Agricultural use of picloram in the United States for 1992	Figures-2

LIST OF WORKSHEETS

- Supplement 1: Picloram -WordPerfect Worksheets for Human Health and Ecological Risk Assessments, SERA SERA WPWS 03-43-16-01a, Version 2.04b, June 30, 2003.
- Supplement 2: Hexachlorobenzene -WordPerfect Worksheets for Human Health and Ecological Risk Assessments, SERA WPWS 03-43-16-02a, Version 2.04, April 25, 2003.

ACRONYMS, ABBREVIATIONS, AND SYMBOLS

4A-TCP	2,3,5-trichloro-4-aminopyridine
ACGIH	American Conference of Governmental Industrial Hygienists
a.e.	acid equivalents
AEL	adverse-effect level
a.i.	active ingredient
ARS	Agricultural Research Service
ATSDR	Agency for Toxic Substances and Disease Registry
BCF	bioconcentration factor
bw	body weight
CBI	confidential business information
CI	confidence interval
cm	centimeter
CNS	central nervous system
DAA	days after application
DAT	days after treatment
d.f.	degrees of freedom
EC _x	concentration causing X% inhibition of a process
EC_{25}	concentration causing 25% inhibition of a process
EC_{50}	concentration causing 50% inhibition of a process
ExToxNet	Extension Toxicology Network
F	female
FH	Forest Health
FIFRA	Federal Insecticide, Fungicide and Rodenticide Act
FQPA	Food Quality Protection Act
g	gram
GLEAMS	Groundwater Loading Effects of Agricultural Management Systems
ha	hectare
HQ	hazard quotient
IARC	International Agency for Research on Cancer
IRIS	Integrated Risk Information System
k _a	absorption coefficient
k _e	elimination coefficient
kg	kilogram
K _{o/c}	organic carbon partition coefficient
K _{o/w}	octanol-water partition coefficient
K _p	skin permeability coefficient
L	liter
lb	pound
LC ₅₀	lethal concentration, 50% kill
LD ₅₀	lethal dose, 50% kill
LOAEL	lowest-observed-adverse-effect level
m	meter

ACRONYMS, ABBREVIATIONS, AND SYMBOLS (continued)

М	male
MMAD	mass median aerodynamic diameter
MCS	multiple chemical sensitivity
mg	milligram
mg/kg/day	milligrams of agent per kilogram of body weight per day
mL	milliliter
mM	millimole
MOS	margin of safety
MRID	Master Record Identification Number
MRL	minimum risk level
MSDS	material safety data sheet
MW	molecular weight
NAWQA	National Water Quality Assessment
NCI	National Cancer Institute
NIOSH	National Institute for Occupational Safety and Health
NOAEL	no-observed-adverse-effect level
NOEC	no-observed-effect concentration
NOEL	no-observed-effect level
NOS	not otherwise specified
NRC	National Research Council
NTP	National Toxicology Program
OM	organic matter
OPP	Office of Pesticide Programs
OPPTS	Office of Pesticide Planning and Toxic Substances
OSHA	Occupational Safety and Health Administration
PCBs	polychlorinated biphenyls
ppm	parts per million
RBC	red blood cells
RED	re-registration eligibility decision
RfD	reference dose
SARA	Superfund Amendments and Reauthorization Act
SERA	Syracuse Environmental Research Associates
SRC	Syracuse Research Corporation
UF	uncertainty factor
U.S.	United States
USDA	U.S. Department of Agriculture
U.S. EPA	U.S. Environmental Protection Agency
USGS	U.S. Geological Survey
WHO	World Health Organization
μ	micron
•	greater than
≥	greater than or equal to

ACRONYMS, ABBREVIATIONS, AND SYMBOLS (continued)

- less than or equal to \leq
- equal to =
- approximately equal to approximately \simeq
- ~

To convert	Into	Multiply by
acres	hectares (ha)	0.4047
acres	square meters (m ²)	4,047
atmospheres	millimeters of mercury	760
centigrade	Fahrenheit	1.8 °C+32
centimeters	inches	0.3937
cubic meters (m ³)	liters (L)	1,000
Fahrenheit	centigrade	0.556 °F-17.8
feet per second (ft/sec)	miles/hour (mi/hr)	0.6818
gallons (gal)	liters (L)	3.785
gallons per acre (gal/acre)	liters per hectare (L/ha)	9.34
grams (g)	ounces, (oz)	0.03527
grams (g)	pounds, (oz)	0.002205
hectares (ha)	acres	2.471
inches (in)	centimeters (cm)	2.540
kilograms (kg)	ounces, (oz)	35.274
kilograms (kg)	pounds, (lb)	2.2046
kilograms per hectare (hg/ha)	pounds per acre (lb/acre)	0.892
kilometers (km)	miles (mi)	0.6214
liters (L)	cubic centimeters (cm ³)	1,000
liters (L)	gallons (gal)	0.2642
liters (L)	ounces, fluid (oz)	33.814
miles (mi)	kilometers (km)	1.609
miles per hour (mi/hr)	cm/sec	44.70
milligrams (mg)	ounces (oz)	0.000035
meters (m)	feet	3.281
ounces (oz)	grams (g)	28.3495
ounces per acre (oz/acre)	grams per hectare (g/ha)	70.1
ounces per acre (oz/acre)	kilograms per hectare (kg/ha)	0.0701
ounces fluid	cubic centimeters (cm ³)	29.5735
pounds (lb)	grams (g)	453.6
pounds (lb)	kilograms (kg)	0.4536
pounds per acre (lb/acre)	kilograms per hectare (kg/ha)	1.121
pounds per acre (lb/acre)	mg/square meter (mg/m ²)	112.1
pounds per acre (lb/acre)	μ g/square centimeter (μ g/cm ²)	11.21
pounds per gallon (lb/gal)	grams per liter (g/L)	119.8
square centimeters (cm ²)	square inches (in ²)	0.155
square centimeters (cm ²)	square meters (m ²)	0.0001
square meters (m ²)	square centimeters (cm ²)	10,000
yards	meters	0.9144

COMMON UNIT CONVERSIONS AND ABBREVIATIONS

Note: All references to pounds and ounces refer to avoirdupois weights unless otherwise specified.

Scientific Notation	Decimal Equivalent	Verbal Expression
$1 \cdot 10^{-10}$	0.000000001	One in ten billion
1 · 10 ⁻⁹	0.00000001	One in one billion
1 · 10 ⁻⁸	0.0000001	One in one hundred million
$1 \cdot 10^{-7}$	0.0000001	One in ten million
$1 \cdot 10^{-6}$	0.000001	One in one million
$1 \cdot 10^{-5}$	0.00001	One in one hundred thousand
$1 \cdot 10^{-4}$	0.0001	One in ten thousand
$1 \cdot 10^{-3}$	0.001	One in one thousand
$1 \cdot 10^{-2}$	0.01	One in one hundred
1 · 10 ⁻¹	0.1	One in ten
$1 \cdot 10^{0}$	1	One
$1 \cdot 10^{1}$	10	Ten
$1 \cdot 10^2$	100	One hundred
$1 \cdot 10^{3}$	1,000	One thousand
$1 \cdot 10^{4}$	10,000	Ten thousand
$1 \cdot 10^{5}$	100,000	One hundred thousand
$1 \cdot 10^{6}$	1,000,000	One million
$1 \cdot 10^{7}$	10,000,000	Ten million
$1 \cdot 10^{8}$	100,000,000	One hundred million
$1 \cdot 10^{9}$	1,000,000,000	One billion
$1 \cdot 10^{10}$	10,000,000,000	Ten billion

CONVERSION OF SCIENTIFIC NOTATION

EXECUTIVE SUMMARY

INTRODUCTION

Picloram is a herbicide used in the control a number of broadleaf weeds and undesirable brush. In the preparation of this risk assessment, literature searches of picloram were conducted in the open literature using PubMed, TOXLINE as well as the U.S. EPA CBI files. There is a very large body of literature on the environmental fate and toxicology of picloram and several reviews by U.S. EPA as well as other published reviews were used in the preparation of the current risk assessment. The U.S. EPA re-registration eligibility decision (RED) document also includes a summary of the product chemistry, mammalian toxicology, and ecotoxicology studies that were submitted by industry to the U.S. EPA as part of the registration process for this compound. Full text copies of key studies obtained and reviewed, and synopses of the information that can be disclosed from these studies are included in this document.

Technical grade picloram contains hexachlorobenzene as a contaminant and hexachlorobenzene is classified as a carcinogen. Because of the importance of and level of concern for this endpoint in humans, the human health risk assessment discusses the potential effects of hexachlorobenzene in some detail.

While this document discusses the studies required to support the risk assessments, it makes no attempt to re-summarize all of the information cited in the existing reviews. The Forest Service will update this and other similar risk assessments on a periodic basis and welcomes input from the general public on the selection of studies included in the risk assessment.

PROGRAM DESCRIPTION

Picloram is a herbicide used in the control of a number of broadleaf weeds and undesirable brush. Picloram is used in Forest Service programs almost exclusively for the control of noxious weeds. Very minor uses include rights-of-way management and general maintenance. Tordon K and Tordon 22K are the formulations of picloram currently available and used by the Forest Service. Both formulations are produced by Dow AgroSciences as a liquid containing the potassium salt of picloram (24.4% w/v). This is equivalent to a concentration of 2 lb a.e./gallon. The remaining 75.6% of the formulation consists of inerts, including a polyglycol. The U.S. EPA has placed the polyglycol on List 3 of the inerts that may be used in the formulation of pesticides. Very little additional information is available on this compound.

The most common methods of ground application for Tordon involve backpack (selective foliar) and boom spray (broadcast foliar) operations. The Forest Service does not typically use aerial applications for picloram. Nonetheless, Tordon is registered for aerial applications and aerial applications are included in this risk assessment in the event the Forest Service may wish to consider this application method. The labeled application rates for picloram range from 0.125 to 1 lb a.e./acre. Typically, the Forest Service uses rates in the lower part of this range and some applications may be below the lower range of the labeled rate. For this risk assessment, the typical rate of 0.35 lb a.e./acre with a lower range of 0.1 lb a.e./acre is use to reflect Forest Service practice. An upper range of 1 lb a.e./acre is used to assess the consequences of using the

highest labeled rate should the Forest Service need to consider this option.

HUMAN HEALTH RISK ASSESSMENT

Hazard Identification – The toxicity of picloram to experimental mammals has been very wellcharacterized. Most of the studies have been conducted in support of the registration of picloram and are summarized in the U.S. EPA re-registration eligibility decision document. Picloram has a low order of acute toxicity, with acute oral LD_{50} values in the range of 3000 to 5000 mg/kg body weight. Picloram can cause irritation to the eyes. Although picloram is not a strong skin irritant, repeated dermal exposures may lead to skin sensitization.

In chronic toxicity studies, the most sensitive effect for picloram in mammals involves effects on the liver. The current U.S. EPA RfD is based on a two-year feeding study in male and female Fischer rats in which picloram acid was administered at dietary concentrations that resulted in daily doses of 20, 60, and 200 mg/kg/day. The only statistically significant observations included an increase in liver size and an alteration in the staining properties of centrilobular hepatocytes in the 60 and 200 mg/kg/day dose groups. Dogs appear to be somewhat more sensitive to picloram than rats. In a six month feeding study in which male and female beagle dogs were administered picloram at levels that resulted in average daily doses of 0, 7, 35, and 175 mg/kg/day, the two higher dose levels resulted in increase in absolute and relative liver weight in two males and changes in liver enzyme activity.

Although technical grade picloram has been subject to several chronic bioassays for carcinogenicity and none of the bioassays have shown that picloram has carcinogenic potential, technical grade picloram does contain hexachlorobenzene, a compound that has shown carcinogenic activity in three mammalian species and has been classified as a potential human carcinogen by the U.S. EPA. Thus, this effect is considered both qualitatively and quantitatively in this risk assessment.

Exposure Assessment –Exposure assessments are conducted for both workers and members of the general public for the typical application rate of 0.35 lb/acre. The consequences of using the maximum application rate, 1 lb/acre, are discussed in the risk characterization. For both workers and members of the general public, the upper ranges of all acute exposures are below 1 mg/kg and most exposures are much lower. The highest modeled exposure is about 0.7 mg/kg and is associated with the consumption of contaminated water by a child following an accidental spill of picloram into a small pond. The upper ranges of non-accidental acute exposure scenarios for members of the general public are associated with doses from about 0.00002 to 0.07 mg/kg. The highest dose estimates for non-accidental exposure scenarios are associated with the consumption of contaminated water scenarios are associated with the consumption of contaminated with doses from about 0.00002 to 0.07 mg/kg.

General exposure assessments for workers are in the range of exposures modeled for the general public. For workers, three types of application methods are modeled: directed ground, broadcast ground, and aerial. Central estimates of exposure span a relatively narrow range: 0.005 to 0.008 mg/kg. The upper ranges of exposures are also similar for the different groups of workers: 0.03

to 0.05 mg/kg/day. All of the accidental exposure scenarios for workers involve dermal exposures. Because picloram is not readily absorbed across the skin, all of these accidental exposures lead to estimates of dose that are either in the range of or substantially below the general exposure estimates for workers.

Hexachlorobenzene is a contaminant in technical grade picloram. The average concentration of hexachlorobenzene in technical grade picloram is 8 ppm and the maximum concentration is 50 ppm. For all exposure assessments detailed in this risk assessment, the average concentration of 8 ppm is used. The impact of the 50 ppm level is detailed in the risk characterization. Hexachlorobenzene is ubiquitous and persistent in the environment. The major sources of general exposure for the public to hexachlorobenzene involve industrial emissions, proximity to hazardous waste sites, and the consumption of contaminated food. Virtually all individuals are exposed to hexachlorobenzene and virtually all individuals have detectable concentrations of hexachlorobenzene in their bodies. Based on current concentrations of hexachlorobenzene in environmental media and food, daily doses of hexachlorobenzene (i.e., background levels of exposure) are in the range of 0.000001 (1×10^{-6}) mg/kg/day. Based on the amount of hexachlorobenzene in picloram and the amount of picloram used in Forest Service programs, the use of picloram by the Forest Service will not substantially contribute to any wide-spread increase of ambient levels of hexachlorobenzene. Nonetheless, the potential impact of local contamination is considered for workers as well as for several acute and chronic exposure scenarios for members of the general public. For both workers, the upper range of longer term exposure scenarios result in dose estimates of about 2×10^{-7} mg/kg/day to 4×10^{-7} mg/kg/day, below general background levels of exposure by about a factor of 2 to 5. For members of the general public, the upper range of longer term exposure scenarios are about 1×10^{-10} mg/kg/day to 2×10^{-8} mg/kg/day, below general background levels of exposure by about a factor of 50 to 10,000. The upper range of estimated doses associated with acute exposure scenarios for both workers and members of the general public are about 0.002 mg/kg/day, higher than background levels of exposure by about a factor of 2000.

Dose-Response Assessment – The Office of Pesticide Programs of the U.S. EPA has derived an RfD of 0.2 mg/kg/day for picloram. This RfD is based on a chronic rat NOAEL of 20 mg/kg/day and an uncertainty factor of 100. In the same study, the LOAEL was 60 mg/kg/day and the effect noted was a change in the staining properties of liver cells. No frank signs of toxicity were seen at this or higher dose levels. This NOAEL for chronic toxic effects is below the NOAELs for reproductive effects. Thus, doses at or below the RfD will be below the level of concern for reproductive effects.

The contamination of technical grade picloram with hexachlorobenzene can be quantitatively considered to a limited extent. The U.S. EPA has derived an RfD and cancer potency parameter for hexachlorobenzene. Based on the levels of contamination of technical grade picloram with these compounds and the relative potencies of these compounds to picloram, this contamination is not significant in terms of potential systemic toxic effects. This assessment, however, does not impact the potential carcinogenicity associated with hexachlorobenzene and this risk, based on the U.S. EPA's cancer potency parameter, is quantitatively considered in the risk

characterization.

Risk Characterization – Typical exposures to picloram do not lead to estimated doses that exceed a level of concern. For workers, no exposure scenarios, acute or chronic, exceeds the RfD even at the upper ranges of estimated dose. For members of the general public, the upper limits for hazard quotients are below a level of concern except for the accidental spill of a large amount of picloram into a very small pond. Even this exposure scenario results in only a small excursion above the chronic RfD and is not likely to be toxicologically significant, because of the short duration of exposure relative to those considered in the derivation of the RfD. Thus, based on the available information and under the foreseeable conditions of application, there is no route of exposure or scenario suggesting that workers or members of the general public will be at any substantial risk from longer-term exposure to picloram.

Irritation and damage to the eyes can result from exposure to relatively high levels of picloram (i.e., placement of picloram directly onto the eye) and repeated exposures to picloram can lead to skin sensitization. From a practical perspective, eye irritation and skin sensitization are likely to be the only overt effects as a consequence of mishandling picloram. These effects can be minimized or avoided by prudent industrial hygiene practices during the handling and application of picloram.

Based on the standard assumptions used in this and other Forest Service risk assessments, the contamination of picloram with hexachlorobenzene does not appear to present any substantial cancer risk even at the upper ranges of plausible exposure. Administratively, the Forest Service has adopted a cancer risk level of one in one-million $(1\div1,000,000)$ as a trigger that would require special steps to mitigate exposure or restrict and possibly eliminate use. Based on relatively conservative exposure assumptions and at the typical application rate of 0.35 lb a.e. picloram/acre, the highest cancer risk is about 0.7 in one-million – i.e., for workers involved in broadcast ground spray. At the upper range of the application rate – i.e., 1 lb a.e./acre – this risk would scale to 2 in one-million. This is not, however, an appropriate approach for risk scaling because it would assume that the same worker applies picloram at an atypically high application rate over a lifetime. For members of the general public, the highest cancer risk is estimated at 0.1 in one-million at the typical application rate of 0.35 lb a.e./acre. Scaled to an application rate of 1 lb/acre, the cancer risk would be about 0.3 in one-million.

ECOLOGICAL RISK ASSESSMENT

Hazard Identification – The toxicity of picloram is relatively well characterized in experimental mammals but few wildlife species have been assayed relative to the large number of nontarget species that might be potentially affected by the use of picloram. Within this admittedly substantial reservation, picloram appears to be relatively non-toxic to terrestrial animals but is moderately toxic to aquatic animals, particularly fish.

The assessment of the toxicity of picloram to nontarget terrestrial animals is based almost exclusively on toxicity studies using experimental mammals (i.e., the same studies used in the human health risk assessment). Acute oral LD_{50} values for picloram are in the range of 3000 to

5000 mg/kg body weight and NOAEL from chronic studies range from 7 mg/kg/day to 20 mg/kg/day. Some additional studies are available on birds, bees, and snails that generally support the characterization of picloram as relatively non-toxic to terrestrial animals. This assessment is supported by field studies that reported no detectable effects on mammalian or avian diversity after the application of picloram.

Picloram is a pyridine herbicide that acts as a plant growth regulator. This is to say that picloram mimics naturally occurring plant auxins or hormones in a manner that leads to uncontrolled and abnormal growth. These effects can in turn lead to gross signs of toxicity or death. The toxicity of picloram to terrestrial plants has been assayed in relatively standardized studies of seed emergence, seed germination, and post-emergence applications that have been submitted to the U.S. EPA to support the registration of picloram. Picloram is more toxic to broadleaf plants than grains or grasses. The lowest reported adverse effect (the EC_{25} for the inhibition of seed emergence in soybeans) for the potassium salt of picloram is 0.000014 kg or about 0.000012 lb a.e./acre. The highest reported NOEC in any of the terrestrial plant bioassays is about 0.4 lb a.e./acre for seedling emergence in corn.

The acute and chronic toxicity of picloram to aquatic animals has been assayed in various species of fish and invertebrates. Acute (96-hour) LC_{50} value for trout range from 0.8 mg/L to 19.3 mg/L. Bluegill sunfish and fathead minnows, common test species used in aquatic toxicity studies, appear to be less sensitive to picloram, with LC_{50} values ranging from about 15 mg/L to 55 mg/L. Two sets of longer-term (egg and fry) studies are available in trout and the results of these studies are not consistent. Studies accepted by the U.S. EPA indicate a NOEC of 0.55 mg/L in rainbow trout. An earlier series of studies of using lake trout indicate a NOEC of <0.035 mg/L. Limitations in the this earlier series of studies include the failure to use an acetone control and the failure to measure concentrations in the test solutions. An early life-stage study in the fathead minnow, yielded a NOEC of 0.71 mg a.e./L, very similar to the NOEC reported in trout. This study, however, was conducted on the triisopropanolamine salt of picloram rather than the potassium salt.

In *Daphnia*, an aquatic invertebrate commonly used in aquatic toxicity studies, the reported acute (48-hours) LC_{50} value is 68.3 (63–75) mg/L. Chronic studies in the same species using reproductive or developmental parameters identified a no-effect level at 11.8 mg/L and a lowest effect level at 18.1 mg/L. Thus, it appears that trout are more sensitive than daphnids to both the acute and chronic effects of picloram.

As with aquatic animals, the toxicity of picloram to aquatic plants varies substantially among different species. Based on the available toxicity bioassays, the most sensitive species is *Navicula pelliculosa*, a freshwater diatom, with an EC₅₀ for growth of 0.94 mg a.e./L and a NOEC of 0.23 mg a.e./L. The least sensitive aquatic plants appear to be from the genus *Chlorella* (another group of freshwater algae), with EC₅₀ values greater than 160 mg a.e./L. Macrophytes appear to have a sensitivity that is in the upper range of that seen in algae, with a reported EC₅₀ of 164 mg a.e./L in duckweed.

Exposure Assessment – Terrestrial animals might be exposed to any applied herbicide from direct spray, the ingestion of contaminated media (vegetation, prey species, or water), grooming activities, or indirect contact with contaminated vegetation. In acute exposure scenarios, the highest exposures for small terrestrial vertebrates will occur after a direct spray and could reach up to about 85 mg/kg under typical exposure conditions and up to about 859 mg/kg under more extreme conditions. Substantially lower doses are anticipated from the consumption of contaminated vegetation: up to about 6 to 9 mg/kg under typical conditions with an upper range of 17 to 27 mg/kg. The consumption of contaminated water will generally lead to much lower levels of exposure. A similar pattern is seen for chronic exposures. Estimated daily doses for the a small vertebrate from the consumption of contaminated vegetation are in the range of 0.00006 to 3 mg/kg/day. The upper ranges of exposure from contaminated vegetation far exceed doses that are anticipated from the consumption of contaminated water, 0.000005 mg/kg/day to 0.0002 mg/kg/day. Based on general relationships of body size to body volume, larger vertebrates will be exposed to lower doses and smaller animals, such as insects, to much higher doses than small vertebrates under comparable exposure conditions. Because of the apparently low toxicity of picloram to animals, the rather substantial variations in the different exposure assessments have little impact on the assessment of risk to terrestrial animals.

For terrestrial plants, five exposure scenarios are considered quantitatively: direct spray, spray drift, runoff, wind erosion and the use of contaminated irrigation water. Unintended direct spray is expressed simply as the application rates considered in this risk assessment, 0.35 lb a.e./acre and should be regarded as an extreme/accidental form of exposure that is not likely to occur in most Forest Service applications. Estimates for the other routes of exposure are much less. All of these exposure scenarios are dominated by situational variability because the levels of exposure are highly dependent on site-specific conditions. Thus, the exposure estimates are intended to represent conservative but plausible ranges that could occur but these ranges may over-estimate or under-estimate actual exposures in some cases. Spray drift is based on estimates AGDRIFT. The proportion of the applied amount transported off-site from runoff is based on GLEAMS modeling of clay, loam, and sand. The amount of picloram that might be transported off-site from wind erosion is based on estimates of annual soil loss associated with wind erosion and the assumption that the herbicide is incorporated into the top 1 cm of soil. Exposure from the use of contaminated irrigation water is based on the same data used to estimate human exposure from the consumption of contaminated ambient water and involves both monitoring studies as well as GLEAMS modeling.

Exposures to aquatic plants and animals is based on essentially the same information used to assess the exposure to terrestrial species from contaminated water. The peak estimated rate of contamination of ambient water associated with the normal application of picloram is 0.05 (0.01 to 0.2) mg a.e./L at an application rate of 1 lb a.e./acre. For longer-term exposures, average estimated rate of contamination of ambient water associated with the normal application of picloram is 0.001 (0.0001 to 0.004) mg a.e./L at an application rate of 1 lb a.e./acre. For longer-term exposures, average estimated rate of contamination of ambient water associated with the normal application of picloram is 0.001 (0.0001 to 0.004) mg a.e./L at an application rate of 1 lb a.e./acre. For the assessment of potential hazards, these contamination rates are adjusted based on the application rates considered in this risk assessment.

Dose-Response Assessment – For terrestrial mammals, the dose-response assessment for chronic exposure is based on a NOAEL of 7 mg/kg/day from a 6-month dog feeding study. For acute exposures, a NOAEL of 34 mg/kg/day is used based on a teratogenicity study in rabbits. For birds, short term feeding studies are used to estimate an acute NOAEL of 1,500 mg/kg body weight. No adequate data are available on chronic toxicity in birds and the chronic NOAEL of 7 mg/kg/day derived for mammals is used to characterize risk. Relatively little data is available on terrestrial invertebrates and a standard LD_{50} value of >1000 mg/kg in bees is used to characterize risk in terrestrial invertebrates.

For assessing the potential consequences of exposures to nontarget plants via runoff, a NOEC of 0.000012 lb a.e./acre is used for sensitive species and a NOEC of 0.4 lb a.e./acre is used for tolerant species. For assessing the impact of drift, bioassays on vegetative vigor are used with a NOEC of 0.00021 lb a.e./acre for sensitive species and a NOEC of 0.062 lb a.e./acre for tolerant species.

Soil microorganisms may display detectable responses to picloram at relatively low concentrations and true NOEC values for effects on microorganisms are not available. For this risk assessment, a soil concentration of 1 ppm is used as benchmark dose and the potential consequences of soil contamination by picloram is considered further in the risk characterization.

The general dose-response assessment for aquatic species is characterized by substantial variability within different groups (fish, invertebrates, and plants) but few substantial difference among the different groups. In general, sensitive species have NOECs or LC_{50} values in the 0.2 to 4 mg/L range and tolerant species have NOECs or LC_{50} values in the 10 to over 100 mg/L range. Trout appear to be the most sensitive animal species, with acute LC_{50} values as low as 0.8 mg/L. The dose response assessment for aquatic species is complicated by a very low reported LOEC, 0.035 mg/L in lake trout. This is an older study that was not designed to meet current standards. Nonetheless, this study is well documented and is not discounted. A relatively low LOEC of 0.1 mg/L has also been reported in one species of macrophyte. The observed effect was a transient delay in flowering with no inhibition of growth.

Risk Characterization – Picloram is a herbicide and the most likely damage to nontarget species will involve terrestrial plants. As is the case with any herbicide, the likelihood of damage to nontarget plant species is related directly to the difference between the sensitivity of target species—which dictates the application rate—and the sensitivity of the potential nontarget species. Sensitive plant species could be adversely affected by the off-site transport of picloram under a variety of different scenarios depending on local site-specific conditions that cannot be generically modeled. If picloram is applied in the proximity of sensitive crops or other desirable sensitive plant species, site-specific conditions and anticipated weather patterns will need to considered if unintended damage is to be avoided. More tolerant plant species are not likely to be affected unless they are directly sprayed or subject to substantial drift. A detectable inhibition of the activity of soil microorganisms is also likely at application rates used in Forest Service programs. These changes could lead to an increase in the persistence of picloram in soil and/or a more general decrease in microbial activity. That this inhibition would be associated with

detectable changes in soil productivity or other undesirable gross effects is much less certain. The potential for adverse effects on other terrestrial nontarget animal species appears to be remote. The weight of evidence suggests that no adverse effects in terrestrial animals are plausible using typical or even very conservative worst case exposure assumptions.

There is substantial variability in the toxicity of picloram to aquatic species. While this variability adds uncertainty to the dose-response assessment, it has no substantial impact on the risk characterization. None of the hazard indices for fish, aquatic invertebrates, or aquatic plants reach a level of concern.

The risk characterization for both terrestrial and aquatic species is limited by the relatively few animal and plant species on which data are available compared to the large number of species that could potentially be exposed. This limitation and consequent uncertainty is common to most if not all ecological risk assessments.

1. INTRODUCTION

Two commercial formulations of picloram, Tordon K and Tordon 22K, are used by the Forest Service in vegetation management programs. The USDA Forest Service has conducted previous risk assessments on picloram as part of environmental impact statements (USDA 1989a,b,c) and has prepared a herbicide background statement on picloram (USDA 1989d). In addition, the USDA Forest Service had a risk assessment prepared on picloram in 1999 (SERA 1999). The present document provides an update to the 1999 risk assessment for human health effects and ecological effects to support an assessment of the environmental consequences of using picloram in future Forest Service programs.

This is a technical support document and it addresses some specialized technical areas. Nevertheless an effort was made to ensure that the document can be understood by individuals who do not have specialized training in the chemical and biological sciences. Certain technical concepts, methods, and terms common to all parts of the risk assessment are described in plain language in a separate document (SERA 2001). Some of the more complicated terms and concepts are defined, as necessary, in the text.

In the preparation of this risk assessment, literature searches of picloram were conducted in the open literature using PubMed, TOXLINE as well as the U.S. EPA CBI files. There is a very large body of literature on the environmental fate and toxicology of picloram. In addition to the herbicide background statement on picloram (USDA 1989d), the toxicology and environmental fate of picloram have been reviewed by the U.S. EPA (U.S. EPA 1992a,b; U.S. EPA 1994; U.S. EPA 1995a,b; U.S. EPA 1998a; U.S. EPA 1999) as well as the USDA (USDA 1995; ExToxNet 1996a). An additional review of picloram has been published by Cox (1998) and a review and reevaluation of data supporting the ecological risk assessment of picloram has been submitted to the U.S. EPA (Havens and Peacock 1995). The U.S. EPA (1995a) re-registration eligibility decision (RED) document also includes a summary of the product chemistry, mammalian toxicology, and ecotoxicology studies that were submitted by industry to the U.S. EPA as part of the registration process for this compound. Full text copies of key studies (n=64) were kindly provided by the U.S. EPA Office of Pesticide Programs. The CBI studies were reviewed, and synopses of the information that can be disclosed from these studies are included in this document.

While this document discusses the studies required to support the risk assessments, it makes no attempt to re-summarize all of the information cited in the existing reviews. The Forest Service will update this and other similar risk assessments on a periodic basis and welcomes input from the general public on the selection of studies included in the risk assessment. This input is helpful, however, only if recommendations for including additional studies specify why and/or how the new or not previously included information would be likely to alter the conclusions reached in the risk assessments.

For the most part, the risk assessment methods used in this document are similar to those used in risk assessments previously conducted for the Forest Service as well as risk assessments conducted by other government agencies. Details regarding the specific methods used to prepare the human health risk assessment are provided in SERA (2001). This document has four chapters, including the introduction, program description, risk assessment for human health effects, and risk assessment for ecological effects or effects on wildlife species. Each of the two risk assessment chapters has four major sections, including an identification of the hazards associated with picloram and its commercial formulation, an assessment of potential exposure to the product, an assessment of the dose-response relationships, and a characterization of the risks associated with plausible levels of exposure. These are the basic steps recommended by the National Research Council of the National Academy of Sciences (NRC 1983) for conducting and organizing risk assessments.

Variability and *uncertainty* may be dominant factors in any risk assessment, and these factors should be expressed. Within the context of a risk assessment, the terms *variability* and *uncertainty* signify different conditions.

Variability reflects the knowledge of how things may change. Variability may take several forms. For this risk assessment, three types of variability are distinguished: *statistical*, situational, and arbitrary. Statistical variability reflects, at least, apparently random patterns in data. For example, various types of estimates used in this risk assessment involve relationships of certain physical properties to certain biological properties. In such cases, best or maximum likelihood estimates can be calculated as well as upper and lower confidence intervals that reflect the statistical variability in the relationships. Situational variability describes variations depending on known circumstances. For example, the application rate or the applied concentration of a herbicide will vary according to local conditions and goals. As discussed in the following section, the limits on this variability are known and there is some information to indicate what the variations are. In other words, *situational variability* is not random. *Arbitrary* variability, as the name implies, represents an attempt to describe changes that cannot be characterized statistically or by a given set of conditions that cannot be well defined. This type of variability dominates some spill scenarios involving either a spill of a chemical on to the surface of the skin or a spill of a chemical into water. In either case, exposure depends on the amount of chemical spilled and the area of skin or volume of water that is contaminated.

Variability reflects a knowledge or at least an explicit assumption about how things may change, while *uncertainty* reflects a lack of knowledge. For example, the focus of the human health dose-response assessment is an estimation of an "acceptable" or "no adverse effect" dose that will not be associated with adverse human health effects. For picloram and for most other chemicals, however, this estimation regarding human health must be based on data from experimental animal studies, which cover only a limited number of effects. Generally, judgment is the basis for the methods used to make the assessment. Although the judgments may reflect a consensus (i.e., be used by many groups in a reasonably consistent manner), the resulting

estimations of risk cannot be proven analytically. In other words, the estimates regarding risk involve uncertainty.

In considering different forms of variability, almost no risk estimate presented in this document is given as a single number. Usually, risk is expressed as a central estimate and a range, which is sometimes very large. Because of the need to encompass many different types of exposure as well as the need to express the uncertainties in the assessment, this risk assessment involves numerous calculations.

Most of the calculations are relatively simple, and the very simple calculations are included in the body of the document. Some of the calculations, however, are cumbersome. For those calculations, a set of worksheets is included as an attachment to the risk assessment. The worksheets provide the detail for the estimates cited in the body of the document. The worksheets are divided into the following sections: general data and assumptions, chemical specific data and assumptions, exposure assessments for workers, exposure assessments for the general public, and exposure assessments for effects on nontarget organisms. The worksheets are included at the end of this risk assessment and further documentation for these worksheets are available: one in a word processing format and one in a spreadsheet format. The work sheets that are in the spreadsheet format are used only as a check of the worksheets that are in the word processing format. Both sets of worksheets are provided with the hard-text copy of this risk assessment as well as with the electronic version of the risk assessment.

Technical grade picloram contains hexachlorobenzene as a contaminant and hexachlorobenzene is classified as a carcinogen. Because of the importance of and level of concern for this endpoint in humans, the human health risk assessment discusses the potential effects of hexachlorobenzene in some detail and a separate subset of worksheets for hexachlorobenzene are provided at the end of this document. Again, these worksheets are provided in both a word processing format and one in a spreadsheet format.

2. PROGRAM DESCRIPTION

2.1. OVERVIEW

Picloram is a herbicide used in the control of a number of broadleaf weeds and undesirable brush. Picloram is used in Forest Service programs almost exclusively for the control of noxious weeds. Very minor uses include rights-of-way management and general maintenance. Tordon K and Tordon 22K are the formulations of picloram currently available and used by the Forest Service. Both formulations are produced by Dow AgroSciences as a liquid containing the potassium salt of picloram (24.4% w/v). This is equivalent to a concentration of 2 lb a.e./gallon. The remaining 75.6% of the formulation consists of inerts, including a polyglycol. The U.S. EPA has placed the polyglycol on List 3 of the inerts that may be used in the formulation of pesticides. Very little additional information is available on this compound.

The most common methods of ground application for Tordon involve backpack (selective foliar) and boom spray (broadcast foliar) operations. Mist blower application of picloram is not permitted. The Forest Service does not typically use aerial applications for picloram. Nonetheless, Tordon is registered for aerial applications and aerial applications are included in this risk assessment in the event the Forest Service may wish to consider this application method. The labeled application rates for picloram range from 0.125 to 1 lb a.e./acre. Typically, the Forest Service uses rates in the lower part of this range and some applications may be below the lower range of the labeled rate. For this risk assessment, the typical rate of 0.35 lb a.e./acre with a lower range of 0.1 lb a.e./acre is used to reflect Forest Service practice. An upper range of 1 lb a.e./acre is used to assess the consequences of using the highest labeled rate should the Forest Service need to consider this option.

2.2. CHEMICAL DESCRIPTION AND COMMERCIAL FORMULATIONS

Picloram is the common name for 4-amino-3,5,6-trichloropicolinic acid:



Selected chemical and physical properties of picloram are summarized in Table 2-1. Additional information is presented in Worksheet B03.

The only formulations of picloram used by the Forest Service are Tordon K and Tordon 22K, both of which are produced by Dow AgroSciences. Both of these are formulated as a liquid containing the potassium salt of picloram (24.4% w/v). This is equivalent to a concentration of 2 lb a.e./gallon. The remaining 75.6% of the formulation consists of inerts. The identity of all inerts has been disclosed to the U.S. EPA as part of the registration process and this information has been reviewed in the preparation of this risk assessment (Lanman 1996a,b,c). This

information is classified as CBI (confidential business information) under Section 7(d) and Section (10) of FIFRA. Except as noted below, this information cannot be specifically disclosed in this risk assessment.

Some inerts - i.e., those listed under SARA Title III, Section 313 - are specified on the product material safety data sheets and can be publicly disclosed. On the MSDS's for Tordon K and Tordon 22K, one inert is listed as polyglycol, with a CAS No. 069029-39-6 (C&P Press 2003). On the U.S. EPA list of inerts used in pesticides (U.S. EPA/OPP 2003), the polyglycol is listed as polyoxypropylene mono(disec-butylphenyl) ether and classified as a List 3 inert. List 3 inerts designate those inerts for which the available toxicology data are insufficient to classify the the compound as of toxicologic concern (List 1), possible toxicologic concern (List 2), or of minimal concern (List 4)(U.S. EPA/OPP 2003). Additional information on this and other adjuvants is presented in Section 3.1.14.

Technical grade picloram contains hexachlorobenzene as a contaminant (U.S. EPA 1995b). Nominal or average concentrations of hexachlorobenzene are 8 ppm and the maximum concentration of hexachlorobenzene currently in technical grade picloram is 50 ppm (McMaster 1999). The impact of this contaminant and other impurities on the human health risk assessment is detailed in Section 3.1.15.

Both of the Tordon formulations are labeled for the control of a variety of broadleaf weeds, woody plants, and vines in non-crop areas. Only Torodon K is registered specifically for forestry. Tordon 22 is registered for rights-of-way and other non-cropland uses such as along road sites and fence rows (C&P Press 2003). When Tordon K is used in forest planting sites, periods of 6 to 12 months after treatment are recommended before the planting of conifers. A cautionary note on the product label indicates that legume seedlings may not grow for two years after treatments. Recommended adjuvants include non-ionic surfactants. In aerial and some ground broadcast applications, thickening agents are also recommended to reduce drift (C&P Press 2003).

2.3. APPLICATION METHODS

The most common methods of ground application for Tordon involve backpack (selective foliar) and boom spray (broadcast foliar) operations. In selective foliar applications, the herbicide sprayer or container is carried by backpack and the herbicide is applied to selected target vegetation. Application crews may treat up to shoulder high brush, which means that chemical contact with the arms, hands, or face is plausible. To reduce the likelihood of significant exposure, application crews are directed not to walk through treated vegetation. Usually, a worker treats approximately 0.5 acre/hour with a plausible range of 0.25–1.0 acre/hour.

Boom spray is used primarily in rights-of-way management. Spray equipment mounted on tractors or trucks is used to apply the herbicide on either side of the roadway. Usually, about 8 acres are treated in a 45-minute period (approximately 11 acres/hour). Some special truck mounted spray systems may be used to treat up to 12 acres in a 35-minute period with

approximately 300 gallons of herbicide mixture (approximately 21 acres/hour and 510 gallons/hour) (USDA 1989a, p. 2-9 to 2-10). The Tordon formulations may not be applied with a mist-blower.

Both Tordon formulations are registered for aerial applications (C&P Press 2003). Although this is not an application method that the Forest Service will typically employ for picloram, this method is covered by this risk assessment in the event that the Forest Service may need to consider aerial applications. Aerial applications may be made using helicopters. Tordon is applied under pressure through specially designed spray nozzles and booms. The nozzles are designed to minimize turbulence and maintain a large droplet size, both of which contribute to a reduction in spray drift. In aerial applications, approximately 40–100 acres may be treated per hour.

2.4. MIXING AND APPLICATION RATES

The specific application rates used in a ground application vary according to local conditions and the nature of the target vegetation. Application rates may be expressed in various units such as gallons of formulation per acre (used in most product labels), lb a.i. per acre (designating the amount of the potassium salt of picloram), or lb a.e. per acre (designating the amount of the picloram acid equivalents). Unless otherwise specified, all application rates and other expressions of amounts are based on acid equivalents.

Application rates of $\frac{1}{4}$ to 2 quarts Tordon/acre are recommended on the product labels and no more than 2 quarts Tordon/acre may be applied in a single growing season (C&P Press 2003). The application rates of $\frac{1}{4}$ to 2 quarts Tordon/acre are equivalent to 0.0625–0.5 gallons Tordon per acre. Given that there is 2 lbs picloram a.e./gallon in the Tordon formulations, these rates correspond to 0.125 to 1 lb picloram a.e./acre.

The use of picloram in Forest Service Programs for fiscal year 2001, the most recent year for which data are available, is summarized in Table 2-2. Picloram is used currently in Forest Service Programs primarily almost exclusively in noxious weed control (99.92%). Other minor uses (totaling about 0.1% of total use) include recreations improvement and rights-of-way management. Based on the total amount used and number of acres treated, the application rates are about 0.2 lb/acre for noxious weed control, 0.5 lb/acre for recreation improvement, and 0.25 lb/acre for rights-of-way management.

For this risk assessment, the typical application rate will be taken as 0.35 lb a.e./acre. This is about the average value of all individual applications conducted by the Forest Service in 2001 (0.3432 lbs/acre). It should be noted that this is greater than the arithmetic average of about 0.2 lbs/acre for all applications combined – i.e., the total pounds applied divided by the total acres treated.

The range of application rates will be taken as 0.1 lb a.e./acre to 1 lb a.e./acre to reflect plausible ranges that the Forest Service may use. The lower range of the application is essentially

arbitrary. Lower application rates have been reported and these reports may involve spot applications conducted sporadically over a relatively large area. The upper range of the application rate will be taken as 1 lb/acre, the maximum labeled rate. The worksheets that accompany this risk assessment are based on the typical application rate of 0.35 lb/acre rather than the full range of application rates. The consequences of varying application rates within the range of 0.1 to 1 lb/acre is considered in the risk characterization for human health (Section 3.4) and ecological effects (Section 4.4).

For forestry applications, mixing volumes of 5 to 25 gallons of water per acre are recommended for aerial applications. Recommended mixing volumes for ground applications range from 10 to 100 gallons of water per acre (C&P Press 2003). For this risk assessment, the extent to which a picloram formulation is diluted prior to application primarily influences dermal and direct spray scenarios, both of which are dependent on the 'field dilution' (i.e., the concentration of picloram in the applied spray). The higher the concentration of picloram, the greater the risk. For this risk assessment, the lowest dilution will be taken at 5 gallons/acre, the minimum recommended for aerial applications. The highest dilution (i.e., that which results in the lowest risk) will be based on 100 gallons of water per acre, the highest application volume recommended for ground applications. The typical dilution rate will be taken as 30 gallons/acre, approximately the geometric mean of the range recommended for ground applications [$(10 \times 100)^{0.5}=31.6$].

It should be noted that the selection of application rates and dilution volumes in this risk assessment is intended to simply reflect typical or central estimates as well as plausible lower and upper ranges. Forest Service analysts may use different input variables such as application rate, rainfall or slope to recalculate values in the attached worksheets in order to assess any differences in potential human health or ecological risks for site-specific projects.

2.5. USE STATISTICS

The USDA Forest Service (USDA/FS 2002) tracks and reports use by geographical areas referred to as "*Regions*". As illustrated in Figure 2-1, the Forest Service classification divides the U.S. into nine regions designated from Region 1 (Northern) to Region 10 (Alaska). [Note: There is no *Region 7* in the Forest Service system.] As illustrated in Figure 2-1 and detailed further in Table 2-3, the heaviest used of picloram occurs in the northern central regions: Region 1 (Northern) Region 2 (Rocky Mountain), and Region 4 (Inter-mountain) with a lesser amount used in Region 6 (Pacific Northwest). Only very small quantities of picloram are used in the Region 9 (Eastern) and Region 8 (Southeastern).

Picloram is used extensively in agriculture. A summary of the agricultural use of picloram is presented in Figure 2-3 (USGS 1998). These use statistics are for 1992, the most recent year for which data are available. As indicated in this figure, over 1,700,000 lbs of picloram are applied to crops annually, primarily to pasture, with minor uses including hay, wheat, grains, flax, oats, and barley. Picloram is used for agriculture in the same areas in which the use by the Forest Service is predominant. In addition, substantial amounts of picloram are used in the south central and south western areas of the United States. As noted in Table 2-3, the total annual use

of picloram by the Forest Service for 2001 is about 12,900 lbs, which is 0.76 percent of the agricultural use. While the use of picloram by the Forest Service is not trivial, this use is less than that of agricultural uses by a factor of over 130. Thus, there is no basis for asserting that Forest Service programs will substantially contribute to general concentrations of picloram nationally. The potential for local contamination of environmental media by the use of picloram in Forest Service programs is discussed in detail in the human health risk assessment (Section 3) and the ecological risk assessment (Section 4).

3. HUMAN HEALTH RISK ASSESSMENT

3.1. HAZARD IDENTIFICATION

3.1.1. Overview. The toxicity of picloram to experimental mammals has been very well-characterized. Most of the studies have been conducted in support of the registration of picloram and are summarized in the U.S. EPA re-registration eligibility decision document. Picloram has a low order of acute toxicity, with acute oral LD_{50} values in the range of 3000 to 5000 mg/kg body weight. Picloram can cause irritation to the eyes. Although picloram is not a strong skin irritant, repeated dermal exposures may lead to skin sensitization.

In chronic toxicity studies, the most sensitive effect for picloram in mammals involves effects on the liver. The current U.S. EPA RfD is based on a two-year feeding study in male and female Fischer rats in which picloram acid was administered at dietary concentrations that resulted in daily doses of 20, 60, and 200 mg/kg/day. The only statistically significant observations included an increase in liver size and an alteration in the staining properties of centrilobular hepatocytes in the 60 and 200 mg/kg/day dose groups. Dogs appear to be somewhat more sensitive to picloram than rats. In a six month feeding study in which male and female beagle dogs were administered picloram at levels that resulted in average daily doses of 0, 7, 35, and 175 mg/kg/day, the two higher dose levels resulted in increase in absolute and relative liver weight in two males and changes in liver enzyme activity.

Although technical grade picloram has been subject to several chronic bioassays for carcinogenicity and none of the bioassays have shown that picloram has carcinogenic potential, technical grade picloram does contain hexachlorobenzene, a compound that has shown carcinogenic activity in three mammalian species and has been classified as a potential human carcinogen by the U.S. EPA. Thus, this effect is considered both qualitatively and quantitatively in this risk assessment.

3.1.2. Mechanism of Action. While the mechanism of action of picloram in plants is well understood (Section 4.1.2.4), any specific mechanism of action in humans or experimental animals is unclear. As discussed in Section 3.1.5, the most sensitive effect of picloram in mammals appears to involve effects on the liver. At physiologic pH, picloram will be predominantly in the anion (acid) form and the liver does have a non-specific anion active transport system (Moslen 1996). It seems unlikely, however, that this has a substantial impact on the sensitivity of the liver to picloram because other organs, particularly the kidney, also have extremely efficient anion active transport systems (Goldstein and Schnellmann. 1996). The liver also has a major involvement in the metabolism of many chemicals, both naturally occurring and synthetic. Picloram, however, is not metabolized substantially by the liver or other organ (Section 3.1.15) and thus there is not basis for asserting that metabolic activation could account for the apparent sensitivity of the liver to picloram.

3.1.3. Kinetics and Metabolism. The oral absorption kinetics of picloram in humans has been studied by Nolan et al. (1984), who administered picloram in about 100 mL of grape juice to six

male caucasian volunteers at doses of 0.5 and 5.0 mg/kg. The compound was taken orally in grape juice and the investigators report that picloram has a bitter taste that is not masked by grape juice. The average body weights of the volunteers was 78.5 kg. Thus, the concentrations of picloram in the grape juice were about 392.5 mg/L ($0.5 \text{ mg/kg} \times 78.5 \text{ kg} \div 0.1\text{L}$) and 3925 mg/L ($5 \text{ mg/kg} \times 78.5 \text{ kg} \div 0.1\text{L}$). Taste thresholds for picloram in water or other liquids have not been encountered in the literature. The absorption and elimination kinetics of picloram in the human volunteers was described by a two compartment model and the average oral first order absorption rate coefficient was about 2 hour⁻¹. The elimination kinetics of picloram by humans follows a two-compartment model with halftimes of about 1 and 19 hours. Over 75% of the administered picloram was eliminated after 6 hours and over 90% of the administered dose was eliminated after 72 hours (Nolan et al. 1984).

Biphasic elimination kinetics have been observed in rats, dogs, and cattle (U.S. EPA 1992b). Based on the plateau principle (e.g., Goldstein et al. 1974; O'Flaherty 1981), the concentration at infinite time (C_{∞}) relative to the concentration after the first treatment (C_{0}) may be calculated as:

$$C_{\infty} \div C_0 = 1 \div (1 - e^{-k \Delta t})$$

where, k is the elimination rate in units of reciprocal time and Δt is the time interval between treatments. The terminal halftime of 40 hours reported by Nolan et al. 1984 corresponds to an elimination rate of 0.017 hours⁻¹ [k = ln(2)÷t_{1/2}]. Assuming an exposure interval of 1 day (24 hours), the maximum accumulation of picloram in humans would be expected to be about a factor 3:

$$1 \div (1 - e^{-0.017 \times 24 \text{ hours}}) = 2.94$$

Most of the occupational exposure scenarios and many of the exposure scenarios for the general public involve the dermal route of exposure. For these exposure scenarios, dermal absorption is estimated and compared to an estimated acceptable level of oral exposure based on subchronic or chronic toxicity studies in animals. Thus, it is necessary to assess the consequences of dermal exposure relative to oral exposure and the extent to which picloram is likely to be absorbed from the surface of the skin. Two types of dermal exposure scenarios are considered: immersion and accidental spills. As detailed in SERA (2001), the calculation of absorbed dose for dermal exposure scenarios involving immersion or prolonged contact with chemical solutions use Fick's first law and require an estimate of the permeability coefficient, K_p , expressed in cm/hour. For exposure scenarios like direct sprays or accidental spills, which involve deposition of the compound on the skin's surface, dermal absorption rates (proportion of the deposited dose per unit time) rather than dermal permeability rates are used in the exposure assessment.

The dermal absorption of picloram in humans was also studied by Nolan et al. (1984). A dose of 2 mg/kg was applied to the back of each volunteer (over about a 1000 cm² area) and the volunteers were instructed to shower 12 to 14 hours after application. An average proportion (P) of 0.0018 of the applied dose was excreted by 6 human volunteers over a 72 hour period (Nolan

et al. 1984, Table 1, column 3). Among the 6 volunteers, proportion of the dose excreted in the urine ranged from 0.0004 to 0.0048. As with the oral absorption kinetics, Nolan et al. (1984) used a two compartment model to describe the elimination kinetics of picloram. Unlike the study of the oral absorption kinetics, however, the low dermal absorption of picloram prevented the direct estimate of the kinetic parameters from the dermal phase of the study and the group average kinetic parameters for excretion from the oral study were used to estimate dermal absorption rates.

Nolan et al. (1984) report an average first-order dermal absorption rate 0.056 hour⁻¹ with a range of 0.031 hour⁻¹ to 0.075 hour⁻¹ (Nolan et al. 1984, Table 1, column 6). These values, however, are not consistent with the reported recovery of picloram in the urine. Under the assumption of first-order absorption, the proportion absorbed (P) at time t is:

$$P = 1 - e^{-kt}$$

Assuming rapid urinary excretion, as noted in the oral phase of the Nolan et al. (1984) study, a dermal absorption rate of 0.056 hour⁻¹ over a 13 hour exposure period – i.e., the central point in the showering interval – the proportion absorbed would be 0.51 or about 50%. As noted above, however, the average proportion recovering in the urine was only 0.0018 of the applied dose or about 0.2%. The reason for the discrepancy between the dermal absorption rates and the urinary recovery reported by Nolan et al. (1984) is not apparent.

By rearrangement of the above equation, an alternative absorption rate can be estimated as:

$$k = -\ln(1-P)/t$$

where *P* is the proportion absorbed and *t* is the period of exposure (e.g., Goldstein et al. 1974, p. 302). Thus, if 0.0018 of the administered dose is recovered in the urine following a 13 hour period of exposure, the first order dermal absorption rate can be estimated at about 0.00014 hour⁻¹:

$$ka = -\ln(1 - 0.0018)/13$$
 hours = 0.00014 hours⁻¹

Taking the range of the proportions of the applied dose recovered in the urine, 0.0004 to 0.0048 (Nolan et al. 1984, Table 1, column 3), the range of dermal absorption rates may be calculated as 0.00003 hour⁻¹ to 0.0004 hour⁻¹.

By comparison, the first-order dermal absorption rate estimated from the molecular weight and octanol-water partition coefficient of picloram is 0.0013 hour⁻¹ with a range of 0.00051 to 0.0035 hour⁻¹ (Worksheet B03). The central estimate is higher than the value based on the Nolan et al. (1984) data by about a factor of about 9 [0.0013 hour⁻¹ \div 0.00015 hours⁻¹ = 8.66].

For this risk assessment, the estimates based on the study by Nolan et al. (1984) are used. The Nolan et al. (1984) study appears to have been properly conducted and there is no basis for using the much higher estimated dermal absorption rates from Worksheet B03.

3.1.4. Acute Oral Toxicity. Standard acute toxicity studies, required by the U.S. EPA for pesticide registration, have been conducted as are summarized in U.S. EPA (1995b). These studies have been obtained and reviewed in the preparation of this risk assessment. U.S. EPA (1995b) cites an LD_{50} value in excess of 5000 mg/kg for male rats for picloram acid (Jeffrey 1987a) as well as the potassium salt of picloram as Tordon K salt liquor, an intermediate in the manufacture of picloram (Jeffrey et al. 1987e). Female rats appear to be somewhat more sensitive, with LD_{50} values of 4012 mg/kg (95% CI 3091-6654 mg/kg) for picloram acid (Jeffrey 1987a) and 3536 mg/kg for the potassium salt of picloram, again as Tordon K salt liquor (Jeffrey et al. 1987e). Based on the Tordon 22K formulation, however, there is no apparent difference in sensitivity between male and females rats, with no mortality or signs of toxicity over a two-week observation period at an acute dose of 5,000 mg/kg, expressed as the formulation (Jeffrey et al. 1987g).

In a review, Cox (1998) interpreted to the data of Hayes et al. (1986) as suggesting that the acute mammalian toxicity is approximately 5-fold lower for female rats and several fold lower for male rats (LD_{50} values of 690 mg/kg and 950 mg/kg, respectively). These results, however, were based on tests using doses of unneutralized potassium picloram solution (pH>11) (Hayes et al. 1986). U.S. EPA (1992b) concluded that the relatively low LD_{50} values obtained by Hayes et al. (1986) compared to the results of other investigators "*is probably due in part to the extreme alkalinity of the dosing solution*" (U.S. EPA 1992b, page V-3).

3.1.5. Subchronic or Chronic Systemic Toxic Effects. Systemic toxicity encompasses virtually any effects that a chemical has after the chemical has been absorbed. Certain types of effects, however, are of particular concern and involve a specific subset of tests. Such special effects are considered in following subsections and include effects on the nervous system (Section 3.1.6) and immune system (Section 3.1.7), development or reproduction (Section 3.1.8), and carcinogenicity or mutagenicity (Section 3.1.9). This section encompasses the remaining signs of general and non-specific toxicity.

Although some subchronic dermal toxicity studies have been conducted on picloram (Section 3.1.11), most of the subchronic and chronic toxicity studies on picloram have involved oral exposures. While these studies are summarized in the RED (U.S. EPA 1995b), more detailed summaries are given in various U.S. EPA reports that document the RfD for picloram (U.S. EPA 1992c, 1994, 1999).

The most sensitive effect for picloram in mammals involves effects on the liver. The current U.S. EPA RfD (Section 3.3.2) is based on a two-year feeding study in male and female Fischer rats (50 rats/sex/dose) in which picloram (acid) was administered at dietary concentrations that resulted in daily doses of 20, 60, and 200 mg/kg/day (Landry et al. 1986). Interim sacrifices

(10 rats/sex/dose) were made at 6-months and 1-year. At the end of the two-year exposure, the only statistically significant observations included an increase in liver size and an alteration in the staining properties of centrilobular hepatocytes in the 60 and 200 mg/kg/day dose groups. Both of these effects were more pronounced in males than in females. Increased liver weights as well as slight increases in the size and pallor of centrilobular hepatocytes were also seen in the 6-month and 12-month interim sacrifices. While the U.S. EPA has classified 60 mg/kg/day as the LOAEL for this study (U.S. EPA 1992b, 1999), the U.S. EPA/OPP RfD workgroup (U.S. EPA 1994) "...*felt that the LOAEL might have been higher*". In other words, while effects were seen at 60 mg/kg/day, the magnitude and severity of these effects were not regarded with substantial concern by the workgroup.

Similar effects on the liver have been noted in a 6-month dog feeding study (Barna-Lloyd et al. 1982). In this study, male and female beagle dogs were administered picloram (acid) in the diet at levels that resulted in average daily doses of 0, 7, 35, and 175 mg/kg/day. At the two higher dose levels, increases were noted in absolute and relative liver weight in two males and changes in liver enzyme activity were noted at the highest dose level. Additional effects included decreased food consumption and decreased body weight gain. As discussed further in Section 3.3.2, this study served as the based for the previous U.S. EPA RfD on picloram.

As reviewed in some detail by U.S. EPA (1992b), increased liver weight as well as changes in the appearance of the liver have also been observed in studies in mice as well as other studies in rats and dogs. There is very little evidence, however, that picloram is likely to impact other organs. At very high doses (i.e., 370 and 740 mg/kg/day) 80-week dietary exposures to picloram induced a number of different gross effects in rats including dermatitis accompanied by changes in hair coats and alopecia, diarrhea, abdominal distention, discolored urine, and vaginal bleeding. Pathological changes were also noted in the parathyroid, thyroid, and testes but not at frequencies that were statistically significantly different from control animals.

3.1.6. Effects on Nervous System. As discussed in Durkin and Diamond (2002), a *neurotoxicant* is chemical that disrupts the function of nerves, either by interacting with nerves directly or by interacting with supporting cells in the nervous system. This definition of *neurotoxicant* is critical because it distinguishes agents that act directly on the nervous system (*direct neurotoxicants*) from those agents that might produce neurologic effects that are secondary to other forms of toxicity (*indirect neurotoxicants*). Virtually any chemical will cause signs of neurotoxicat. This is the case for picloram. At high doses that produce a broad spectrum of toxicologic effects, clinical signs of acute picloram poisoning include ataxia, tremors, convulsions, and weakness (U.S. EPA 1992b). These reports, however, do not implicate picloram as a direct neurotoxicant.

No studies designed specifically to detect impairments in motor, sensory, or cognitive functions in animals or humans exposed picloram have been reported in the open literature or in the listing of studies submitted to the U.S. EPA to support the registration an re-registration of picloram.

Specifically, the U.S. EPA/OPTS (2003) has standard protocols for a number of types of neurotoxicity studies including a neurotoxicity screening battery (Guideline 870.6200), acute and 28-day delayed neurotoxicity of organophosphorus substances (Guideline 870.6100). Neither of these types of studies have been conducted on picloram and the U.S. EPA/OPP (1995b) specifically states that such studies are not required. This is not surprising, since the undertaking of such studies on a substance such as picloram, for which the clinical and experimental toxicology experience provides no reason to suspect a direct neurotoxic potential, would be highly unusual.

3.1.7. Effects on Immune System. There is very little direct information on which to assess the immunotoxic potential of picloram. The only studies specifically related to the effects o picloram on immune function are skin sensitization studies conducted on picloram (Section 3.1.11). While these studies provide support for asserting that picloram may cause skin sensitization, they provide no information useful for directly assessing immune suppressive potential of picloram.

A commercial formulation of picloram and 2,4-D, Tordon 202C, has been shown to inhibit immune function in mice (Blakley 1997). The design of this study does not permit the determination of which agent caused the immune response or whether the immune response was attributable to a toxicologic interaction of the two herbicides. This formulation is not used in Forest Service programs.

3.1.8. Effects on Endocrine System. In terms of functional effects that have important public health implications, effects on endocrine function would be expressed as diminished or abnormal reproductive performance. This issue is addressed specifically in the following section (Section 3.1.9). Mechanistic assays are generally used to assess the potential for direct action on the endocrine system (Durkin and Diamond 2002). Picloram, however, has not been tested for activity as an agonist or antagonist of the major hormone systems (e.g., estrogen, androgen, thyroid hormone). Thus, all inferences concerning the potential effect of triclopyr on endocrine function must be based on inferences from standard toxicity studies.

A two-generation reproduction study of picloram (K salt) in CD rats reported no endocrine effects at doses as high as 1000 mg/kg/day (Breslin et al. 1991, as reviewed by U.S. EPA 1995b). Endocrine effect endpoints examined in this study included reproductive outcomes, histopathological examination of tissues. In this study, renal effects and increased body weight gain were observed at 1000 mg/kg/day (i.e., the maximum tolerated dose was tested). Of the other studies reviewed in this risk assessment, no evidence for picloram producing direct effects on the endocrine system was found. As noted above, however, picloram has not been specifically tested for activity as an agonist or antagonist of the major hormone systems.

3.1.9. Reproductive and Teratogenic Effects. The potassium salt of picloram has been tested for teratogenic in rats and rabbits and reproductive effects in rats. None of these studies have

been published in the open literature and all were submitted to U.S. EPA (1992b, 1995b; 1999) in support of the registration or reregistration of picloram.

In an oral gavage study with the potassium salt of picloram, doses of 0, 34, 172, and 344 mg a.e./kg/day were administered to New Zealand rabbits from days 6 to 18 of gestation (John et al. 1984). No effects were noted on offspring at the highest dose tested. The only effects noted on the dosed adults was decreased body weight, which occurred at 172 mg a.e./kg/day with a NOAEL of 34 mg a.e./kg/day. Another gavage teratology study on the potassium salt of picloram has been conducted in rats at doses of 0, 30, 150, 298 mg a.e./kg/day on days 6-15 of gestation (Schroeder 1990). The only effect seen in this study was excessive salivation in dams at 298 a.e./kg/day with a corresponding NOAEL of 150 mg a.e./kg/day. No adverse reproductive effects were noted. Other teratology studies summarized in various reviews by U.S. EPA (1992b, 1995b; 1999) involve salts or esters of picloram that are not used in Forest Service programs. In addition to these teratology studies, a two-generation reproduction study has been conducted on picloram acid. In this study, male and female rats were administered picloram in the diet at levels corresponding to doses at 0, 20, 200, or 1000 mg/kg/day. Histopathological effects on the kidney as well as other signs of kidney damage were noted at 1000 mg/kg/day. There were, however, no effects on reproductive performance (Breslin et al. 1991).

As with potential effects on the immune system (Section 3.1.7), there may be greater concern for mixtures of picloram and 2,4-D. Tordon 202c, a commercial formulation of picloram and 2,4-D, has been associated with adverse reproductive effects in mice (Blakley et al. 1989a,b,c). More recently, Oakes et al. (2002b) reported a statistically significant reduction in absolute and relative testicular weight (17% and 26%, respectively) in male Sprague Dawley rats exposed to Tordon 75D (75 g a.e./L picloram and 300 g a.e./L 2,4-D) by gavage 5 days/week for 9 weeks. The reduction in testicular weight occurred in the absence of adverse reproductive effects (Oakes et al. 2002a). The dose of Tordon 75D associated with a statistically significant reduction in testicular weight was 37.5 mg/kg, the NOAEL was 18.7 mg/kg (Oakes et al. 2002b). Exposure to Tordon 75D did not result in any evidence of male-mediated birth defects (Oakes et al. 2002a). Again, this formulation of 2,4-D and picloram is not used in Forest Service programs.

3.1.10. Carcinogenicity and Mutagenicity. Picloram has been tested for mutagenicity in a number of different test systems and has been assayed for carcinogenic activity in rats and mice. A review and detailed evaluation of the mutagenicity assays on picloram by U.S. EPA (1992b) concluded that:

No compelling evidence of a mutagenic effect in relevant biological systems was uncovered. Although picloram at a single reported dose was mutagenic in *S. coelicolor*, the weight of evidence from well-conducted microbial (Ames test), mammalian cell, and *Drosophila* mutagenicity studies tends to support the conclusion that picloram does not possess mutagenic activity (U.S. EPA 1992b, pp. V19 to V20).

Some additional studies have reported mutagenic activity in assays using higher plants. Mohammed and Ma (1999) reported dose dependent increase in *Tradescantia* micronucleus formation. Tomkins and Grant (1976) reported that picloram treatment produced a statistically significant increase in the frequency of chromosome aberration in *Pastinaca sativa* growing in normal field conditions.

The Health Effects Division Carcinogenicity Peer Review Committee of the U.S. EPA Office of Pesticides has reviewed the carcinogenicity data on picloram acid as well as the potassium salt of picloram and has classified these agents as Group E (no evidence of carcinogenicity) based on the lack of carcinogenic activity in rats and mice (U.S. EPA 1999).

In the past, some commercial preparations of picloram were formulated as the isooctyl ester of picloram. The compound used to produce this ester (ethylhexyl phthalate) is a potential carcinogen (U.S. EPA 1994). Formulations of picloram as the ethylhexyl ester are not used by the Forest Service.

Technical grade picloram is contaminated with hexachlorobenzene, a compound classified as a potential carcinogen by the U.S. EPA (1997). A recent review of the extensive toxicity data on hexachlorobenzene is available from ATSDR (2002). As discussed further in Section 3.1.15.1, the risk of cancer from this contaminant is considered both qualitatively and quantitatively in this risk assessment.

3.1.11. Irritation and Sensitization (Effects on the Skin and Eyes). As part of the herbicide registration process, standardized tests for skin and eye irritation as well as dermal sensitization are required and have been summarized in the RED for picloram (U.S. EPA 1995b). These data suggest that picloram exposure produces substantial, but temporary eye injury (Berdasco 1990a; Jeffrey 1987d). Both picloram acid and the potassium salt of picloram, which is contained in the formulations used by the Forest Service, are classified as moderate eye irritants (Category III) but as non-irritant to the skin (Category IV).

Tordon 22K (Gilbert 1996c) as well as a formulation mixture of picloram and 2,4-D (Tordon 101 M, Gilbert 1996a) noted transient skin irritation in standard acute dermal studies . Haut and Bell (1997) reported that Tordon 22K (20.6% picloram) produced a delayed hypersensitivity reaction in Hartley guinea pigs (8/10 animals exhibited a positive response after 48 hours). Haut and Bell (1997) used 0.4 mL of a 75% solution of Tordon 22K in water, during induction phase, and 0.4 mL of 50% solution of Tordon 22K in water, during the challenge phase. Other delayed hypersensitivity testing with lower concentrations of picloram (acid) in combination with 2,4-D found no dermal sensitization (Jeffrey 1987b; Gilbert 1996b; Berdasco 1990c).

3.1.12. Systemic Toxic Effects from Dermal Exposure. The toxicity studies summarized in the RED for picloram (U.S. EPA 1995b) indicate that dermal exposure to 2000 mg/kg picloram acid or the potassium salt of picloram was not associated with any signs of systemic toxicity in rabbits based on standard acute/single application bioassays with 14-day observation periods. In

general, dermal LD_{50} values are higher than oral LD_{50} values (e.g., Gaines 1969). Since the reported acute oral LD_{50} values of picloram are all 5000 mg/kg or greater, the lack of toxic effects at dermal doses of up to 2000 mg a.e./kg/day in New Zealand White rabbits (Jeffrey 1987c) is to be expected. In a test of Tordon K salt liquor (38.8% a.i., 33.4% a.e.) in New Zealand rabbits, no signs of toxicity were observed over a 2-week period after the application of 2000 mg/kg formulation (about 668 mg a.e./kg)(Jeffrey et al. 1987f). Similarly, no signs of toxicity were observed in New Zealand White rabbits after a dose of 2000 mg/kg Tordon 22K (20.36% a.e.), equivalent to about 407 mg a.e./kg (Jeffrey et al. 1987).

In addition to the acute dermal studies, a 21-day dermal toxicity assay of the potassium salt of picloram has been conducted in New Zealand white rabbits at doses of 0, 65, 217, and 650 mg a.e./kg/day, five days/week, for three weeks. No systemic toxic effects were observed (U.S. EPA 1995b). Another 21-day repeat dermal exposure study of XRM-5179 (mixture containing 2.86 % picloram and 10.71% 2,4-D) found localized dermal irritation but no signs of systemic toxicity were noted based on clinical signs, body weights, clinical pathology, organ weights, gross pathology, or histology (Mizzell et al. 1990).

3.1.13. Inhalation Exposure. Picloram is relatively non-volatile (Table 2-2). Acute aerosol inhalation toxicity data from tests of Tordon 22K (24.1% picloram acid) in Fisher 344 rats reported no deaths among 10 rats (5/sex) that were exposed to 8.11 mg Tordon 22K/L (equivalent to 1.95 mg/L picloram acid) respirable air for 4 hours (MMAD = 1.74 m). No clinical signs were noted during exposure or during the two-week post exposure period, body weights were not significantly different from unexposed controls, and no gross abnormalities were noted on necropsy (McGuirk and Cieszlak 1996). These results are consistent with an earlier report of no toxic effects resulting from acute inhalation exposure to aerosolized picloram acid (Streeter et al. 1987a). Streeter et al. (1987a) exposed F344 rats (5/sex) to a time-weighted average concentration of 35.1 mg/m³ picloram acid for 4 hours (the highest aerosol concentration attainable; MMAD = 7.96μ , GSD = 3.59). All animals appeared normal during the two-week post exposure observation period, with one rat showing porphyrin staining around its nares on day 2. Body weights were not significantly different from unexposed controls, and no gross abnormalities were noted on necropsy (Streeter et al. 1987a). Tordon K salt liquor (38.8% a.i., 33.4% a.e.) caused a transient decrease in body weight in male rats after a four hour exposure to a concentration of 1.63 mg/L as formulation (0.54 mg a.e./L). No significant decrease in body weight were noted in female rats (Streeter et al. 1987b). Similarly, inhalation exposure to a nominal concentration of 18.3 mg/L Tordon 22 K (with a TWA exposure concentration of 0.65 mg/L) resulted in only transients decreases in body weight in male and female rats (Streeter et al. 1988).

3.1.14. Adjuvants. As indicated in Section 2, the commercial formulation of picloram used by the Forest Service is in the form of the potassium salt of picloram. Both of the Tordon formulations also contain Polyglycol 26-2 (CAS No. 069029-39-6) (C&P Press 1998). This compound is classified by the U.S. EPA (2003) as a List 3 inert. In other words, there is insufficient information to categorize this compound as either hazardous (Lists 1 or 2) or
non-toxic (List 4). Notwithstanding this classification, surfactants are surface active agents that can disrupt cellular membranes and lead to a number of different adverse effects (e.g., Warisnoicharoen et al. 2003). In an *in vitro* study on oxidative phosphorylation in submitochondrial particles derived from a marine algae, Oakes and Pollak (1999) noted that a commercial preparation of 2,4-D and picloram that contained Polyglycol 26-2 as well as Polyglycol 26-2 both inhibited oxidative function in the submitochondrial preparations at a concentration of about 0.01%. While this study clearly indicates that Polyglycol 26-2 will impact mitochondrial function *in vitro*, the implications for potential effects in humans at plausible levels of exposure are not apparent.

Other inerts used in Tordon K and Tordon 22K have been publicly disclosed by Northwest Coalition for Alternatives to Pesticides (<u>http://www.pesticide.org/FOIA/picloram.html</u>). These include emulsified silicone oil (CAS No. 63148-62-9), ethoxylated cetyl ether (CAS No. 9004-95-9), and potassium hydroxide (CAS No. 1310-58-3). All of these compounds are classified by U.S. EPA (2003) as List 4B, inerts of minimal concern. Potassium hydroxide is a GRAS (generally recognized as safe) compound and is approved as an indirect food additive (Clydesdale 1997). Both formulations also contain water as an inert.

The limited toxicity data on formulations of picloram do not suggest any substantial differences between the toxicity of the formulations and the toxicity of picloram when expressed in units of acid equivalents. Dow Chemical Co. (1970) specifically compared the acute oral toxicity of picloram (98.5% a.e.) to a Tordon formulation (22% a.e.). The acute oral LD₅₀ in rats for the formulation was 8.2 mg a.e./kg and the corresponding LD₅₀ for the formulation is given as approximately 10 mg a.e./kg (Dow Chemical Co. 1970).

3.1.15. Impurities and Metabolites. Virtually no chemical synthesis yields a totally pure product. Technical grade picloram, as with other technical grade products, undoubtedly contains some impurities. To some extent, concern for impurities in technical grade picloram is reduced by the fact that the existing toxicity studies on picloram were conducted with the technical grade product. Thus, if toxic impurities are present in the technical grade product, they are likely to be encompassed by the available toxicity studies on the technical grade product.

An exception to this general rule involves carcinogens, most of which are presumed to act by non-threshold mechanisms. Because of the non-threshold assumption, any amount of a carcinogen in an otherwise non-carcinogenic mixture may pose a carcinogenic risk. This is the situation with picloram. As indicated in Section 2, technical grade picloram contains hexachlorobenzene. Nominal or average concentrations of hexachlorobenzene are 8 ppm and the maximum concentration is 50 ppm. The U.S. EPA has classified hexachlorobenzene as a probable human carcinogen for which the data are adequate to consider risk quantitatively (U.S. EPA 1997). While a detailed review of hexachlorobenzene is beyond the scope of this risk associated with the use of picloram (ATSDR 2002). This is detailed further in Section 3.3.

As with contaminants, the potential effect of metabolites on a risk assessment is often encompassed by the available *in vivo* toxicity studies under the assumption that the toxicologic consequences of metabolism in the species on which toxicity studies are available will be similar to those in the species of concern (i.e., humans). Uncertainties in this assumption are encompassed by using an uncertainty factor in deriving the RfD (Section 3.3) and may sometimes influence the selection of the study used to derive the RfD.

As reviewed by U.S. EPA (1992b), the metabolism of picloram has been studied in several mammalian species and there is no indication that picloram is extensively metabolized. In the environment, however, picloram may undergo decarboxylation by microorganisms, photolysis, or pyrolysis and this may impact the assessment of the toxicity to some nontarget species (Section 4.1.2.5). There are no studies, however, on the toxicity of this environmental metabolite to mammals.

3.2. EXPOSURE ASSESSMENT

3.2.1. Overview. Exposure assessments are conducted for both workers and members of the general public for the typical application rate of 0.35 lb/acre. The consequences of using the maximum application rate, 1 lb/acre, are discussed in the risk characterization. For both workers and members of the general public, the upper ranges of all acute exposures are below 1 mg/kg and most exposures are much lower. The highest modeled exposure is about 0.7 mg/kg and is associated with the consumption of contaminated water by a child following an accidental spill of picloram into a small pond. The upper ranges of non-accidental acute exposure scenarios for members of the general public are associated with doses from about 0.00002 to 0.07 mg/kg. The highest dose estimates for non-accidental exposure scenarios are associated with the consumption of contaminated water scenarios are associated with the consumption of contaminated with doses from about 0.00002 to 0.07 mg/kg.

General exposure assessments for workers are in the range of exposures modeled for the general public. For workers, three types of application methods are modeled: directed ground, broadcast ground, and aerial. Central estimates of exposure span a relatively narrow range: 0.005 to 0.008 mg/kg. The upper ranges of exposures are also similar for the different groups of workers: 0.03 to 0.05 mg/kg/day. All of the accidental exposure scenarios for workers involve dermal exposures. Because picloram is not readily absorbed across the skin, all of these accidental exposures lead to estimates of dose that are either in the range of or substantially below the general exposure estimates for workers.

Hexachlorobenzene is a contaminant in technical grade picloram. The average concentration of hexachlorobenzene in technical grade picloram is 8 ppm and the maximum concentration is 50 ppm. For all exposure assessments detailed in this risk assessment, the average concentration of 8 ppm is used. The impact of the 50 ppm level is detailed in the risk characterization. Hexachlorobenzene is ubiquitous and persistent in the environment. The major sources of general exposure for the public to hexachlorobenzene involve industrial emissions, proximity to hazardous waste sites, and the consumption of contaminated food. Virtually all individuals are exposed to hexachlorobenzene and virtually all individuals have detectable concentrations of hexachlorobenzene in their bodies. Based on current concentrations of hexachlorobenzene in environmental media and food, daily doses of hexachlorobenzene (i.e., background levels of exposure) are in the range of 0.000001 (1×10^{-6}) mg/kg/day. Based on the amount of hexachlorobenzene in picloram and the amount of picloram used in Forest Service programs, the use of picloram by the Forest Service will not substantially contribute to any wide-spread increase of ambient levels of hexachlorobenzene. Nonetheless, the potential impact of local contamination is considered for workers as well as for several acute and chronic exposure scenarios for members of the general public. For both workers, the upper range of longer term exposure scenarios result in dose estimates of about 2×10^{-7} mg/kg/dav to 4×10^{-7} mg/kg/dav. below general background levels of exposure by about a factor of 2 to 5. For members of the general public, the upper range of longer term exposure scenarios are about 1×10^{-10} mg/kg/day to 2×10^{-8} mg/kg/day, below general background levels of exposure by about a factor of 50 to 10,000. The upper range of estimated doses associated with acute exposure scenarios for both

workers and members of the general public are about 0.002 mg/kg/day, higher than background levels of exposure by about a factor of 2000.

3.2.2. Workers. The Forest Service uses a standard set of exposure assessments in all risk assessment documents. While these exposure assessments vary depending on the characteristics of the specific chemical as well as the relevant data on the specific chemical, the organization and assumptions used in the exposure assessments are standard and consistent. All of the exposure assessments for worker as well as members of the general public are detailed in the worksheets on picloram that accompany this risk assessment [WPWS 03-43-16-01]. Detailed documentation for these worksheets is presented in presented in SERA WSD 01-2.04, *Documentation for Worksheets Version 2.04 - Human Health and Ecological Risk Assessments*, dated February 25, 2003. A copy of this documentation is available at <u>www.sera-inc.com</u>. This section on workers and the following section on the general public provides are plain verbal description of the worksheets and discuss picloram specific data that are used in the worksheets.

A summary of the exposure assessments for workers is presented in Worksheet E02 of the worksheets for picloram that accompany this risk assessment. Two types of exposure assessments are considered: general and accidental/incidental. The term *general* exposure assessment is used to designate those exposures that involve estimates of absorbed dose based on the handling of a specified amount of a chemical during specific types of applications. The accidental/incidental exposure scenarios involve specific types of events that could occur during any type of application. The exposure assessments developed in this section as well as other similar assessments for the general public (Section 3.2.3) are based on the typical application rate of 0.35 lbs a.e./acre (Section 2). The consequences of using different application rates in the range considered by the Forest Service are discussed further in the risk characterization (Section 3.4).

3.2.2.1. General Exposures – As described in SERA (2001), worker exposure rates are expressed in units of mg of absorbed dose per kilogram of body weight per pound of chemical handled. Based on analyses of several different pesticides using a variety of application methods, default exposure rates are estimated for three different types of applications: directed foliar (backpack), boom spray (hydraulic ground spray), and aerial. As described in SERA (2001), the ranges of estimated occupational exposure rates vary substantially among individuals and groups, (i.e., by a factor of 50 for backpack applicators and a factor of 100 for mechanical ground sprayers).

The specific assumptions used for each application method are detailed in worksheets C01a (directed foliar), C01b (broadcast foliar), and C01c (aerial). In these worksheets, the central estimate of the amount handled per day is calculated as the product of the central estimates of the acres treated per day and the application rate. The ranges for the amounts handled per day are calculated as the product of the range of acres treated per day and the application rate. Similarly, the central estimate of the daily absorbed dose is calculated as the product of the central estimate of the amount handled per day. The ranges of the

daily absorbed dose are calculated as the range of exposure rates and the range for the amounts handled per day. The lower and upper limits are similarly calculated using the lower and upper ranges of the amount handled, acres treated per day, and worker exposure rate.

Two studies (Lavy et al. 1987; Libich et al. 1984) have been conducted on workers handling picloram that permit an estimate of worker exposure rates in terms of absorbed dose (mg/kg body weight per lb a.e. handled) and both of these studies were used to develop the exposure estimates given in SERA (2001). By far the most detailed study on worker exposure to picloram is that conducted by Lavy et al. (1987). In this study, the uptake of 2,4-D, picloram, and dichlorprop was assayed in four groups of forestry workers using four different application methods: backpack, injection bar, hypohatchet, and hack-and-squirt. In addition, for each method, uptake was studied under standard work practices (referred to as T1 in this publication) and work practices involving special precautions (referred to as T2 in this publication). The special precautions involved the use of new gloves for mixing and application, improved personal hygiene, and exposure avoidance. Absorption of the herbicides was assayed using 5 day complete urine collections. In another study, Libich et al. (1984) studied the exposure of herbicide applicators involved in electric power transmission rights-of-way maintenance to 2,4-D, dichlorprop, and picloram. Absorbed dose was estimated from daily urine sampling rather than total urine collection. Two application methods were examined: spray guns mounted on vehicles and mist blowers connected to a back pack. The spray guns were mounted either on trucks—for roadside spraying—or all terrain vehicles (ATV's)—for spraying less accessible areas. The herbicides used were Tordon 101, a formulated 4:1 mixture of 2,4-D and picloram (463 g/L) and a 1:1 mixture of 2,4-D and dichlorprop (480 g/L). For spray gun applications, the commercial product was diluted with 100 parts water. For the backpack application, the product was diluted with 16 parts water. A limitation in the comparison of this study with the study by Lavy et al. (1987) is that Libich et al. (1984) do not specify the amount of product handled. The ranges of estimated occupational exposure rates vary substantially among individuals and groups, (i.e., by a factor of 50 for backpack applicators and a factor of 100 for mechanical ground sprayers). It seems that much of the variability can be attributed to the hygienic measures taken by individual workers (i.e., how careful the workers are to avoid unnecessary exposure).

An estimate of the number of acres treated per hour is needed to apply these worker exposure rates. These values are taken from previous USDA risk assessments (USDA 1989a,b,c). The number of hours worked per day is expressed as a range, the lower end of which is based on an 8-hour work day with 1 hour at each end of the work day spent in activities that do not involve herbicide exposure. The upper end of the range, 8 hours per day, is based on an extended (10-hour) work day, allowing for 1 hour at each end of the work day to be spent in activities that do not involve herbicide exposure.

It is recognized that the use of 6 hours as the lower range of time spent per day applying herbicides is not a true lower limit. It is conceivable and perhaps common for workers to spend much less time in the actual application of a herbicide if they are engaged in other

activities. Thus, using 6 hours may overestimate exposure. In the absence of any published or otherwise documented work practice statistics to support the use of a lower limit, this approach is used as a protective assumption.

The range of acres treated per hour and hours worked per day is used to calculate a range for the number of acres treated per day. For this calculation as well as others in this section involving the multiplication of ranges, the lower end of the resulting range is the product of the lower end of one range and the lower end of the other range. Similarly, the upper end of the resulting range is the product of the upper end of one range and the upper end of the other range. This approach is taken to encompass as broadly as possible the range of potential exposures.

The central estimate of the acres treated per day is taken as the arithmetic average of the range. Because of the relatively narrow limits of the ranges for backpack and boom spray workers, the use of the arithmetic mean rather than some other measure of central tendency, like the geometric mean, has no marked effect on the risk assessment.

3.2.2.2. Accidental Exposures – Typical occupational exposures may involve multiple routes of exposure (i.e., oral, dermal, and inhalation); nonetheless, dermal exposure is generally the predominant route for herbicide applicators (Ecobichon 1998; van Hemmen 1992). Typical multi-route exposures are encompassed by the methods used in Section 3.2.2.1 on general exposures. Accidental exposures, on the other hand, are most likely to involve splashing a solution of herbicides into the eyes or to involve various dermal exposure scenarios.

As summarized in Section 3.1.11, picloram can produce substantial but transient eye injury and both picloram acid and the potassium salt of picloram are classified as moderate eye irritants. The available literature does not include quantitative methods for characterizing exposure or responses associated with splashing a solution of a chemical into the eyes; furthermore, there appear to be no reasonable approaches to modeling this type of exposure scenario quantitatively. Consequently, accidental exposure scenarios of this type are considered qualitatively in the risk characterization (Section 3.4).

There are various methods for estimating absorbed doses associated with accidental dermal exposure (U.S. EPA/ORD 1992, SERA 2001). Two general types of exposure are modeled: those involving direct contact with a solution of the herbicide and those associated with accidental spills of the herbicide onto the surface of the skin. Any number of specific exposure scenarios could be developed for direct contact or accidental spills by varying the amount or concentration of the chemical on or in contact with the surface of the skin and by varying the surface area of the skin that is contaminated.

For this risk assessment, two exposure scenarios are developed for each of the two types of dermal exposure, and the estimated absorbed dose for each scenario is expressed in units of mg chemical/kg body weight. Both sets of exposure scenarios are summarize in Worksheet E01, which references other worksheets in which the specific calculations are detailed.

Exposure scenarios involving direct contact with solutions of the chemical are characterized by immersion of the hands for 1 minute or wearing contaminated gloves for 1 hour. Generally, it is not reasonable to assume or postulate that the hands or any other part of a worker will be immersed in a solution of a herbicide for any period of time. On the other hand, contamination of gloves or other clothing is quite plausible. For these exposure scenarios, the key element is the assumption that wearing gloves grossly contaminated with a chemical solution is equivalent to immersing the hands in a solution. In either case, the concentration of the chemical in solution that is in contact with the surface of the skin and the resulting dermal absorption rate are essentially constant.

For both scenarios (the hand immersion and the contaminated glove), the assumption of zero-order absorption kinetics is appropriate. Following the general recommendations of U.S. EPA/ORD (1992), Fick's first law is used to estimate dermal exposure. As discussed in Section 3.1.3, an experimental dermal permeability coefficient (Kp) for picloram is not available. Thus, the Kp for picloram is estimated using the algorithm from U.S. EPA/ORD (1992), which is detailed in Worksheet A07b. The application of this algorithm to picloram, based on molecular weight and the $k_{o/w}$, is given in Worksheet B04.

Exposure scenarios involving chemical spills onto the skin are characterized by a spill on to the lower legs as well as a spill on to the hands. In these scenarios, it is assumed that a solution of the chemical is spilled on to a given surface area of skin and that a certain amount of the chemical adheres to the skin. The absorbed dose is then calculated as the product of the amount of the chemical on the surface of the skin (i.e., the amount of liquid per unit surface area multiplied by the surface area of the skin over which the spill occurs and the concentration of the chemical in the liquid) the first-order absorption rate, and the duration of exposure. As summarized in Section 3.1.3, the first-order dermal absorption rates are taken from the study by Nolan et al. (1984): an average value of 0.00014 hour⁻¹ with a range of 0.00003 hour⁻¹ to 0.0004 hour⁻¹. These values are included in Worksheet B05 rather than the values calculated in Worksheet B03, which are based on molecular weight and the $k_{o/w}$ for picloram.

For both scenarios, it is assumed that the contaminated skin is effectively cleaned after 1 hour and the mg of absorbed dose is divided by body weight (kg) to yield an estimated dose in units of mg chemical/kg body weight.

3.2.3. General Public

3.2.3.1. *General Considerations* – Under normal conditions, members of the general public should not be exposed to substantial levels of picloram. Nonetheless, any number of exposure scenarios can be constructed for the general public, depending on various assumptions regarding application rates, dispersion, canopy interception, and human activity. Several scenarios are developed for this risk assessment which should tend to over-estimate exposures in general.

The two types of exposure scenarios developed for the general public include acute exposure and longer-term or chronic exposure. All of the acute exposure scenarios are primarily accidental.

They assume that an individual is exposed to the compound either during or shortly after its application. Specific scenarios are developed for direct spray, dermal contact with contaminated vegetation, as well as the consumption of contaminated fruit, water, and fish. Most of these scenarios should be regarded as extreme, some to the point of limited plausibility. The longer-term or chronic exposure scenarios parallel the acute exposure scenarios for the consumption of contaminated fruit, water, and fish but are based on estimated levels of exposure for longer periods after application.

The exposure scenarios developed for the general public are summarized in Worksheet E02. As with the worker exposure scenarios, details of the assumptions and calculations involved in these exposure assessments are given in the worksheets that accompany this risk assessment (Worksheets D01–D09). The remainder of this section focuses on a qualitative description of the rationale for and quality of the data supporting each of the assessments.

3.2.3.2. *Direct Spray* – Direct sprays involving ground applications are modeled in a manner similar to accidental spills for workers (see Section 3.2.2.2.). In other words, it is assumed that the individual is sprayed with a solution containing the compound and that an amount of the compound remains on the skin and is absorbed by first-order kinetics. For these exposure scenarios, it is assumed that during a ground application, a naked child is sprayed directly with picloram. These scenarios also assume that the child is completely covered (that is, 100% of the surface area of the body is exposed). These exposure scenarios are likely to represent upper limits of plausible exposure. An additional set of scenarios are included involving a young woman who is accidentally sprayed over the feet and legs. For each of these scenarios, some assumptions are made regarding the surface area of the skin and body weight, as detailed in Worksheet A04.

3.2.3.3. Dermal Exposure from Contaminated Vegetation – In this exposure scenario, it is assumed that the herbicide is sprayed at a given application rate and that an individual comes in contact with sprayed vegetation or other contaminated surfaces at some period after the spray operation. For these exposure scenarios, some estimates of dislodgeable residue and the rate of transfer from the contaminated vegetation to the surface of the skin must be available. No such data are available on dermal transfer rates for picloram and the estimation methods of Durkin et al. (1995) are used as defined in Worksheet D02. The exposure scenario assumes a contact period of one hour and assumes that the chemical is not effectively removed by washing for 24 hours. Other estimates used in this exposure scenario involve estimates of body weight, skin surface area, and first-order dermal absorption rates, as discussed in the previous section.

3.2.3.4. *Contaminated Water* – Water can be contaminated from runoff, as a result of leaching from contaminated soil, from a direct spill, or from unintentional contamination from aerial applications. For this risk assessment, the two types of estimates made for the concentration of picloram in ambient water are acute/accidental exposure from an accidental spill and longer-term exposure to picloram in ambient water that could be associated with the application of this compound to a 10 acre block that is adjacent to and drains into a small stream or pond.

3.2.3.4.1. ACUTE EXPOSURE – Two exposure scenarios are presented for the acute consumption of contaminated water: an accidental spill into a small pond (0.25 acres in surface area and 1 meter deep) and the contamination of a small stream by runoff or percolation.

The accidental spill scenario assumes that a young child consumes contaminated water shortly after an accidental spill into a small pond. The specifics of this scenarios are given in Worksheet D05. Because this scenario is based on the assumption that exposure occurs shortly after the spill, no dissipation or degradation of picloram is considered. This scenario is dominated by arbitrary variability and the specific assumptions used will generally overestimate exposure. The actual concentrations in the water would depend heavily on the amount of compound spilled, the size of the water body into which it is spilled, the time at which water consumption occurs relative to the time of the spill, and the amount of contaminated water that is consumed. Based on the spill scenario used in this risk assessment, the concentration of picloram in a small pond is estimated to range from about 0.3 mg/L to 6 mg/L with a central estimate of about 1 mg/L (Worksheet D05).

The other acute exposure scenario for the consumption of contaminated water involves runoff into a small stream. Two monitoring studies are available in which picloram has been detected in ambient water shortly after the application of picloram at a known amount (Davis and Ingebo 1973; Michael and Neary 1993). Michael and Neary (1993) summarize monitoring data on the concentrations of picloram in surface water after the application of picloram by injection, broadcast ground, and broadcast aerial applications. Normalized for application rate, the reported peak concentrations of picloram in water were about 2 µg/L to 50 µg/L per lb/acre applied (Table 3-1). The injection data is also summarized in Michael et al. (1994) and the ground and aerial application data are detailed further in Neary et al. (1993). Both the ground and aerial broadcast application involved pellet formulations and the ground application involved a 140 meter buffer. In another study using an application rate of 10.4 kg/ha (9.3 lb/acre), the maximum concentration noted in stream water draining from the watershed was 370 µg/L, which occurred after a 6.4 cm (about 2.5 inches) rainfall (Davis and Ingebo 1973). This is equivalent to 40 µg/L per lb/acre (370 µg/L÷9.3 lb/acre). In a study by Watson et al. (1989), no picloram was detected in streams, at a limit of detection of 0.5 μ g/L, after the application of picloram at rates of 0.28 kg a.e./ha (about 0.25 lb/acre) or 1.12 kg a.e./ha (1 lb/acre) in areas with loam or sandy loam soil.

While monitoring data provide practical and documented instances of water contamination, monitoring studies may not encompass a broad range of conditions which may occur during program applications – e.g., extremely heavy rainfall – or they may reflect atypical applications that do not reflect program practices. Consequently, for this component of the exposure assessment, the monitored levels in ambient water are compared to modeled estimates based on GLEAMS (Groundwater Loading Effects of Agricultural Management Systems). GLEAMS is a root zone model that can be used to examine the fate of chemicals in various types of soils under different meteorological and hydrogeological conditions (Knisel and Davis 2000). As with many environmental fate and transport models, the input and output files for GLEAMS can be

complex. The general application of the GLEAMS model and the use of the output from this model to estimate concentrations in ambient water are detailed in SERA (2003).

For the current risk assessment, the application site was assumed to consist of a 10 acre square area that drained directly into a small pond or stream. The chemical specific values as well as the details of the pond and stream scenarios used in the GLEAMS modeling are summarized in Table 3-2. The GLEAMS modeling yielded estimates picloram runoff, sediment and percolation that were used to estimate concentrations in the stream adjacent to a treated plot, as detailed in Section 6.4 of SERA (2003). The results of the GLEAMS modeling for the small stream are summarized in Table 3-3 and the corresponding values for the small pond are summarized in Table 3-4. These estimates are expressed as both average and maximum water contamination rates (WCR) - i.e., the concentration of the compound in water in units of mg/L normalized for an application rate of 1 lb a.e./acre.

As indicated in Table 3-3, no stream contamination is estimated in very arid regions – i.e., annual rainfall of 10 inches of less. The modeled maximum concentrations in the stream range from about 10 μ g/L to nearly 200 μ g/L at annual rainfall rates from 15 to 250 inches per year, with the highest concentrations associated with clay. While not detailed in Table 3-3, the losses from clay are associated almost exclusively with runoff (about 90%), with the remaining amount due to sediment loss. For sand, the pesticide loss is associated almost exclusively with percolation. For both clay and sand, the maximum losses occur with the first rainfall after application. For loam, most of the pesticide loss is associated with percolation but time to maximum loss is attenuated due to slower transport in loam and the total loss is less than from sandy soil.

The modeled maximum concentrations are generally consistent with the available monitoring data. As noted above, the WCR of 40 μ g/L per lb/acre reported by Davis and Ingebo (1973) occurred after a rainfall of about 2.5 inches. In terms of the GLEAMS modeling, in which rainfall is assumed to occur every tenth day, 2.5 inches of rain corresponds to an annual rainfall rate of about 90 inches. As noted in Table 3-3, the modeled maximum concentrations range from about 10 μ g/L to 184 μ g/L. The water contamination rates derived from Michael and Neary (1993) – i.e., about 2 ppb to 50 µg/L per lb/acre applied (Table 3-1) – are in the range of modeled concentrations for rainfall rates of 15 inches to 50 inches per year. The GLEAMS modeling is also consistent with the study by Watson et al. (1989), in which no picloram was detected in streams at a limit of detection of 0.5 μ g/L after the application of picloram at a rate of 0.25 lb/acre in loam and 1 lb/acre in sandy loam soil. At the former site (0.25 lb/acre), a total of 431 mm of rain (about 17 inches) occurred over 1 year after application. As indicated in Table 3-3, the expected maximum concentration in the stream at 20 inches per year would be 0.6 µg/L (2.47 μ g/L per lb/acre× 0.25 lb/acre) and the average expected concentration would be only about $0.015 \,\mu\text{g/L} (0.06 \,\mu\text{g/L} \text{ per lb/acre} \times 0.25 \,\text{lb/acre})$. Thus, even if a sample was taken immediately after a rain, the expected concentration would be very close to the limit of detection. Watson et al. (1989) provide details on rainfall at the 1 lb/acre site (sandy loam) but indicate that only 3 mm of rain (about 0.01 inch) fell on day 3 after application and that there was insufficient "...rain to leach picloram into soil or sand" by day 7 after application. In addition, Watson et al. (1989)

indicate that this site was more heavily covered by a canopy of vegetation that may have further reduced soil percolation and subsequent contamination of the adjacent stream.

The GLEAMS scenarios do not specifically consider the effects of accidental direct spray. For example, the steam modeled using GLEAMS is about 6 feet wide and it is assumed that the herbicide is applied along a 660 foot length of the stream with a flow rate of 4,420,000 L/day. At an application rate of 1 lb/acre, accidental direct spray onto the surface of the stream would deposit about 41,252,800 μ g [1 lb/acre = 112,100 μ g/m², 6'x660' = 3960 ft² = 368 m², 112,100 μ g/m² × 368 m² = 41,252,800 μ g]. This would result in a downstream concentration of about 10 μ g/L [41,252,800 μ g/day ÷ 4,420,000 L/day]. As indicated in Table 3-3, the expected peak concentrations from runoff or percolation are generally in excess of 10 μ g/L and then encompass the potential effects of accidental direct spray.

For the the current risk assessment, the upper range for the short-term water contamination rate will be taken as 200 μ g/L per lb/acre, this somewhat higher than the maximum concentration at an annual rainfall rate 200 inches for clay soil and about four times the concentration of the upper range of values derived from the data in Table 3-1 (Michael and Neary 1993). This value, converted to 0.2 mg/L per lb/acre, is entered into Worksheet B06. The central estimated will be taken as 100 μ g/L (0.1 mg/L), about the maximum concentration for clay at an annual rainfall rate of 50 inches. The lower range will be taken as 10 μ g/L (0.01 mg/L), concentrations that might be expected in relatively arid regions – i.e., annual rainfall of 15 inches.

3.2.3.4.2. LONGER-TERM EXPOSURE – The scenario for chronic exposure to picloram from contaminated water is detailed in worksheet D07. This scenario assumes that an adult (70 kg male) consumes contaminated ambient water from a contaminated pond for a lifetime. The estimated concentrations in pond water are based both the modeled estimates from GLEAMS, discussed in the previous section, as well as NAWQA monitoring data.

The National Water Quality Assessment (NAWQA) of the U.S. Geological Survey (USGS) has involved a large scale monitoring effort to characterize pesticides in surface and ground water. A detailed description of the USGS program may be obtained at <u>http://water.usgs.gov/nawqa/index.html</u>. In brief, the USGS has monitored concentrations of a large number of pesticides, including picloram, in over 50 major river basins and aquifers. The monitoring data are given separately for streams and ground water for three types of sites: agricultural land use areas, urban areas, and major aquifers or large rivers of streams. Detailed data for rivers and streams covering a period from 1992 to 1998 are available at <u>http://ca.water.usgs.gov/pnsp/</u> and corresponding data covering a period from 1992 to 1996 is available at <u>http://ca.water.usgs.gov/pnsp/allsum/#t1</u>. Over the period from 1992 to 1998, picloram was detected in streams at a maximum concentration of 0.01 µg/L. The frequency of detection is extremely low . Picloram was not detected in any of the agricultural stream sites or in 31 large rivers and streams and was found in only 1 or 573 samples from 22 urban streams. Higher concentrations have been found in ground water, with a maximum concentration of 3.91 µg/L. Again, the frequency of detection is very

low: 0.17% of for agricultural land use wells with a maximum concentration of 2.2 μ g/L, 0.5% for urban land use wells with a maximum concentration of 3.91 μ g/L, and 0.14% for major aquifers with a maximum concentration of 0.17 μ g/L.

The the NAWQA data and the GLEAMS modeling are not directly comparable. The NAWQA data may be viewed as general exposure levels that are not directly associated with a fixed application while the GLEAMS modeling is an effort to characterize, at least generically, concentrations that might be expected after applications associated with Forest Service programs. Nonetheless, the maximum concentration monitored in NAWQA, 0.01 μ g/L, suggests that applications of picloram near streams may result in average yearly concentrations that are substantially higher – i.e., about 0.1 to 1 μ g/L – than those generally found in the environment. On the other hand, while the application of GLEAMS does not specifically model groundwater, the average modeled concentrations in lakes – i.e., about 0.6 to 5 μ g/L from Table 3-4 – are reasonably consistent with the maximum concentrations reported in NAWQA for groundwater – i.e., about 2 to 4 μ g/L.

For this risk assessment, the typical WCR is taken as $1 \mu g/L$ or 0.001 mg/L per lb/acre. This is about the average concentration that modeled in lake using GLEAMS at a rainfall rate of 100 inches per year in clay soil. The upper range of the WCR is taken as $4 \mu g/L$ or 0.004 mg/L per lb/acre. This is the highest average concentration modeled from sandy soil and is identical to the highest concentration in groundwater from NAWQA. The lower range is taken as $0.1 \mu g/L$ or 0.0001 mg/L per lb/acre. This selection is somewhat arbitrary but would tend to encompass concentrations near the limit of detection that might be found in relatively arid areas.

The WCR values discussed in this section summarized in Worksheet B06 and used for all longer term exposure assessments involving contaminated water. As with the corresponding values for a small stream, these estimates are expressed as the water contamination rates (WCR) in units of mg/L per lb/acre.

3.2.3.5. Oral Exposure from Contaminated Fish -- Many chemicals may be concentrated or partitioned from water into the tissues of animals or plants in the water. This process is referred to as bioconcentration. Generally, bioconcentration is measured as the ratio of the concentration in the organism to the concentration in the water. For example, if the concentration in the organism is 5 mg/kg and the concentration in the water is 1 mg/L, the bioconcentration factor (BCF) is 5 L/kg [5 mg/kg \div 1 mg/L]. As with most absorption processes, bioconcentration depends initially on the duration of exposure but eventually reaches steady state. Details regarding the relationship of bioconcentration factor to standard pharmacokinetic principles are provided in Calabrese and Baldwin (1993).

Bidlack (1980a) attempted to measure the bioconcentration factor of ¹⁴C-picloram in bluegill sunfish over a 28 day exposure period at concentrations of 0.1 and 1 mg/L in water. Only trace amounts of ¹⁴C were recovered in fish and Bidlack (1980a) concluded that the bioconcentration factor of picloram is less than one. Similar results were obtained with channel catfish in an

aquatic microcosm study (Bidlack 1980b). Thus, the bioconcentration factor for picloram appears to be substantially less than one (USDA 1989d; U.S. EPA 1995b). For this risk assessment a bioconcentration factor of 1 L/kg is used. This assumption will over-estimate exposure but has no substantial impact on the risk assessment (Section 3.4).

For both the acute and longer-term exposure scenarios involving the consumption of contaminated fish, the water concentrations of picloram used are identical to the concentrations used in the contaminated water scenarios (see Section 3.2.3.4). The acute exposure scenario is based on the assumption that an adult angler consumes fish taken from contaminated water shortly after an accidental spill of 200 gallons of a field solution into a pond that has an average depth of 1 m and a surface area of 1000 m² or about one-quarter acre. No dissipation or degradation is considered. Because of the available and well-documented information and substantial differences in the amount of caught fish consumed by the general public and native American subsistence populations (U.S. EPA 1996), separate exposure estimates are made for these two groups, as illustrated in worksheet D08. The chronic exposure scenario is constructed in a similar way, as detailed in worksheet D09, except that estimates of picloram concentrations in ambient water are based on GLEAMS modeling as discussed in Section 3.2.3.4.

3.2.3.6. Oral Exposure from Contaminated Vegetation – None of the Forest Service applications of picloram will involve the treatment of crops. Thus, under normal circumstances and in most types of applications conducted as part of Forest Service programs, the consumption by humans of vegetation contaminated with picloram is unlikely. Nonetheless, any number of scenarios could be developed involving either accidental spraying of crops or the spraying of edible wild vegetation, like berries. In most instances, and particularly for longer-term scenarios, treated vegetation would probably show signs of damage from exposure to picloram (Section 4.3.2.4), thereby reducing the likelihood of consumption that would lead to significant levels of human exposure. Notwithstanding that assertion, it is conceivable that individuals could consume contaminated vegetation. One of the more plausible scenarios involves the consumption of contaminated berries after treatment of a right-of-way or some other area in which wild berries grow.

The two accidental exposure scenarios developed for this exposure assessment include one scenario for acute exposure, as defined in Worksheet D03 and one scenario for longer-term exposure, as defined in Worksheet D04. In both scenarios, the concentration of picloram on contaminated vegetation is estimated using the empirical relationships between application rate and concentration on vegetation developed by Fletcher et al. (1994) which is in turn based on a re-analysis of data from Hoerger and Kenaga (1972). These relationships are defined in worksheet A04. For the acute exposure scenario, the estimated residue level is taken as the product of the application rate and the residue rate (Worksheet D03).

For the longer-term exposure scenario (D04), a duration of 90 days is used. The rate of decrease in the residues over time is taken form the vegetation half-time of 8 days reported by Knisel and Davis (2000). Although the duration of exposure of 90 days is somewhat arbitrarily chosen, this

duration is intended to represent the consumption of contaminated fruit that might be available over one season. Longer durations could be used for certain kinds of vegetation but would lower the estimated dose (i.e., would reduce the estimate of risk).

For the longer-term exposure scenarios, the time-weighted average concentration on fruit is calculated from the equation for first-order dissipation. Assuming a first-order decrease in concentrations in contaminated vegetation, the concentration in the vegetation at time t after spray, C_t , can be calculated based on the initial concentration, C_{θ_t} as:

$$C_t = C_0 \times e^{-kt}$$

where k is the first-order decay coefficient $[k=ln(2)\div t_{50}]$. Time-weighted average concentration (C_{TWA}) over time *t* can be calculated as the integral of C_t (De Sapio 1976, p. p. 97 ff) divided by the duration (*t*):

$$\mathbf{C}_{\mathrm{TWA}} = C_{\theta} \left(1 - \mathrm{e}^{\mathrm{-k} t} \right) \div (\mathbf{k} t).$$

A separate scenario involving the consumption of contaminated vegetation by drift rather than direct spray is not developed in this risk assessment. As detailed further in Section 3.4, this elaboration is not necessary because the direct spray scenario leads to estimates of risk that are below a level of concern. Thus, considering spray drift and a buffer zone quantitatively would have no impact on the characterization of risk.

3.2.4. Hexachlorobenzene. As mentioned in Section 2.2, technical grade picloram is contaminated with hexachlorobenzene. The average concentration of hexachlorobenzene in technical grade picloram is 8 ppm and the maximum concentration is 50 ppm. For all exposure assessments detailed in this risk assessment, the average concentration of 8 ppm is used. The impact of the 50 ppm level is detailed in the risk characterization.

As discussed in Section 3.1.15, the potential effect of a contaminant on a risk assessment is often encompassed by the available *in vivo* toxicity studies under the assumption that the toxicologic consequences of the contaminant is encompassed by the use of technical grade material – in this case picloram containing hexachlorobenzene. This rationale cannot be applied to hexachlorobenzene, however, because hexachlorobenzene is both more persistent than picloram and because hexachlorobenzene is classified as a carcinogen (Section 3.1.15). Thus, in order to quantitatively consider the potential cancer risk from hexachlorobenzene posed by the use of technical grade picloram in Forest Service programs, separate exposure assessments are required for hexachlorobenzene. Summaries of the exposure assessments for workers and members of the general public are the hexachlorobenzene worksheets that accompany this risk assessment (Supplement 2). All worksheets mentioned in this section refer those in Supplement 2.

The following discussion of the exposure assessments for hexachlorobenzene focuses on aspects of the exposure assessments that differ substantially from those used for picloram.

3.2.4.1. Dermal Absorption – No studies have been encountered on the dermal absorption rate of hexachlorobenzene in humans. In a study using rats, Koizumi (1991) estimated a first-order dermal absorption rate coefficient of 0.0014 hour⁻¹. Based on empirical relationships of molecular weight and the octanol-water partition coefficient to human dermal absorption rates, central estimate of the first-order dermal absorption rate coefficient for hexachlorobenzene is 0.022 hour⁻¹ with a range of about 0.0047 to 0.1 hour⁻¹ (Supplement 2, Worksheet B03). While a case could be made for using the lower dermal absorption rate from Koizumi (1991) because it is based on an experimental measurement, the higher first-order dermal absorption rates from Supplement 2, Worksheet B03, are used in the exposure scenarios involving first-order dermal absorption for both workers (Worksheets C03a, C03b) and members of the general public (D01a, D01b, and D02). This approach is taken because of uncertainties in the application of absorption rate data from rates for exposure assessments in humans.

As with first-order dermal absorption, no measurements of dermal permeability (K_p in cm/hr) in humans have been encountered for hexachlorobenzene. As with picloram, the K_p for hexachlorobenzene is estimated using the algorithm from U.S. EPA/ORD (1992), which is detailed in Worksheet A07b and applied to hexachlorobenzene in Worksheet B04 of Supplement 2.

3.2.4.2. Acute Exposures – For all of the worker exposure assessments as well as the acute exposure assessments for members of the general public, the exposure estimates follow the same general methods used for the picloram exposure assessments, as detailed in Sections 3.2.2 and 3.2.3. The major differences in the exposure assessments for picloram and hexachlorobenzene involve lipophilicity and water solubility. Picloram is highly water soluble (430mg/L, Table 2-1). Consequently, picloram does not partition substantially into fatty tissue (K_{o/w} of about 84 or less) and thus dermal absorption, binding to soil, and bioconcentration of picloram are low compared to hexachlorobenzene.

Hexachlorobenzene, on the other hand, is highly lipophilic. The $K_{o/w}$ of hexachlorobenzene is about 1,500,000 and the water solubility of hexachlorobenzene is only about 0.006 mg/L. Thus, hexachlorobenzene may be readily absorbed across the skin, will bind tightly to most soils, and will bioconcentrate in fish (ATSDR 2002). Although the amount of hexachlorobenzene in technical grade picloram is relatively low, the potential for human exposure, in terms of the proportion of the exposure dose that might be absorbed, is higher than that for picloram itself.

Because of the extremely high lipophilicity and low water solubility of hexachlorobenzene, one adjustment considered in the acute exposure assessments concerns the impact of water solubility on the dermal spill scenarios. As detailed in hexachlorobenzene Worksheets B01 and B02, the calculation of the concentration of a compound, either a herbicide or contaminant, in a solution that is applied in the field is dependent on the concentration of the compound in the formulation as well as the dilution rates for the formulation recommended by the manufacturer. For hexachlorobenzene, the range of concentrations in a field solution based on these rates can be calculated as 0.0000034 mg/mL to 0.000067 mg/mL (Worksheet B01). The upper range exceeds

the water solubility of hexachlorobenzene, which is 0.006 mg/L or 0.000006 mg/mL. Thus, following the dermal exposure guidelines proposed by U.S. EPA (1992a), the functional exposure to hexachlorobenzene would be based on the water solubility of hexachlorobenzene rather than the maximum nominal concentration. For this risk assessment, however, the nominal concentrations are used. This approach is taken both to remain protective and because the presence of adjuvants in the Tordon formulations may increase the solubility of hexachlorobenzene in the formulations and this may result in a higher water solubility of hexachlorobenzene in dilute aqueous solutions of the formulation – i.e., as in an accidental spill.

For acute exposure scenario involving an accidental spill into a small pond (Worksheets D05), both the central estimated and upper range of the concentration of hexachlorobenzene in the field solution also exceed the nominal concentration of hexachlorobenzene in water. As with the dermal exposure scenarios, and for the same reasons, these concentrations are used in the exposure assessment.

As with picloram, both the acute and chronic scenarios for the consumption of fish contaminated with hexachlorobenzene (Worksheets D07 and D10) require estimates of a bioconcentration factor (i.e., the concentration in fish divided by the concentration in water). As reviewed in ATSDR (2002), reported bioconcentration factors in fish range from about 2,000 to 20,000. For this risk assessment, the upper range of these bioconcentration factors is used in the chronic exposure scenarios. The application of a bioconcentration factor of 20,000 to the acute exposure scenario for contaminated fish (hexachlorobenzene D07) is a protective assumption. All of the bioconcentration factors reported in ATSDR (2002) involved exposure periods of at least one month. As detailed by Calabrese and Baldwin (1993, pp. 12–22), the kinetics of bioconcentration in fish are essentially identical to standard pharmacokinetic zero-order absorption and first-order elimination models (e.g., Goldstein et al. 1974). Consequently, for compounds that are extensively bioconcentrated, such as hexachlorobenzene, the levels in fish after one day will reflect bioconcentration factors that are typically much less than those seen after long-term exposures. Thus, for the acute exposure scenarios, the lower range of the bioconcentration factors reported in ATSDR (2002) is used – i.e., a BCF of 2000 L/kg.

3.2.4.3. *General Considerations for Chronic Exposures* – Hexachlorobenzene is ubiquitous and persistent in the environment. The major sources of general exposure for the public to hexachlorobenzene involve industrial emissions, proximity to hazardous waste sites, and the consumption of contaminated food. Virtually all individuals are exposed to hexachlorobenzene and virtually all individuals have detectable concentrations of hexachlorobenzene in their bodies (ATSDR 2002). Based on current concentrations of hexachlorobenzene in environmental media and food, daily doses of hexachlorobenzene (i.e., background levels of exposure) are in the range of 0.000001 mg/kg/day (ATSDR 2002). The major source of hexachlorobenzene release to the environment is from the manufacture of chlorinated solvents which accounts for an annual release of 70,343 to 241,311 kg (154,000 to 532,000 pounds). The presence of hexachlorobenzene as a contaminant in all pesticides containing hexachlorobenzene as a contaminant in all pesticides containing hexachlorobenzene as a contaminant results in the release of about 17,366 kg/year (38,285 lbs/year) (ATSDR 2002). As

detailed below, only a small fraction of this amount is associated with the use of picloram in Forest Service programs.

The use of picloram by the Forest Service is currently about 13,000 lbs/year (Table 2-3) or about 5,900 kg/year [13,000 lbs \times 0.4536 kg/lb = 5896.8 kg]. Given an average concentration of 8 ppm hexachlorobenzene in technical grade picloram, the amount of hexachlorobenzene released to the environment as a result of Forest Service programs using picloram is about 0.05 kg:

 $5900 \text{ kg} \times 0.000008 = 0.0472 \text{ kg}.$

This amount represents a factor of about one in $350,000 (17,366 \div 0.05 = 347,320)$ relative to the amount of hexachlorobenzene released as a contaminant in all pesticides and a fraction of about 1 in 1.4 million (70,343 ÷ 0.05) to 1 in 4.8 million (241,311 ÷ 0.05) compared to the amount released from the manufacture of contaminated solvents. Thus, the use of picloram by the Forest Service will not substantially contribute to any wide-spread increase of ambient levels of hexachlorobenzene.

While the use of picloram by the Forest Service will not result in any general increase in environmental levels of hexachlorobenzene, this does not demonstrate that localized contamination would be insignificant. In order to better assess the potential impact of local contamination, three chronic exposure scenarios are considered quantitatively: contaminated vegetation, contaminated water, and contaminated fish.

3.2.4.4. Chronic Exposures Involving Contaminated Vegetation – Immediately after direct foliar application to vegetation, hexachlorobenzene will volatilize relatively rapidly from the surface of the vegetation and relatively little will be absorbed and available for longer-term exposures. Once hexachlorobenzene is absorbed into the soil column, however, it is relatively persistent, with reported half times in soil ranging from 3 to 6 years (ATSDR 2002). Thus, the primary concern for chronic exposures to contaminated vegetation is soil contamination with subsequent uptake by plants. This type of scenario requires estimates of long-term levels in soil as well as bioconcentration factors for terrestrial plants. The highest bioconcentration factor for the uptake of hexachlorobenzene from soil into plants is 19 (ATSDR 2002). This BCF was measured in the edible portion of carrots and is used directly for this exposure assessment (Worksheets D03 and D04). As illustrated in these worksheets, this bioconcentration factor is multiplied by the concentration of hexachlorobenzene in soil to estimate the concentration of hexachlorobenzene in soil to estimate the concentration to those used for picloram.

GLEAMS is used to estimate the concentration of hexachlorobenzene in soil. As with picloram, GLEAMS simulations were conducted over a wide range of annual rainfall rates in three types of soil: clay, loam, and sand. The chemical and site specific parameters used in the GLEAMS simulations are summarized in Table 3-5. The basic pond and stream scenarios are very similar to the scenarios used for picloram except that a root zone of 12 inches is used for

hexachlorobenzene rather than the 60 inch root zone used for picloram. Most loss of hexachlorobenzene is due to runoff rather than percolation and use of a shallower root zone favors runoff (Knisel and Davis 2000, p. 28). Because of the shallow root zone, only two soil horizons are used, the top 1 inch and the remaining 11 inches. While hexachlorobenzene is extremely persistent in soil once it has become incorporated into the soil, hexachlorobenzene will rapidly volatilize from the soil surface and the relatively short halftime for the upper soil horizon is based on the study by Beall (1976) in which a very rapid decrease in hexachlorobenzene in the upper soil layer (0-2 cm or about 1 inch) was attributed to volatilization. The much longer halftimes for deeper soil layers is taken from a range of soil halftimes reported by ATSDR (2002).

The concentrations of hexachlorobenzene in soil from the GLEAMS simulations are summarized in Table 3-6. As with the corresponding tables for picloram, the concentrations are expressed as contamination rates – concentrations associated with an application rate of 1 lb/acre. Adjustments to these concentrations are made in the worksheets in which they are used.

The maximum concentrations in soil are independent of rainfall and reflect initial concentrations in soil immediately after application. Differences among soils are not remarkable. For the acute exposure assessment involving the consumption of contaminated vegetation (Worksheet D03), the highest concentration, 0.67 ppm, is used uniformly. This has no impact on the characterization of risk.

For the longer-term exposure scenario (Worksheet D04), the central estimate of the soil contamination rate is taken as 0.026 mg/kg soil per lb/acre. This is near the simulated values for all soil types over a wide range of rainfall rates. The upper range is only modestly higher, 0.031 mg/kg soil per lb/acre simulated for sandy soil at annual rainfalls of 200 or 250 inches. The lower range is taken as 0.007 mg/kg soil per lb/acre, the simulated value for clay at an annual rainfall of 250 inches.

For comparison, Beall (1976) monitored hexachlorobenzene in the top sandy loam at a concentration of about 0.1 mg/kg after the application of hexachlorobenzene. Although Beall (1976) does not specify an application rate in units of quantity per unit area, such as lb/acre, Beall specifies that the hexachlorobenzene was applied to yield an initial concentration of 10 mg/kg soil in the top 5 cm of soil. With this information, an approximate application rate can be calculated. A 1 cm² soil surface that is 5 cm deep has a volume of 5 cm³. The soil type used in the Beall (1976) study is specified as sandy loam but detailed soil characteristics are not provided in the publication. Taking a bulk density of 1.6 g/cm³ for sandy loam soil (Knisel and Davis 2000, p. 46), a 5 cm³ volume of soil would weigh 0.008 kg:

$$5 \text{ cm}^3 \times 1.6 \text{g/cm}^3 = 8 \text{ g} = 0.008 \text{ kg}.$$

To achieve a nominal concentration of 10 mg hexachlorobenzene/kg soil, the amount applied to a 1 cm^2 surface of soil would be :

 $0.008 \text{ kg} \times 10 \text{ mg} \text{ HCB/kg soil} = 0.08 \text{ mg} = 80 \mu \text{g}.$

The application rate can be calculated as 80 μ g/cm² or about 7.1 lbs/acre (1.0 lb/acre = 11.21 μ g/cm²):

 $80 \ \mu g/cm^2 \div (11.21 \ \mu g/cm^2 \div 1 \ lb/acre) = 7.136 \ lbs/acre.$

Thus, the soil contamination rate from the study by Beall (1976) is about 0.014 mg/kg per lb/acre [0.1 mg/kg \div 7.136 lbs/acre], only about a factor of 2 less than the average concentration modeled in GLEAMS and within the range of variability in the GLEAMS simulations.

3.2.4.5. *Exposures Involving Contaminated Streams and Ponds* – Immediately after application of a pesticide that is contaminated with hexachlorobenzene to soil or plants, there is not likely to be any immediate contamination of water attributable to the hexachlorobenzene in the contaminated pesticide. Nonetheless, because of the persistence of hexachlorobenzene, it will remain in the soil and could be transferred to surface waters where most of the hexachlorobenzene will be bound to sediments or bioconcentrated in aquatic organisms (ATSDR 2002).

No monitoring studies have been encountered that permit a direct estimate of the amount of hexachlorobenzene that would be found in ambient water as a result of applying a herbicide contaminated with hexachlorobenzene. Nonetheless, there are ample monitoring data to indicate that hexachlorobenzene can, over time, be transported to water either by runoff or by volatilization with subsequent redeposition in rainwater. Because hexachlorobenzene binds tightly to and is relatively immobile in soils, hexachlorobenzene is not likely to percolate through soils and directly contaminate ground water (ATSDR 2002). While volatilization may be an important route of environmental transport, volatilized hexachlorobenzene will be rapidly dispersed and transported over a relatively wide area. Although this will contribute to general background levels of hexachlorobenzene, the amounts of hexachlorobenzene released in Forest Service programs will not substantially contribute to background levels of hexachlorobenzene (Section 3.2.4.2). Consequently, for this risk assessment, the contamination of ambient water is based on estimates of hexachlorobenzene runoff from contaminated soil.

Based on the GLEAMS simulations described in the previous section, concentrations in streams and ponds at various annual rainfall rates and soils are summarized in Tables 3-7 and 3-8. The greatest concentrations of hexachlorobenzene will be from runoff sediment from clay, with lesser concentrations from loam or sand. For acute exposures, a peak concentration rate of 90 μ g/L per lb/acre is used as the central estimate for water in a contaminated stream. This is about the maximum concentration from clay at an annual rainfall of 100 inches (Table 3-7). The upper range is taken as 300 μ g/L per lb/acre, somewhat above the highest concentration from clay at an annual rainfall of 250 inches. The lower range is set somewhat arbitrarily at 1 μ g/L per lb/acre, in the range of concentrations that could be expect from runoff from clay in arid regions. These values are entered into Worksheet B06 and used in all exposure scenarios involving acute exposures to hexachlorobenzene associated with drinking water from a stream.

Longer-term concentrations in streams (Table 3-7) are estimated to be somewhat higher than those in lakes or ponds (Table 3-8). This is because the stream scenario assumes a rocky stream bed and ignores binding to sediment. For this risk assessment, the longer term concentrations are based on the simulations for a stream. The central estimate is taken as $0.5 \ \mu g/L$ per lb/acre, about the simulated concentration from clay at an annual rainfall of 100 inches as well as the the simulated concentration from loam at an annual rainfall of 250 inches. The longer-term concentration is taken as $1 \ \mu g/L$ per lb/acre, somewhat above the simulated concentration from clay at an annual rainfall of 250 inches. The lower limit is again somewhat arbitrarily set at 0.03 $\mu g/L$ per lb/acre. These values are entered into Worksheet B06 and used in all exposure scenarios involving longer term exposures to hexachlorobenzene associated with drinking water from a stream.

3.3. DOSE-RESPONSE ASSESSMENT

3.3.1. Overview. The Office of Pesticide Programs of the U.S. EPA has derived an RfD of 0.2 mg/kg/day for picloram. This RfD is based on a chronic rat NOAEL of 20 mg/kg/day and an uncertainty factor of 100. In the same study, the LOAEL was 60 mg/kg/day and the effect noted was a change in the staining properties of liver cells. No frank signs of toxicity were seen at this or higher dose levels. This NOAEL for chronic toxic effects is below the NOAELs for reproductive effects. Thus, doses at or below the RfD will be below the level of concern for reproductive effects.

The contamination of technical grade picloram with hexachlorobenzene can be quantitatively considered to a limited extent. The U.S. EPA has derived an RfD and cancer potency parameter for hexachlorobenzene. Based on the levels of contamination of technical grade picloram with these compounds and the relative potencies of these compounds to picloram, this contamination is not significant in terms of potential systemic toxic effects. This assessment, however, does not impact the potential carcinogenicity associated with hexachlorobenzene and this risk, based on the U.S. EPA's cancer potency parameter, is quantitatively considered in the risk characterization.

3.3.2. Existing Guidelines for Picloram. The most recent RfD for picloram is 0.2 mg/kg/day, a value derived by the U.S. EPA's Office of Pesticide Programs (U.S. EPA 1994). This RfD is based on a NOAEL of 20 mg/kg/day from a two-year rat feeding study (Landry et al. 1986), discussed in Section 3.1.5, with an uncertainty factor of 100. At doses of 60 and 200 mg/kg/day, changes in the staining properties of liver cells, but no frank signs of toxicity, were noted. This RfD is also cited in the RED for picloram (U.S. EPA 1995b) as well as proposed pesticide tolerances for picloram (U.S. EPA 1999).

An earlier RfD of 0.07 mg/kg/day is listed on IRIS (U.S. EPA 1992c). This RfD is based on a no-observed-effect level (NOEL) of 7 mg/kg/day from a 6-month dog feeding study (Barna-Lloyd et al. 1982), also discussed in Section 3.1.5, and this RfD also was derived using an uncertainty factor of 100. While the U.S. EPA (1994) does not specifically discuss the rationale for adopting the higher RfD of 0.2 mg/kg/day, the two-year rat study was probably preferred because it is a lifetime study whereas a 6-month dog feeding study is substantially less than lifetime. In addition, dogs are generally considered a poor animal model because of a decreased ability to secrete weak acids via the kidney.

The U.S. EPA (1999) has not derived an acute exposure limit for picloram. In discussing the available acute toxicity data, the U.S. EPA (1999) has concluded that:

No toxicological effect that could be attributable to a single oral exposure was identified, and therefore picloram is not expected to present an acute dietary hazard (p. 421, Section C.1.i.).

This statement appears to be based primarily on a review of the acute toxicity studies submitted to the U.S. EPA as part of the registration process. No information has been encountered in the published literature that would contradict EPA's assertion. Nonetheless, it is likely that higher doses could be tolerated after a single acute exposure than would be tolerated in chronic exposures.

A surrogate acute toxicity value could be derived from the available data. For example, the reproductive NOAEL values for picloram range up to 400 mg/kg/day, a factor of 20 above the lifetime NOAEL used to derive the chronic RfD. The lack of an acute RfD, however, has a relatively minor impact on this risk assessment. As discussed in Section 3.4, the only acute exposure scenario that results in a modest increase above the RfD involves the accidental spill into a small pond (3.2.3.4.1). Thus, no surrogate acute toxicity value is proposed in this risk assessment.

3.3.3. Existing Guidelines for Hexachlorobenzene.

3.3.3.1. Systemic Toxicity – The U.S. EPA RfD for hexachlorobenzene is 0.0008 mg/kg/day. This RfD is based on a 130-week feeding study in male and female rats that also included a 90-day exposure to offspring. The U.S. EPA judged the NOAEL for liver effects at a dose of 0.08 mg/kg/day with a LOAEL at 0.29 mg/kg/day. The LOAEL was characterized by U.S. EPA (1997) as "an increase (p<0.05) in hepatic centrilobular basophilic chromogenesis" in the offspring of the chronically exposed rats. As with picloram and for the same reasons as with picloram, the U.S. EPA used an uncertainty factor of 100 to derive the RfD of 0.0008 mg/kg/day.

ATSDR (2002) has derived an acute MRL for hexachlorobenzene of 0.008 mg/kg/day, a factor of 10 above the chronic RfD derived by U.S. EPA. The Office of Drinking Water of the U.S. EPA has derived a maximum contaminant level of 0.001 mg/L of drinking water and a maximum short term health advisory of 0.05 mg/L of drinking water (U.S. EPA 1998a).

3.3.3.2. Carcinogenic Potency – In addition to systemic toxicity, hexachlorobenzene has been shown to cause tumors of the liver, thyroid and kidney in three species of rodents—mice, rats, and hamsters (ExToxNet 1996a; U.S. EPA 1997). Based on a two-year feeding study in rats, the U.S. EPA (1997) derived a cancer slope factor for lifetime exposures of $1.6 \text{ (mg/kg/day)}^{-1}$. The recent review of hexachlorobenzene by ATSDR (2002) reports no new data that would impact this estimate of the cancer potency factor.

Cancer risk over a lifetime (*P*) is calculated as the product of the daily dose (*d*) over a lifetime and the potency parameter (β):

$$P = d\beta$$

and the lifetime daily dose associated with a given risk level is:

 $d = P \div \beta$

Thus, the lifetime daily dose of hexachlorobenzene associated with a risk of one in one-million $(1\div1,000,000 \text{ or } 0.000001)$ is 0.000000625 mg/kg/day:

$$d_{(mg/kg/day)} = 0.000001 \div (1.6 (mg/kg/day)^{-1}).$$

As noted in Section 3.1, picloram is not classified as a carcinogen. While it can be argued that the technical grade picloram used in the standard bioassays encompasses any toxicologic effects that could be caused by hexachlorobenzene, this argument is less compelling for carcinogenic effects because, for most cancer causing agents, the cancer risk is viewed as a non-threshold phenomenon (i.e., zero risk is achieved only at zero dose).

The potency factor of 1.6 (mg/kg/day)⁻¹ is intended to be applied to lifetime daily doses. As summarized in Section 3.2, many of the exposure assessments used in this risk assessment involve much shorter periods of time. Following the approach recommended by U.S. EPA (1997, p. 35), this risk assessment assumes that the average daily dose over a lifetime is the appropriate measure for the estimation of cancer risk.

3.4. RISK CHARACTERIZATION

3.4.1. Overview. Typical exposures to picloram do not lead to estimated doses that exceed a level of concern. For workers, no exposure scenarios, acute or chronic, exceeds the RfD even at the upper ranges of estimated dose. For members of the general public, the upper limits for hazard quotients are below a level of concern except for the accidental spill of a large amount of picloram into a very small pond. Even this exposure scenario results in only a small excursion above the chronic RfD and is not likely to be toxicologically significant, because of the short duration of exposure relative to those considered in the derivation of the RfD. Thus, based on the available information and under the foreseeable conditions of application, there is no route of exposure or scenario suggesting that workers or members of the general public will be at any substantial risk from longer-term exposure to picloram.

Irritation and damage to the eyes can result from exposure to relatively high levels of picloram (i.e., placement of picloram directly onto the eye) and repeated exposures to picloram can lead to skin sensitization. From a practical perspective, eye irritation and skin sensitization are likely to be the only overt effects as a consequence of mishandling picloram. These effects can be minimized or avoided by prudent industrial hygiene practices during the handling and application of picloram.

Based on the standard assumptions used in this and other Forest Service risk assessments, the contamination of picloram with hexachlorobenzene does not appear to present any substantial cancer risk even at the upper ranges of plausible exposure. Administratively, the Forest Service has adopted a cancer risk level of one in one-million $(1 \div 1,000,000)$ as a trigger that would require special steps to mitigate exposure or restrict and possibly eliminate use. Based on relatively protective exposure assumptions and at the typical application rate of 0.35 lb a.e. picloram/acre, the highest cancer risk is about 0.7 in one-million – i.e., for workers involved in broadcast ground spray. At the upper range of the application rate – i.e., 1 lb a.e./acre – this risk would scale to 2 in one-million. This is not, however, an appropriate approach for risk scaling because it would assume that the same worker applies picloram at an atypically high application rate over a lifetime. For members of the general public, the highest cancer risk is estimated at 0.1 in one-million at the typical application rate of 0.35 lb a.e. Scaled to an application rate of 1 lb/acre, the cancer risk would be about 0.3 in one-million.

3.4.2. Workers. A quantitative summary of the risk characterization for workers associated with exposure to picloram is presented in Worksheet E02 (Supplement 1). The quantitative risk characterization is expressed as the hazard quotient, which is the ratio of the estimated doses from Worksheet E01 to the chronic RfD for picloram, 0.2 mg/kg/day (Section 3.3.2). As indicated in Section 2, the hazard quotients in Worksheet E02 are based on the typical application rate of 0.35 lb a.e./acre and the "level of concern" is one – i.e., if the hazard quotient is below 1.0, the exposure is less than the RfD. For all exposure scenarios, the estimated dose scales linearly with application rate. Thus, at an application rate of 1 lb a.e./acre, the highest application rate contemplated by the Forest Service, the level of concern would be 0.35.

Under typical conditions of exposure, none of the exposure scenarios approach a level of concern. Even at the upper limits of exposure which are based on the worst-case exposure assumptions, the hazard indices do not exceed unity for the general worker exposures. In addition, level of concern of 0.35 for the highest application rate is not exceeded, although the upper range of exposures for workers involved in broadcast ground spray leads to a hazard quotient of 0.3, which approaches the level of concern for an application rate of 1 lb a.e./acre. Nonetheless, it should be noted that using the highest application rate to assess chronic risks is implausible in that it assumes that a worker would repeatedly apply picloram at the highest application rate. As indicated in Section 2, this is not likely given the application rates typically used in Forest Service programs.

While the accidental exposure scenarios are not the most severe one might imagine (e.g., complete immersion of the worker or contamination of the entire body surface for a prolonged period of time) they are representative of reasonable accidental exposures. None of these hazard quotients approach a level of concern that the upper ranges. The simple verbal interpretation of this quantitative characterization of risk is that under the most protective set of exposure assumptions, workers would not be exposed to levels of picloram that are regarded as unacceptable so long as reasonable and prudent handling practices are followed.

As discussed in Section 3.1.11, picloram can cause irritation and damage to eyes and skin sensitization. Quantitative risk assessments for irritation are not derived; however, from a practical perspective, effects on the eyes or skin are likely to be the only overt effects as a consequence of mishandling picloram. These effects can be minimized or avoided by prudent industrial hygiene practices during the handling of picloram.

3.4.3. General Public. The quantitative hazard characterization for the general public associated with exposure to picloram is summarized in Worksheet E04 (Supplement 1). Like the quantitative risk characterization for workers, the quantitative risk characterization for the general public is expressed as the hazard quotient using the chronic RfD of 0.2 mg/kg/day for both acute and chronic exposures.

Although there are several uncertainties in the longer-term exposure assessments for the general public, as discussed in Section 3.2.3, the upper limits for hazard quotients associated with the longer-term exposures are sufficiently below a level of concern that the risk characterization is relatively unambiguous: based on the available information and under the foreseeable conditions of application, there is no route of exposure or scenario suggesting that the general public will be at any substantial risk from longer-term exposure to picloram even if the level of concern is set to 0.35 - i.e., that associated with the maximum application rate that will be used in Forest Service programs.

For the acute/accidental scenarios, none of the central estimates representing typical exposure conditions exceed the RfD. The estimate of the upper range of exposure resulting from the consumption by a child of contaminated water from a small pond immediately after an accidental

spill (Section 3.2.3.4.1) does exceed the level of concern by a factor of 3. It does not seem likely that this exposure would result in any frank adverse effects. As noted in Section 3.3.2, the chronic NOAEL of 20 mg/kg/day used to derive the chronic RfD is a factor of about 20 below reported NOAELs from shorter term exposures.

Each of the hazard quotients summarized in Worksheet E04 involves a single exposure scenario. In some cases, individuals could be exposed by more than one route and in such cases risk can be quantitatively characterized by simply adding the hazard quotients for each exposure scenario. For picloram, considerations of multiple exposure scenarios has little impact on the risk assessment. For example, typical levels of exposure for a woman being directly sprayed on the lower legs, staying in contact with contaminated vegetation, eating contaminated fruit, drinking contaminated water from a stream, and consuming contaminated fish at rates characteristic of subsistence populations leads to a combined hazard quotient of 0.088 (0.0004 + 0.0006 + 0.02 + 0.007 + 0.06). Similarly, for all of the chronic exposure scenarios, the addition of all possible pathways lead to hazard quotients that are substantially less than unity. Some acute exposure scenarios could be constructed at the upper limits of exposure that could lead to modest excursions about the RfD but compounding several different upper range assessments would constitute an implausible series of events.

3.4.4. Sensitive Subgroups. There is no information to suggest that specific groups or individuals may be especially sensitive to the systemic effects of picloram. As discussed in Sections 3.1 and 3.3.2, the likely critical effect of picloram in humans cannot be identified clearly. In animals, the most sensitive effect of picloram involves changes in the staining characteristics of liver cells. These effects, however, were only noted in one study and are not consistent among species or even between different studies in the same species. Thus, it is unclear if individuals with pre-existing diseases of the liver would be particularly sensitive to picloram exposures, although individuals with any severe disease condition could be considered more sensitive to many toxic agents.

3.4.5. Connected Actions. As noted in Section 3.1.3, a commercial formulation of picloram and 2,4-D, Tordon 202C, has been shown to inhibit immune response in mice (Blakley 1997). The effects of mixtures of picloram, alachlor, and atrazine compared to the activity of each chemical given alone have been characterized in both 90-day drinking water studies (10 ppm in water) and 90-day gavage studies (100 mg/kg in corn oil) using mice (Chaturvedi 1993). Again, the design of this study does not permit a quantitative characterization of interactions of these three pesticides. Nonetheless, exposures to picloram were associated with increased weights of the spleen, kidney, and liver.

3.4.6. Cumulative Effects. As noted above, this risk assessment specifically considers the effect of repeated exposure in that the chronic RfD is used as an index of acceptable exposure. Consequently, repeated exposure to levels below the toxic threshold should not be associated with cumulative toxic effects.

3.4.7. Hexachlorobenzene.

3.4.7.1. Workers – Summaries of the exposure assessments and risk characterization for workers are given in the hexachlorobenzene worksheets that accompany this risk assessment (Supplement 2). Worksheet E01 summarizes the exposure assessment for workers and is analogous to the corresponding worksheet for picloram. Worksheet E02 summarizes the risk characterization for workers.

Unlike picloram, the toxicity data on hexachlorobenzene allows for separate dose-response assessments for acute and chronic exposures. For acute exposures, the hazard quotients are based on ATSDR's short-term MRL of 0.008 mg/kg/day (ATSDR 2002). For chronic exposures, the hazard quotients are based on the chronic RfD from U.S. EPA of 0.0008 mg/kg/day.

For general worker exposures, the hazard quotients associated with hexachlorobenzene (Worksheet E02) are approximately four orders of magnitude below the corresponding hazard quotients for picloram. Similarly, hazard quotients associated with accidental scenarios are consistently lower for hexachlorobenzene than the corresponding scenarios for picloram. Thus, for the reasonably diverse exposure scenarios covered in this risk assessment, the amount of hexachlorobenzene in technical grade picloram is not toxicologically significant.

The cancer risks presented in Worksheet E02 are presented as the estimated exposure divided by the lifetime dose associated with a cancer risk of 1 in one million. Thus, the interpretation of these quotients is identical to that of hazard quotients for toxicity -i.e., if the hazard quotient is below unity, the cancer risk is below 1 in one million. As indicated in Worksheet E02, none of the cancer risks in workers exceed 1 in one million.

As indicated in Section 3, all of these risk characterizations are based on the typical or average 8 ppm concentration of hexachlorobenzene in technical grade picloram. Particularly for chronic exposures in which workers would be exposed over several years to hexachlorobenzene in many different batches of picloram, the use of the average concentration appears to be a much more reasonable approach than the use of the upper limit.

While there are substantial uncertainties involved in any cancer risk assessment, the verbal interpretation of the numeric risk characterization derived in this risk assessment is relatively simple. Using the assumptions and methods typically applied in Forest Service risk assessments, there is no plausible basis for asserting that the contamination of picloram with hexachlorobenzene will result in any substantial risk of cancer in workers applying picloram under normal circumstances.

While the chronic cancer potency could be scaled linearly and the cancer risk associated with short term exposures could be calculated, this sort of extrapolation is highly uncertain and, more importantly, ignores the normal background exposures to hexachlorobenzene from other sources. For example, background levels of exposure to hexachlorobenzene are in the range of 0.000001 mg/kg/day or 1×10^{-6} mg/kg/day (Section 3.2.4.3). As summarized in Worksheet E01, even the

upper range worker exposure values are below this background dose – i.e., in the range of 2 to 4×10^{-7} mg/kg/day. As discussed in the next section, the upper range of the longer term exposure scenarios for the general public are substantially below the background dose – i.e., about 1×10^{-10} to 6×10^{-8} . Thus, there is no basis for asserting that the presence of hexachlorobenzene in picloram will impact substantially cancer risk under conditions characteristic of applications made in Forest Service programs.

3.4.7.2. General Public – Summaries of the acute exposure assessments and risk characterization for the general public are given in the hexachlorobenzene worksheets that accompany this risk assessment and parallel those for the risk characterization for workers discussed in the previous section: Worksheet E03 summarizes the exposure assessments and Worksheets E04a, E04b, and E04c summarize the risk characterizations.

Worksheet E04a presents the hazard quotients for the general public associated with the acute exposure scenarios. As with the corresponding worksheet for workers, the hazard quotients for acute exposure are based on the short-term MRL of 0.008 mg/kg/day and the hazard quotients for chronic exposures are based on the U.S. EPA RfD of 0.0008 mg/kg/day.

All exposure scenarios result in hazard quotients that are below unity (i.e., the level of exposure is below the RfD for chronic exposures and below the MRL for acute exposures). In addition, all of the acute exposure scenarios result in hazard quotients that are substantially below the corresponding hazard quotient for picloram. The highest acute hazard quotient for hexachlorobenzene is 0.1, the upper range of the hazard quotient associated with the consumption of contaminated fish by subsistence populations.

As with worker exposures, none of the hazard quotients for cancer risk levels of 1 in 1-million exceed unity. As noted in Section 3.2.4.3, the typical background exposure to hexachlorobenzene is about 0.000001 or 1×10^{-6} mg/kg/day (ATSDR 2002). As indicated in hexachlorobenzene Worksheet E03, the highest longer-term exposure rate associated with Forest Service programs is 6.48×10^{-8} mg/kg/day – i.e., the upper range of exposure for the consumption of contaminated fish by subsistence populations. This is below the typical background exposure by a factor of about 15.

The simple verbal interpretation of this risk characterization is that, in general, the contamination of picloram with hexachlorobenzene does not appear to pose a risk to the general public. This is consistent with the conclusions reached by the U.S. EPA (1995b).

4. ECOLOGICAL RISK ASSESSMENT

4.1. HAZARD IDENTIFICATION

4.1.1. Overview. The toxicity of picloram is relatively well characterized in experimental mammals but few wildlife species have been assayed relative to the large number of nontarget species that might be potentially affected by the use of picloram. Within this admittedly substantial reservation, picloram appears to be relatively non-toxic to terrestrial animals but is moderately toxic to aquatic animals, particularly some species of fish.

The assessment of the toxicity of picloram to nontarget terrestrial animals is based almost exclusively on toxicity studies using experimental mammals (i.e., the same studies used in the human health risk assessment). Acute oral LD_{50} values for picloram are in the range of 3000 to 5000 mg/kg body weight and NOAEL from chronic studies range from 7 mg/kg/day to 20 mg/kg/day. Some additional studies are available on birds, bees, and snails that generally support the characterization of picloram as relatively non-toxic to terrestrial animals. This assessment is supported by field studies that reported no detectable effects on mammalian or avian diversity after the application of picloram.

Picloram is a pyridine herbicide that acts as a plant growth regulator. This is to say that picloram mimics naturally occurring plant auxins or hormones in a manner that leads to uncontrolled and abnormal growth. These effects can in turn lead to gross signs of toxicity or death. The toxicity of picloram to terrestrial plants has been assayed in relatively standardized studies of seed emergence, seed germination, and post-emergence applications that have been submitted to the U.S. EPA to support the registration of picloram. Picloram is more toxic to broadleaf plants than grains or grasses. The lowest reported adverse effect (the EC_{25} for the inhibition of seed emergence in soybeans) for the potassium salt of picloram is 0.000014 kg or about 0.000012 lb a.e./acre. The highest reported NOEC in any of the terrestrial plant bioassays is about 0.4 lb a.e./acre for seedling emergence in corn.

The acute and chronic toxicity of picloram to aquatic animals has been assayed in various species of fish and invertebrates. Acute (96-hour) LC_{50} value for trout range from 0.8 mg/L to 19.3 mg/L. Bluegill sunfish and fathead minnows, common test species used in aquatic toxicity studies, appear to be less sensitive to picloram, with LC_{50} values ranging from about 15 mg/L to 55 mg/L. Two sets of longer-term (egg and fry) studies are available in trout and the results of these studies are not consistent. Studies accepted by the U.S. EPA indicate a NOEC of 0.55 mg/L in rainbow trout. An earlier series of studies of using lake trout indicate a NOEC of <0.035 mg/L. Limitations in the use of this earlier series of studies include the failure to use an acetone control and the failure to measure concentrations in the test solutions. An early life-stage study in the fathead minnow, yielded a NOEC of 0.71 mg a.e./L, very similar to the NOEC reported in trout. This study, however, was conducted with the triisopropanolamine salt of picloram rather than the potassium salt.

In *Daphnia*, an aquatic invertebrate commonly used in aquatic toxicity studies, the reported acute (48-hours) LC_{50} value is 68.3 (63–75) mg/L. Chronic studies in the same species using reproductive or developmental parameters identified a no-effect level at 11.8 mg/L and a lowest effect level at 18.1 mg/L. Thus, it appears that trout are more sensitive than daphnids to both the acute and chronic effects of picloram.

As with aquatic animals, the toxicity of picloram to aquatic plants varies substantially among different species. Based on the available toxicity bioassays, the most sensitive species is *Navicula pelliculosa*, a freshwater diatom, with an EC₅₀ for growth of 0.94 mg a.e./L and a NOEC of 0.23 mg a.e./L. The least sensitive aquatic plants appear to be from the genus *Chlorella* (another group of freshwater algae), with EC₅₀ values greater than 160 mg a.e./L. Macrophytes appear to have a sensitivity that is in the upper range of that seen in algae, with a reported EC₅₀ of 164 mg a.e./L in duckweed.

4.1.2. Toxicity to Terrestrial Organisms.

4.1.2.1. *Mammals* – As detailed in Section 3.1 (Hazard Identification for the human health risk assessment), the toxicity of picloram to experimental mammals is well characterized and these data are relevant to the assessment of effects in wildlife species. Two issues, however, are unique to the ecological risk assessment for mammalian species: the NOAEL for acute exposures and differences in chronic NOAELs for different mammalian species.

Picloram has a low acute oral toxicity. A common measure of acute oral toxicity is the LD_{50} , the estimate of the dose that may be lethal to 50% of the exposed animals. As summarized in Section 3.1.4, reported acute oral LD_{50} values in rats range from 690 mg/kg (Hayes et al. 1986) to >5000 mg/kg in male rats (Jeffrey 1987a). The lower end of this ranged involved exposures to unneutralized potassium picloram solution (pH>11) and the atypically high toxicity relative to other studies is probably due to the extreme alkalinity of the test solution (U.S. EPA 1992b). In terms of practical significance, these acute LD_{50} values are several orders of magnitude higher than any plausible exposures and have no practical impact on the risk assessment.

As noted in the human health risk assessment (Section 3.3), the U.S. EPA (1999) has not derived an acute exposure limit for picloram because:

No toxicological effect that could be attributable to a single oral exposure was identified, and therefore picloram is not expected to present an acute dietary hazard (p. 421, Section C.1.i.).

For the current risk assessment, however, the failure to define an acute NOAEL limits the interpretation of potential risks associated with many of the acute exposure scenarios developed for mammals (Section 4.2). As discussed further in the dose-response assessment, the reproduction studies in experimental mammals (Section 3.1.9) can be used to estimate acute exposures that are not likely to result in adverse effects in mammals.

In the human health risk assessment, a NOAEL of 20 mg/kg/day is used as the basis for the RfD (Section 3.3.2). This NOAEL is from a two-year feeding study in male and female Fischer rats (50 rats/sex/dose) in which picloram (acid) was administered at dietary concentrations that resulted in daily doses of 20, 60, and 200 mg/kg/day (Landry et al. 1986). The dose of 60 mg/kg/day was considered an LOAEL but the severity of the effect – i.e., changes in liver weight and appearance – were considered marginal by the U.S. EPA (U.S. EPA 1994). Typically, the NOAEL used for the RfD is applied directly to the ecological risk assessment for mammalian species. However, as detailed in Section 3.1.5, a LOAEL of 35 mg/kg/day with a corresponding NOAEL of 7 mg/kg/day has been noted in dogs over the course of a 6-month feeding study (Barna-Lloyd et al. 1982). The U.S. EPA had used this study in an earlier RfD. Dogs, however, are generally considered a poor animal model for toxicity to the kidney. Thus, this study is not used in the human health risk assessment. Nonetheless, for the ecological risk assessment, 7 mg/kg/day is the most sensitive NOAEL and the use of the 7 mg/kg/day NOAEL from the study in dogs is considered further in the dose-response assessment (Section 4.3).

In addition to these laboratory bioassays, field studies are available in which the impact of picloram applications were assessed on mammalian wildlife communities. Brooks et al. (1995) examined the impact of a mixture of picloram and triclopyr as well as imazapyr and hexazinone, all used in site preparation, on small mammal and avian communities. The study area was located in Georgia and consisted of a 157-ha tract of residual hardwoods. Picloram was applied in combination with 2,4-D (Tordon 101) at a rate of 2.7 kg a.e./ha (approximately 2.4 lbs a.e./acre). After herbicide treatment and a prescribed burn, loblolly pine were planted. Data on small mammals was collected by trapping and data on birds involved visual surveys. Observations were made at pre-treatment and three-times per year at 1, 2, and 3 years after treatment. No substantial differences were noted among the different herbicides. With all herbicides, the number of small animals trapped after treatment was diminished compared to pre-treatment levels. Because no non-herbicide treated sites (i.e., control sites) were used in this study, observed changes in populations of small mammals or birds cannot be clearly associated with herbicide treatment.

Nolte and Fulbright (1997) have conducted a field study assessing the effects of a combination of picloram and triclopyr on community structure. Each herbicide was applied by helicopter at a rate of 1.9 liters/ha, but neither the rate in units of lb/acre nor the specifics of the commercial formulation are given in the publication. This paper does indicate that the application is that "commonly used on mesquite in southern Texas". No effects were seen on mammalian or avian diversity. In addition, no statistically significant effects were noted on vegetation species richness and evenness or the number of rare plant species.

4.1.2.2. Birds – As summarized by U.S. EPA (1995b), the acute toxicity of the potassium salt of picloram to birds by gavage administration appears to be similar to that in mammals with acute oral LD_{50} values of >2000 mg/kg in quail and mallard ducks. In mallard ducks, Beavers (1983)

reported an oral LD_{50} of more than 2510 mg/kg for technical grade picloram with no deaths at the highest dose tested.

In addition, 14-day dietary LC_{50} values in these species are >10,000 ppm and supplemental reproductive studies on chickens and pheasants indicated NOAELs, expressed as application rates, of 2.8 kg/ha and 11.2 kg/ha, respectively (U.S. EPA 1995b). Assuming that birds consume about 15% of their body weight during a standard toxicity study (similar to the values used in other risk assessments for which food consumption data are reported), the dietary concentration of 10,000 ppm corresponds to a 14-day NOAEL of about 1500 mg/kg body weight.

An 8-day dietary LC_{50} value of 385,260 ppm (about 58,000 mg/kg body weight) in mallard ducklings has been reported by Stevenson (1965) for Tordon but the specific formulation and % a.e. in the formulation is not specified in the study.

4.1.2.3. *Terrestrial Invertebrates* – Acute toxicity studies using bees are required by the U.S. EPA in the registration of pesticides—using both dietary and direct contact exposures. For picloram (K salt), the acute contact LD_{50} is >0.1 mg/bee and the dietary LC_{50} is >1000 ppm. Taking an estimated body weight for the honey bee of 0.093 g (USDA/APHIS 1993), the contact toxicity LD_{50} value of >0.1 mg/bee corresponds to a dose of >1 mg/g or 1,000 mg/kg. The RED for picloram (U.S. EPA 1995b) reported that the LD_{50} study found no significant mortality at the highest exposure level tested. There is relatively little additional data on terrestrial invertebrates. At dietary concentrations of about 5000 mg/kg over a 14-day period, picloram (acid) did not increase mortality in the brown garden snail, *Helix aspersa* (Schuytema et al. 1994). Based on these albeit limited data, there is no basis for asserting that picloram is likely to be more toxic to terrestrial invertebrates than it is to terrestrial mammals or birds.

4.1.2.4. *Terrestrial Plants (Macrophytes)* – Picloram is a pyridine herbicide that acts as a plant growth regulator. This is to say that picloram mimics naturally occurring plant auxins or hormones in a manner that leads to uncontrolled and abnormal growth that can lead to gross signs of toxicity or death (Grossmann et al. 2001; Hansen and Grossmann 2000; Webb and Hall 1995). In general, picloram is more toxic to broad leaf plants than to grasses or grains (Cox 1998; ExToxNet 1996b; USDA 1995; U.S. EPA 1995a,b). The yellow starthistle, *Centaurea solstitialis*, has developed resistance to picloram with resistant plants being more tolerant by factors ranging from 3 to 35 fold compared to non-resistant plants (Fuerst et al. 1996). Based on growth inhibition in sunflower seedlings, picloram was more toxic than metabolites of picloram by factors of about 300 to 3000 (Grover et al. 1975).

In assessing the potential effects of herbicides on nontarget plant species, the U.S. EPA has developed a standardized set of plant bioassays for seed germination, seed emergence, and post-emergence applications.

As summarized in the RED for picloram, the lowest reported adverse effect for the potassium salt of picloram is 0.014 g a.e./ha or 0.000014 kg a.e./ha, the EC_{25} for seed emergence in soybeans

(U.S. EPA 1995b, p. 55). This corresponds to an application rate of about 0.000012 lb a.e./acre. A more recent Tier II seed emergence study using Tordon K (Schwab 1995) has also been submitted to U.S. EPA. In this study, the most sensitive species was the pinto bean, with a NOEC of 0.27 g a.i./hectare and an EC_{25} of 7.4 g a.e./hectare. These values correspond to about 0.1 g a.i./acre and 3 g/acre [2.471 acres/ha] or 0.086 g a.e./acre and 2.6 g a.e./acre [conversion factor of 0.8606, Table 2-1]. The NOEC of 0.086 g a.e./acre, in turn, corresponds to an application rate of about 0.00019 lb a.e./acre [1 pound = 453.6 g]. Soybeans were much less sensitive than the pinto bean, with a NOEC of 8.75 g/Ha, corresponding to about 0.006 lb a.e./acre. In the seedling emergence phytotoxicity test by Schwab (1995), corn was found to be the most resistant of 10 plant species (barley, corn, onion, pinto bean, radish, rape, soybean, sunflower, tomato, and wheat) to the toxic effects of picloram (K salt) with a NOEC of 560 g a.i./ha or about 0.4 lb a.e./acre (Schwab 1995).

Studies on vegetative vigor indicate that the sunflower is the most sensitive of 12 plant species (alfalfa, barley, corn, cucumber, onion, pinto bean, radish, rape, soybean, sunflower, tomato, and wheat) (Schwab 1996). These Tier 2 studies identified effect levels for picloram in sunflower of 0.67 and 3.19 g a.i./ha (EC_{25} and EC_{50} , respectively) with a NOEC of 0.27 g a.i./ha (Schwab 1996). The most tolerant plant species in an assay for vegetative vigor is wheat, with a NOEC of 70 g a.e./ha (U.S. EPA 1995b, p. 55).

Several studies have been published in the open literature concerning the toxicity of picloram to non-target plant species. Harrington et al. (1998) followed the two-year development of southern pine seedlings and associated vegetation following spray-and-burn site preparation with various herbicides, including picloram. Miller et al. (1999) measured floristic diversity, stand structure, and composition 11 years after herbicide site preparation. Sparkes et al. (2002) examined the effects of various herbicides, including picloram, on *Bryophyllum pinnatum (Lam.)*. Numerous studies have been published on efficacy and effects of picloram applications on various crop species. Rates as low as 25 g a.i./ha of picloram reduced tobacco yields as long as four years after application (Sheets and Harrell 1986). Another study found that application of picloram (K salt) at 0.05 pounds per acre resulted in a statistically significant reduction in cotton growth and yield when applied directly (Jacoby et al. 1990). Thus, picloram can be considered highly selective to broad leaf plants such as soybean but may be toxic to many different plant species at the typical application rate of 0.5 lb a.e./acre.

4.1.2.5. *Terrestrial Microorganisms* – The persistence of picloram in soil increases with increasing application rates or soil concentrations and this suggests that picloram is toxic to soil microorganisms. In soil column studies conducted over a 30 day period, Krzyszowska et al. (1994) noted that the soil halftimes of picloram is directly related to application rates. Application rates of 0.47, 0.97, and 1.85 kg/ha (about 0.4, 0.86, and 1.6 lb/acre) were associated with halftimes in soil of 13, 19, and 23 days, respectively.

Most of the data on the effects of picloram on soil microorganisms involve assays of microbial activity in soils with defined concentrations of picloram. Consistent with the study by

Krzyszowska et al. (1994), USDA/ARS (1995) notes a direct relationship between aerobic soil halftimes and concentrations of picloram in soil: 18 days at a concentration of 0.0025 ppm, 29 days at a concentration of 0.025 ppm, 150 days at a concentration of 0.25 ppm, and 300 days at a concentration of 2.5 ppm. At a level of 10 ppm in sandy loam soil, picloram — and several other herbicides—caused a transient decrease in nitrification after 2 but not 3 weeks of incubation (Tu 1994). As discussed by this investigator, the decrease in nitrification is relatively mild and does not suggest the potential for a substantial or prolonged impact on microbial activity. In the same study, picloram had no effect on ammonia formation or sulfur oxidation. Prado and Airoldi (2001) assayed the effect of picloram on mixed microbial activity using microcalorimetry – assays microbial activity by measuring changes in heat production from soil treated with glucose (microbial food source) and various concentrations of picloram. Time to peak heat production was attenuated and the magnitude of peak heat production was reduced by picloram concentrations as low as 1 ppm.

Welp and Bruemmer (1999) described the pH dependence of toxicity measurements of picloram (acid) in soil as determined by Fe(III) reduction test. The results showed that EC_{50} ranged from 1.93 mmol/kg [about 465 ppm] soil to more than 16.6 mmol/kg [about 4000 ppm] soil over a pH range of 3.5–7.8 (Welp and Bruemmer 1999).

Unlike the case in macrophytes, the metabolism of picloram may result in increased rather than decreased toxicity in some microorganisms. In three species of fungi, EC_{50} values for growth inhibition by picloram acid were >1600 ppm (the highest concentration tested). Corresponding values for the decarboxylated metabolite, 4A-TCP, were 50 to 80 ppm. In two species of bacteria, *Arthrobacter globiformis* and *Pseudomonas pictorum*, differences in toxicity were not substantial and ranged between 60 and 380 ppm for picloram acid and 4A-TCP (Baarschers et al. 1988). Using an assay based on the reduction of Fe(III) in soil, Welp and Bruemmer (1999) reported EC_{50} values of 1.93 to >16.6 mMoles/kg soil, corresponding to about 466 mg/kg soil to somewhat of over 4,000 mg/kg soil [MW=241.48].

4.1.3. Aquatic Organisms.

4.1.3.1. Fish

4.1.3.1.1. Acute Toxicity – As with terrestrial species, the acute lethal potency of picloram and picloram formulations has been relatively well-defined. These values are typically expressed as time-specific LC_x values where x is the estimate of the proportion of fish that die – e.g., 96 hour LC_{50} . A large number of acute LC_{50} values have been determined in various species of fish. Studies submitted to U.S. EPA for the registration and reregistration of picloram are summarized in U.S. EPA (1995b) and a large number of additional studies have been reviewed by Mayes and Oliver (1985). Additional studies in trout have been published by Woodward (1976, 1979, 1982).

Based on studies considered by U.S. EPA (1995b), salmonids appear to be marginally more sensitive to picloram (acid) than other fish species with a reported LC_{50} value of 5.5 mg/L in rainbow trout (confidence limits not reported) compared to LC_{50} values of 14.5 to 19.4 mg/L for

bluegill sunfish (a species of warm water fish commonly used in toxicity bioassays). These species differences are not apparent in the toxicity data summarized by U.S. EPA (1995b) on the potassium salt of picloram used in the formulations covered by this risk assessment. For the potassium salt, LC_{50} values of 13 mg/L and 26 mg/L are reported for rainbow trout and a corresponding value of 24 mg/L is reported for bluegills (U.S. EPA 1995b). Based on these data, the U.S. EPA (1995b) classified picloram acid as moderately toxic to freshwater fish based on the LC_{50} of 5.5 mg/l in trout and also classified the potassium salt of picloram as moderately toxic to freshwater fish based on the LC_{50} of 13 mg/L in trout.

The review of additional studies by Mayes and Oliver (1985) suggests differences in sensitivity between salmonids and other species of fish are more substantial (Table 4-1). For technical grade picloram (acid), the LC₅₀ values for salmonids are reported as 4.8 (3.8-6.2) mg/L in cutthroat trout to LC₅₀ values ranging from 5.5 to 19.3 mg/L in rainbow trout. For bluegill sunfish, the reported LC₅₀ values range from 14.5 to 44.5 mg/L and an LC₅₀ value of 55.3 (47.2 to 69.6) mg/L is reported for fathead minnow. As summarized in Table 4-1, the potassium salt of picloram appears to have similar toxicity to the acid form in cutthroat trout, rainbow trout, bluegill sunfish, and fathead minnow. Thus, while there is variability among multiple bioassays in the same species, the general tendency is for the LC_{50} values to be lower in salmonids than in other fish species. The quantitative consideration of higher sensitivity of salmonids relative to other fish species is discussed further in the dose-response assessment (Section 4.3.3.1). In their review, Mayes and Oliver (1985) do not clearly specify whether or not solvents were used in the various bioassays. In some cases, solvents do appear to have been used. For example, Mayes and Oliver (1985) cite a 96-hour LC₅₀ value of 13 mg/L for the potassium salt of picloram. This appears to have been taken from the unpublished study by Alexander and Batchelder (1965), which was submitted to the U.S. EPA in the support of the registration of picloram.

Woodward (1976) also conducted static acute toxicity tests of technical grade picloram (acid form) with cutthroat trout and lake trout. Woodward (1976) reports LC_{50} values for cutthroat trout of 3.5 (3.4-4.0 mg/L) mg/L and for lake trout of 1.6 (1.2 - 2.0) mg/L. A later publication by Woodward (1982) reported a 96-hour LC_{50} of 3.9 (3.2 - 4.8) mg/L for fingerling cutthroat trout. As discussed in the following subsection on chronic toxicity in fish, Woodward (1976) used acetone as a vehicle whereas vehicles were not used in the studies summarized by Mayes and Oliver (1986). Unlike differences observed in the longer-term studies (Section 4.1.3.1.2), the acute toxicity data reported by Woodward (1976, 1982) is reasonably consistent with the data summarized by Mayes and Oliver (1985).

A toxicity test in a saltwater species, the sheepshead minnow, has been submitted to U.S. EPA since the publication of the RED (Boeri et al. 1995b). No mortality was observed in this species at the highest concentration tested: 131 mg/L expressed as Tordon 22 K or 27.2 mg a.e./L (Boeri et al. 1995b).

4.1.3.1.2. Subchronic Effects – For assessing the longer-term effects of picloram on fish, the U.S. EPA (1995b) used the egg and fry study in rainbow trout (*Salmo gairdneri*) by Mayes et al.

(1984), which was subsequently published in the open literature (Mayes et al. 1987). In this study, rainbow trout embryos (10 days pre-hatch) were exposed to technical grade picloram at average measured concentrations of 0 (control), 0.23, 0.38, 0.55, 0.88, 1.34, and 2.02 mg/L for 60-days post-hatch. Stock solutions were prepared by mixing technical grade picloram (acid) in deionized water and adjusting the pH to 8 with potassium hydroxide. Thus, the U.S. EPA (1995b) appropriately considered this assay as directly applicable to the potassium salt of picloram. Concentrations were maintained during the bioassay using a standard proportional diluter. No statistically significant or dose-related effects were noted on embryo hatching. Larval survival, however, was significantly reduced at (72.5% of controls) at 2.02 mg/L. In addition, concentration-related reductions were noted in both body weight and body length of fry at concentrations of 0.88 mg/L with an apparent NOEC of 0.55 mg/L.

Woodward (1976) conducted a similar bioassay using lake trout (*Salvelinus namaycush*) in which embryos were exposed to nominal concentrations 0, 0.035, 0.075, 0.240, 0.5, or 1 mg/L over 60 days. As in the study by Mayes et al. (1984), Woodward (1976) used a standard proportional diluter to adjust the concentrations among the different exposure groups. Unlike the study by Mayes et al. (1984), Woodward (1976) prepared stock solutions in acetone rather than water and concentration of picloram in the test solution was measured only in the highest test concentration. Woodward (1976) reports only a single control group and it is unclear if this represents an untreated or acetone control. Statistically significant reductions in lake trout fry survival and growth (yolk sac absorption rate, length, and weight) were noted at all concentrations and the effects are clearly concentration related for survival in fry prior to yolk sac absorption, as well as 60-day post-hatching survival, body weight, and length.

In another study of cutthroat trout (*Salmo clarki*) in which runoff exposure to picloram (acid) was simulated, Woodward (1979) reported that fluctuating concentrations of picloram (with 48 hour pulses of picloram acid on days 1, 8, 15, and 22 over total time of 24 days) resulted in a statistically significant reduction in cutthroat trout fry survival (LOEC = 1.3 mg/L) and growth (LOEC = 0.61 mg/L). The maximum concentration of picloram that did not affect survival and growth of cutthroat trout was 0.29 mg/L (NOEC) (Woodward 1979).

The discrepancies between the results of Mayes et al. (1984) and those of Woodward (1976, 1979) cannot be clearly resolved. The 1995 RED for picloram did not cite Woodward (1976). The 1995 RED for picloram considered the chronic toxicity data from field runoff studies by Woodward (1979) supplemental to the risk assessment because the studies were not required for registration but does not explicitly discuss the intermittent LOEC of 0.61 mg/L which is below the NOEC of 0.55 mg/L from the study by Mayes et al. (1984). Mayes and Oliver (1985) cite the study by Woodward (1979) without comment on the LOEC and do not cite the study by Woodward (1976), which is very similar in design to the study by Mayes et al. (1984).

The most obvious difference between the studies by Mayes et al. (1984) and those of Woodward (1976) involves the species tested, rainbow trout in the former and lake trout in the latter. Based on the acute toxicity data, lake trout have an acute LC_{50} of 1.6 (1.2 - 2.0) mg/L (Woodward
1976). Rainbow trout, on the other hand, have reported LC_{50} values in the range of 5.5 to 19.3 mg/L (Mayes and Oliver 1986). Thus, there is some basis for asserting that lake trout may be more sensitive to picloram than rainbow trout.

Other differences, however, involve the nature of the form of picloram (acid vs potassium salt), the failure of Woodward (1976) to monitor concentrations in all test solutions, and the use of acetone as a vehicle by Woodward (1976). The first two factors are not likely to be important. As reviewed by Mayes and Oliver (1985), differences in the toxicity of the various forms of picloram are not substantial. The failure of Woodward (1976) to monitor concentrations in the test solutions does increase uncertainty in the interpretation of the study but, assuming that the proportional diluters were properly calibrated, the failure to monitor concentrations would not in itself reduce substantially the credibility of the study.

The failure of Woodward (1976) to specify or possibly use an acetone control, however, is potentially serious. The current U.S. EPA/OPPTS. (2003) test guidelines for fish-early life stage toxicity studies require the use of a solvent-only control. This requirement is reasonable because solvent can impact the test organism directly and/or can impact the uptake of the test compound by the organism. Because the concentration of acetone in the test water is not reported by Woodward (1976), the potential significance of the use of acetone cannot be assessed directly. Under current standards, the study by Woodward (1976) would not be classified as acceptable using the U.S. EPA/OPPTS (2003) test criteria. While no studies have been encountered on the impact of acetone on the toxicity of picloram or other herbicides in longer-term (egg and fry) studies on trout, Mac and Steelye (1981) did note that low concentrations of acetone (10 μ L/L) did increase the uptake of PCBs by lake trout over the course of a 52-day egg and fry study. The differences, however, were only statistically significant at Day 52 and no differences were noted in morality between the various experimental groups: untreated controls, trout exposed to PCBs alone, acetone alone, or PCBs with acetone.

No early life-stage or lifetime studies have been conducted with the Tordon formulations or the potassium salt of picloram in other more tolerant species of fish. Weinberg et al. (1996) have conducted an early life-stage study on fathead minnows using the triisopropanolamine salt of picloram. While the study is reported in detail, the precise method for preparing the high concentration solution is not specified. Methanol was used as a vehicle in the preparation of analytical standards. The study protocol (p. 11 of 39) does specifically state that the delivery system (a standard proportional diluter) would include a solvent control "if necessary". No results for a solvent control are reported. Thus, methanol does not appear to have been used as a vehicle in preparing the test solutions to which the fish were exposed. Statistically significant effects on growth and survival were seen at 20.1 mg/L but not at any lower concentrations (1.6 to 12 mg/L). These results appear to be expressed as the gross weight of the test material. In this study, the test material is reported to contain 10.6% a.i. and 5.9% a.e. Thus, the NOEC of 12 mg/L would correspond to about 0.71 mg a.e./L and the LOEC of 20.1 mg/L would correspond to a concentration of about 1.2 mg a.e./L. This NOEC is only modestly higher than the NOEC of 0.55 mg a.e./L reported in trout by Mayes et al. (1987).

The discrepancy between the LOEC of 0.035 mg/L in lake trout reported by Woodward (1976) and the NOEC values of 0.55 mg/L in rainbow trout (Mayes et al. 1984) and 0.71 mg/L in fathead minnow (Weinberg et al. 1996) has a major impact on this risk assessment and the use of these data in the risk assessment is detailed further in Section 4.3.3.1.

4.1.3.1.3. *Field Observations* – Keys (1992) reports a fish-kill incident that occurred in Montana after the application of picloram. In this incident, Tordon 22K was applied on July 12 and July 20, 1989 at a distance of 0.25 miles upstream from a fish hatchery. The amount applied and the distance of the application from the stream are not specified. The day following the second treatment, rain occurred (amount not specified) and a number of trout (characterized only as 8,880 pounds) died. Given the lack of detail in this report, it is difficult to assess whether or not the fish-kill incident was related to the applications of picloram. The plausibility of this type of event occurring after the application of picloram in Forest Service programs is discussed further in Section 4.4 (Risk Characterization).

4.1.3.2. *Amphibians* – Neither the published literature nor the RED for picloram (U.S. EPA 1995b) includes data regarding the toxicity of picloram to amphibian species.

4.1.3.3. Aquatic Invertebrates – In Daphnia magna, an acute (48-hour) LC_{50} value of 68.3 (63–75) mg/L and a chronic NOAEL of 11.8 mg/L of picloram (technical picloram)—based on mean total young per adult, total number of broods per adult, and mean brood size per adult—has been reported in the open literature (Gersich et al. 1985). At concentrations of 18.1 mg/L and higher, all of the indices for reproductive performance were decreased (Gersich et al. 1985). This information is identical to that summarized in the RED for picloram (U.S. EPA 1995b).

Information about the toxicity of picloram in mollusks is limited. No studies are available on the toxicity of picloram acid. The 1995 RED for picloram (U.S. EPA 1995b) cites the study Heitmuller (1975) for a 48-hour LC₅₀ value of 18–32 mg/L for a 24.9% formulation of the potassium salt of picloram in an embryo/larvae assay in oysters. This corresponds to an LC₅₀ value of about 4.5-8.0 mg a.i./L or 3.8-6.9 mg a.e./L. It is worth noting that this study did involve the testing of Tordon 22K, one of the formulations used by the Forest Service. In addition, the study by Heitmuller (1975) also involved bioassays of pink shrimp (*Penaeus duorarum*) and fiddler crabs (*Uca pubilator*), in addition to the eastern oyster (*Crassostrea virginica*). The 96 hour LC₅₀ value in shrimp was 125 (114-138) mg/L expressed as the formulation, corresponding to about 31 mg a.i./L and 26.8 mg a.e./L. Fiddler crabs were not affected at concentrations of up to 1,000 mg/L of the formulation (249 mg a.i./L or 214 mg a.e./L).

No field studies are available on the toxicity of picloram to aquatic invertebrates. An unusually high number of gonadal neoplasms was identified in softshell clams from three Maine estuaries contaminated with herbicides, including picloram, 2,4-D, and 2,4,5-T (Gardner et al. 1991; Van Beneden 1993). However, the latter reports do not implicate picloram (or any specific herbicide directly) and the cause of tumors is unknown.

4.1.3.4. Aquatic Plants – The toxicity of picloram has been examined in both algae and aquatic macrophytes . The RED for picloram reports a NOEC of 13.1 mg/L and an EC₂₅ for growth inhibition of 52.6 mg/L for a formulation of the potassium salt of picloram (35.2% a.i. or 30.2% a.e.) in *Selenastrum capricornutum* (U.S. EPA 1995b) for a formulation . This LC₅₀ value for the formulation corresponds to an LC₅₀ of 15.9 mg a.e./L. This is the only information summarized in U.S. EPA (1995b) on toxicity to aquatic plants.

Since the publication of the RED for picloram (U.S. EPA 1995b) additional studies have been submitted to the U.S. EPA on the toxicity of picloram to algae. In a Tier 2 acute toxicity test of Tordon K (24.1% picloram a.e.) using *Anabaena flos-aquae*, a freshwater blue-green alga, Boeri et al. (1994b) reports a 120-hour EC₅₀ of 550 mg Tordon/L (95% CI 470-740) and 120-hour EC₂₅ of 430 mg Tordon/L (95% CI 290-640). The associated NOEC at 120 hours was 390 mg Tordon/L. These Tordon K effect levels are equivalent to 132.5, 103.6, and 94.0 mg a.e./L for EC₅₀, EC₂₅, and NOEC, respectively. Acute toxicity testing of Tordon K (24.1% picloram acid) using *Navicula pelliculosa*, a freshwater diatom, suggests that this species is more sensitive to the acute toxic effects of picloram than *Anabaena flos-aquae* (Boeri et al. 1994c). Boeri et al. (1994c) reports a 120-hour EC₅₀ of 3.9 mg Tordon/L (95% CI 2.0-7.8) and 120-hour EC₂₅ of 1.3 mg Tordon/L (95% CI 0.42-3.7). The associated NOEC at 120 hours was 0.97 mg Tordon/L (the lowest concentration tested). These Tordon K effect levels are equivalent to 0.94, 0.31, and 0.23 mg a.e./L for EC₅₀, EC₂₅, and NOEC, respectively.

Additional studies on the toxicity of picloram to algae are available in the open literature. In a study designed primarily to assess the consistency between two different types of algal bioassays (the standard flask assay and a microplate assay), EC₅₀ values for picloram (form not specified) in Selenastrum capricornutum were 21.7 (18.4–25.1) mg/L in the flask assay and 22.7 (18.5–27.0) mg/L in the microplate assay (St-Laurent et al. 1992). This study does not report NOEC values but, based on the reported LC_{50} value in the flask assay, the toxicity of picloram as assayed in this study is comparable to that in the assay summarized by U.S. EPA (1995b), in which the EC_{25} for growth inhibition in this algal species was reported as 52.6 mg/L or about 15.9 mg a.e./L. Peterson et al. (1994) reported that picloram (form not specified) caused slight growth inhibition (2 to 12%) in some algae species as well as substantial growth inhibition in a cyanobacteria, *Microcystis aeruginosa.*, at a concentration of 1.76 mg/L. No inhibition was noted at this concentration in several other cyanobacteria species (Peterson et al. 1994). As with terrestrial microorganisms (Section 4.1.2.5), decarboxylation of picloram appears to increase toxicity to two species of Chlorella (algae), C. vulgaris and C. pyrenoidosa, with LC₅₀ values of greater than 160 mg a.e./L for the picloram and 8 ppm and 49 ppm for 4A-TCP, the decarboxylated metabolite of picloram, in C. pyrenoidosa and C. vulgaris, respectively (Baarschers et al. 1988).

Also since the preparation of the RED, an additional study has been submitted to U.S. EPA on the toxicity of Tordon K to *Lemna gibba*, an aquatic macrophyte (Kirk et al. 1994). This bioassay was required by U.S. EPA (1994b) because of the sensitivity of terrestrial macrophytes to picloram. The NOEC in this assay was 50.5 mg a.i./L corresponding to 43.5 mg a.e./L. The

 EC_{50} in this assay was 192.2 mg a.i./L (95% CI, 43.2 to 349), corresponding to 165.4 mg a.e./L (95% CI, 37.2 to 300). This is consistent with the report by Peterson et al. (1994) indicating that no statistically significant growth inhibition was observed in *Lemna gibba* at a concentration of 1.76 mg/L. Forsyth et al. (1997) assayed the effects of picloram on two other species of aquatic macrophytes: *Potamogeton pectinatus* (fennel pondweed) and *Myriophyllum sibiricum* (common water milfoil) at concentrations of 0.01 and 0.1 mg a.e/L. Picloram had no effect on growth in either species but inhibited flowering in *Myriophyllum sibiricum* at 0.1 mg/L 30-days post-treatment. This inhibition, however, was not statistically significant after 60 days post-treatment. No effects on flowering were observed in *Potamogeton pectinatus*.

As noted in Section 3.1.14, Oakes and Pollak (1999) noted that a commercial preparation of 2,4-D and picloram that contained Polyglycol 26-2 as well as Polyglycol 26-2 both inhibited oxidative function in submitochondrial particles derived from a marine algae. This information is difficult to assess quantitatively compared to the standard toxicity data on picloram or picloram formulations.

4.2. EXPOSURE ASSESSMENT

4.2.1. Overview. Terrestrial animals might be exposed to any applied herbicide from direct spray, the ingestion of contaminated media (vegetation, prey species, or water), grooming activities, or indirect contact with contaminated vegetation. In acute exposure scenarios, the highest exposures for small terrestrial vertebrates will occur after a direct spray and could reach up to about 85 mg/kg under typical exposure conditions and up to about 859 mg/kg under more extreme conditions. Substantially lower doses are anticipated from the consumption of contaminated vegetation: up to about 6 to 9 mg/kg under typical conditions with an upper range of 17 to 27 mg/kg. The consumption of contaminated water will generally lead to much lower levels of exposure. A similar pattern is seen for chronic exposures. Estimated daily doses for the a small vertebrate from the consumption of contaminated vegetation are in the range of 0.00006 to 3 mg/kg/day. The upper ranges of exposure from contaminated vegetation far exceed doses that are anticipated from the consumption of contaminated water, 0.000005 mg/kg/day to 0.0002 mg/kg/day. Based on general relationships of body size to body volume, larger vertebrates will be exposed to lower doses and smaller animals, such as insects, to much higher doses than small vertebrates under comparable exposure conditions. Because of the apparently low toxicity of picloram to animals, the rather substantial variations in the different exposure assessments have little impact on the assessment of risk to terrestrial animals.

For terrestrial plants, five exposure scenarios are considered quantitatively: direct spray, spray drift, runoff, wind erosion and the use of contaminated irrigation water. Unintended direct spray is expressed simply as the application rates considered in this risk assessment, 0.35 lb a.e./acre and should be regarded as an extreme/accidental form of exposure that is not likely to occur in most Forest Service applications. Estimates for the other routes of exposure are much less. All of these exposure scenarios are dominated by situational variability because the levels of exposure are highly dependent on site-specific conditions. Thus, the exposure estimates are intended to represent conservative but plausible ranges that could occur but these ranges may over-estimate or under-estimate actual exposures in some cases. Spray drift is based on estimates AGDRIFT. The proportion of the applied amount transported off-site from runoff is based on GLEAMS modeling of clay, loam, and sand. The amount of picloram that might be transported off-site from wind erosion is based on estimates of annual soil loss associated with wind erosion and the assumption that the herbicide is incorporated into the top 1 cm of soil. Exposure from the use of contaminated irrigation water is based on the same data used to estimate human exposure from the consumption of contaminated ambient water and involves both monitoring studies as well as GLEAMS modeling.

Exposures to aquatic plants and animals is based on essentially the same information used to assess the exposure to terrestrial species from contaminated water. The peak estimated rate of contamination of ambient water associated with the normal application of picloram is 0.05 (0.01 to 0.2) mg a.e./L at an application rate of 1 lb a.e./acre. For longer-term exposures, average estimated rate of contamination of ambient water associated with the normal application of picloram is 0.001 (0.0001 to 0.004) mg a.e./L at an application rate of 1 lb a.e./acre. For longer-term exposures, average

assessment of potential hazards, these contamination rates are adjusted based on the application rates considered in this risk assessment.

4.2.2. Terrestrial Animals. Terrestrial animals might be exposed to any applied herbicide from direct spray, the ingestion of contaminated media (vegetation, prey species, or water), grooming activities, or indirect contact with contaminated vegetation.

In this exposure assessment, estimates of oral exposure are expressed in the same units as the available toxicity data. As in the human health risk assessment, these units are usually expressed as mg of agent per kg of body weight and abbreviated as mg/kg. For dermal exposure, the units of measure usually are expressed in mg of agent per cm of surface area of the organism and abbreviated as mg/cm². In estimating dose, however, a distinction is made between the exposure dose and the absorbed dose. The *exposure dose* is the amount of material on the organism (i.e., the product of the residue level in mg/cm² and the amount of surface area exposed), which can be expressed either as mg/organism or mg/kg body weight. The *absorbed dose* is the proportion of the exposure dose that is actually taken in or absorbed by the animal.

The exposure assessments for terrestrial animals are summarized in Worksheet G01. As with the human health exposure assessment, the computational details for each exposure assessment presented in this section are provided scenario specific worksheets (Worksheets F01 through F14). Given the large number of species that could be exposed to herbicides and the varied diets in each of these species, a very large number of different exposure scenarios could be generated. For this generic – i.e., not site- or species-specific – risk assessment, an attempt is made to limited the number of exposure scenarios.

Because of the relationship of body weight to surface area as well as the consumption of food and water, small animals will generally receive a higher dose, in terms of mg/kg body weight, than large animals will receive for a given type of exposure. Consequently, most general exposure scenarios for mammals and birds are based on a small mammal or bird. For mammals, the body weight is taken as 20 grams, typical of mice, and exposure assessments are conducted for direct spray (F01 and F02a), consumption of contaminated fruit (F03, F04a, F04b), and contaminated water (F05, F06, F07). Grasses will generally have higher concentrations of herbicides than fruits and other types of vegetation (Fletcher et al. 1994; Hoerger and Kenaga 1972). Because small mammals do not generally consume large amounts of grass, the scenario for the assessment of contaminated grass is based on a large mammal – a deer (Worksheets F10, F11a, and F11b). Exposure scenarios for birds involve the consumption of contaminated insects by a small bird (Worksheet F14), the consumption of contaminated fish by a predatory bird (Worksheets F08 and F09), and the consumption of contaminated grasses by a large bird (F12, F13a, and F13b).

While a very large number of other exposure scenarios could be generated, the specific exposure scenarios developed in this section are designed as conservative screening scenarios that may

serve as guides for more detailed site-specific assessments by identifying the groups and routes of exposure that are of greatest concern.

4.2.2.1. Direct Spray – In the broadcast application of any herbicide, wildlife species may be sprayed directly. This scenario is similar to the accidental exposure scenarios for the general public discussed in Section 3.2.3.2. In a scenario involving exposure to direct spray, the amount absorbed depends on the application rate, the surface area of the organism, and the rate of absorption.

For this risk assessment, three groups of direct spray exposure assessments are conducted. The first, which is defined in Worksheet F01, involves a 20 g mammal that is sprayed directly over one half of the body surface as the chemical is being applied. The range of application rates as well as the typical application rate is used to define the amount deposited on the organism. The absorbed dose over the first day (i.e., a 24-hour period) is estimated using the assumption of first-order dermal absorption. In the absence of any data regarding dermal absorption in a small mammal, the estimated absorption rate for humans is used (see Section 3.1.12). An empirical relationship between body weight and surface area (Boxenbaum and D'Souza 1990) is used to estimate the surface area of the animal. The estimates of absorbed doses in this scenario may bracket plausible levels of exposure for small mammals based on uncertainties in the dermal absorption rate of picloram.

Other, perhaps more substantial, uncertainties affect the estimates for absorbed dose. For example, the estimate based on first-order dermal absorption does not consider fugitive losses from the surface of the animal and may overestimate the absorbed dose. Conversely, some animals, particularly birds and mammals, groom frequently, and grooming may contribute to the total absorbed dose by direct ingestion of the compound residing on fur or feathers. Furthermore, other vertebrates, particularly amphibians, may have skin that is far more permeable than the skin of most mammals (Moore 1964). Quantitative methods for considering the effects of grooming or increased dermal permeability are not available. As a conservative upper limit, the second exposure scenario, detailed in Worksheet F02, is developed in which complete absorption over day 1 of exposure is assumed.

Because of the relationship of body size to surface area, very small organisms, like bees and other terrestrial insects, might be exposed to much greater amounts of picloram per unit body weight, compared with small mammals. Consequently, a third exposure assessment is developed using a body weight of 0.093 g for the honey bee (USDA/APHIS 1993). Because there is no information regarding the dermal absorption rate of picloram by bees or other invertebrates, this exposure scenario, detailed in worksheet F02b, also assumes complete absorption over the first day of exposure.

Direct spray scenarios are not given for large mammals. As noted above, allometric relationships dictate that large mammals will be exposed to lesser amounts of a compound in any direct spray scenario than smaller mammals. As detailed further in Section 4.4, the direct spray scenarios for

the small mammal are substantially below a level of concern. Consequently, elaborating direct spray scenarios for a large mammal would have no impact on the characterization of risk.

4.2.2.2. *Indirect Contact* – As in the human health risk assessment (see Section 3.2.3.3), the only approach for estimating the potential significance of indirect dermal contact is to assume a relationship between the application rate and dislodgeable foliar residue. The study by Harris and Solomon (1992) (Worksheet A04) is used to estimate that the dislodgeable residue will be approximately 10 times less than the nominal application rate.

Unlike the human health risk assessment in which transfer rates for humans are available, there are no transfer rates available for wildlife species. As discussed in Durkin et al. (1995), the transfer rates for humans are based on brief (e.g., 0.5 to 1-hour) exposures that measure the transfer from contaminated soil to uncontaminated skin. Wildlife, compared with humans, are likely to spend longer periods of time in contact with contaminated vegetation.

It is reasonable to assume that for prolonged exposures a steady-state may be reached between levels on the skin, rates of absorption, and levels on contaminated vegetation, although there are no data regarding the kinetics of such a process. The bioconcentration data on picloram (Section 3.2.3.5) as well as the estimated rates of dermal absorption in humans (Section 3.1.12) suggest that picloram is not likely to partition from the surface of contaminated vegetation to the surface of skin, feathers, or fur. Thus, a plausible partition coefficient is unity (i.e., the concentration of the chemical on the surface of the animal will be equal to the dislodgeable residue on the vegetation).

Under these assumptions, the absorbed dose resulting from contact with contaminated vegetation will be one-tenth that associated with comparable direct spray scenarios. As discussed in the risk characterization for ecological effects (Section 4.4), the direct spray scenarios result in exposure levels below the estimated NOAEL (i.e., hazard quotients below one). Consequently, details of the indirect exposure scenarios for contaminated vegetation are not further elaborated in this document.

4.2.2.3. *Ingestion of Contaminated Vegetation or Prey* – Since picloram will be applied to vegetation, the consumption of contaminated vegetation is an obvious concern and separate exposure scenarios are developed for acute and chronic exposure scenarios for a small mammal (Worksheets F04a and F04b) and large mammal (Worksheets F10, F11a, and F11b) as well as large birds (Worksheets F12, F13a, and F13b).

A small mammal is used because allometric relationships indicate that small mammals will ingest greater amounts of food per unit body weight, compared with large mammals. The amount of food consumed per day by a small mammal (i.e., an animal weighing approximately 20 g) is equal to about 15% of the mammal's total body weight (U.S. EPA/ORD 1989). When applied generally, this value may overestimate or underestimate exposure in some circumstances. For example, a 20 g herbivore has a caloric requirement of about 13.5 kcal/day. If the diet of the

herbivore consists largely of seeds (4.92 kcal/g), the animal would have to consume a daily amount of food equivalent to approximately 14% of its body weight [(13.5 kcal/day \div 4.92 kcal/g) \div 20g = 0.137]. Conversely, if the diet of the herbivore consists largely of vegetation (2.46 kcal/g), the animal would have to consume a daily amount of food equivalent to approximately 27% of its body weight [(13.5 kcal/day \div 2.46 kcal/g) \div 20g = 0.274] (U.S. EPA/ORD 1993, pp.3-5 to 3-6). For this exposure assessment, the amount of food consumed per day by a small mammal weighing 20 g is estimated at about 3.6 g/day or about 18% of body weight per day from the general allometric relationship for food consumption in rodents (U.S. EPA/ORD 1993, p. 3-6).

A large herbivorous mammal is included because empirical relationships of concentrations of pesticides in vegetation, discussed below, indicate that grasses may have substantially higher pesticide residues than other types of vegetation such as forage crops or fruits (Worksheet A04). Grasses are an important part of the diet for some large herbivores, but most small mammals do not consume grasses as a substantial proportion of their diet. Thus, even though using residues from grass to model exposure for a small mammal is the most conservative approach, it is not generally applicable to the assessment of potential adverse effects. Hence, in the exposure scenarios for large mammals, the consumption of contaminated range grass is modeled for a 70 kg herbivore, such as a deer. Caloric requirements for herbivores and the caloric content of vegetation are used to estimate food consumption based on data from U.S. EPA/ORD (1993). Details of these exposure scenarios are given in worksheets F10 for acute exposures as well as Worksheets F11a and F11b for longer-term exposures.

For the acute exposures, the assumption is made that the vegetation is sprayed directly – i.e., the animal grazes on site – and that100% of the animals diet is contaminated. While appropriately conservative for acute exposures, neither of these assumptions are plausible for longer-term exposures. Thus, for the longer-term exposure scenarios for the large mammal, two subscenarios are given. The first is an on-site scenario that assumes that a 70 kg herbivore consumes short grass for a 90 day period after application of the chemical. In the worksheets, the contaminated vegetation is assumed to account for 30% of the diet with a range of 10% to 100% of the diet. These are essentially arbitrary assumptions reflecting grazing time at the application site by the animal. Because the animal is assumed to be feeding at the application site, drift is set to unity - i.e., direct spray. This scenario is detailed in Worksheet 11a. The second sub-scenario is similar except the assumption is made that the animal consumes 100% of the diet from the contaminated area (increasing risk). For this scenario, detailed in Worksheet F12b, AgDRIFT is used to estimate deposition on the off-site vegetation. Drift estimates from AgDrift are summarized in Worksheet A06 and this model is discussed further in Section 4.2.3.2.

The consumption of contaminated vegetation is also modeled for a large bird. For these exposure scenarios, the consumption of range grass by a 4 kg herbivorous bird, like a Canada Goose, is modeled for both acute (Worksheet F12) and chronic exposures (Worksheets F13a and

F13b). As with the large mammal, the two chronic exposure scenarios involve sub-scenarios for on-site as well as off-site exposure.

For this component of the exposure assessment, the estimated amounts of pesticide residue in vegetation are based on the relationship between application rate and residue rates on different types of vegetation. As summarized in Worksheet A04, these residue rates are based on estimated residue rates from Fletcher et al. (1994).

Similarly, the consumption of contaminated insects is modeled for a small (10g) bird. No monitoring data have been encountered on the concentrations of picloram in insects after applications of picloram. The empirical relationships recommended by Fletcher et al. (1994) are used as surrogates as detailed in Worksheet F14. To be conservative, the residue rates from small insects are used – i.e., 45 to 135 ppm per lb/ac – rather than the residue rates from large insects – i.e., 7 to 15 ppm per lb/ac.

In addition to the consumption of contaminated vegetation and insects, picloram may reach ambient water and fish. Thus, a separate exposure scenario is developed for the consumption of contaminated fish by a predatory bird in both acute (Worksheet F08) and chronic (Worksheet F09) exposures. Because predatory birds usually consume more food per unit body weight than do predatory mammals (U.S. EPA 1993, pp. 3-4 to 3-6), separate exposure scenarios for the consumption of contaminated fish by predatory mammals are not developed.

4.2.2.4. Ingestion of Contaminated Water – Estimated concentrations of picloram in water are identical to those used in the human health risk assessment (Worksheet B06). The only major differences involve the weight of the animal and the amount of water consumed. There are well-established relationships between body weight and water consumption across a wide range of mammalian species (e.g., U.S. EPA 1989). Mice, weighing about 0.02 kg, consume approximately 0.005 L of water/day (i.e., 0.25 L/kg body weight/day). These values are used in the exposure assessment for the small (20 g) mammal. Unlike the human health risk assessment, estimates of the variability of water consumption are not available. Thus, for the acute scenario, the only factors affecting the variability of the ingested dose estimates include the field dilution rates (i.e., the concentration of the chemical in the solution that is spilled) and the amount of solution that is spilled. As in the acute exposure scenario for the human health risk assessment, the amount of the spilled solution is taken as 200 gallons. In the exposure scenario involving contaminated ponds or streams due to contamination by runoff or percolation, the factors that affect the variability are the water contamination rate, (see Section 3.2.3.4.2) and the application rate. Details regarding these calculations are summarized in Worksheets F06 and Worksheet F07.

4.2.3. Terrestrial Plants. In general, the primary hazard to non-target terrestrial plants associated with the application of most herbicides is unintended direct deposition or spray drift. In addition, herbicides may be transported off-site by percolation or runoff or by wind erosion of soil.

4.2.3.1. Direct Spray – Unintended direct spray will result in an exposure level equivalent to the application rate. For many types of herbicide applications – e.g., rights-of-way management – it is plausible that some non-target plants immediately adjacent to the application site could be sprayed directly. This type of scenario is modeled in the human health risk assessment for the consumption of contaminated vegetation.

4.2.3.2. *Off-Site Drift* – Because off-site drift is more or less a physical process that depends on droplet size and meteorological conditions rather than the specific properties of the herbicide, estimates of off-site drift can be modeled using AgDRIFT (Teske et al. 2001). AGDRIFT is a model developed as a joint effort by the EPA Office of Research and Development and the Spray Drift Task Force, a coalition of pesticide registrants. AGDRIFT is based on the algorithms in FSCBG (Teske and Curbishley. 1990), a drift model previously used by USDA.

For aerial applications, AGDRIFT permits very detailed modeling of drift based on the chemical and physical properties of the applied product, the configuration of the aircraft, as well as wind speed and temperature. For ground applications, AGDRIFT provides estimates of drift based solely on distance downwind as well as the types of ground application: low boom spray, high boom spray, and orchard airblast. Representative estimates based on AGDRIFT (Version 1.16) are given in Worksheet A06. For the current risk assessment, the AGDRIFT estimates are used for consistency with comparable exposure assessments conducted by the U.S. EPA. In addition, AGDRIFT represents a detailed evaluation of a very large number of field studies and is likely to provide more reliable estimates of drift. Further details of AGDRIFT are available at <u>http://www.agdrift.com/.</u>

Estimates of drift for ground and aerial applications is given in Worksheet A06. In ground broadcast applications, picloram will typically be applied by low boom ground spray and thus these estimates are used in the current risk assessment.

Drift associated with backpack (directed foliar applications) are likely to be much less although studies quantitatively assessing drift after backpack applications have not been encountered. Drift distance can be estimated using Stoke's law, which describes the viscous drag on a moving sphere. According to Stoke's law:

$$v = \frac{D^2 \cdot g}{18n}$$

or
$$v = 2.87 \cdot 10^5 \cdot D^2$$

where v is the velocity of fall (cm sec⁻¹), D is the diameter of the sphere (cm), g is the force of gravity (980 cm sec⁻²), and n is the viscosity of air (1.9 \cdot 10⁻⁴ g sec⁻¹ cm⁻¹ at 20°C) (Goldstein et al. 1974).

In typical backpack ground sprays, droplet sizes are greater than 100μ , and the distance from the spray nozzle to the ground is 3 feet or less. In mechanical sprays, raindrop nozzles might be used. These nozzles generate droplets that are usually greater than 400 μ , and the maximum distance above the ground is about 6 feet. In both cases, the sprays are directed downward.

Thus, the amount of time required for a 100 μ droplet to fall 3 feet (91.4 cm) is approximately 3.2 seconds,

$$91.4 \div (2.87 \cdot 10^{5} (0.01)^{2}).$$

The comparable time for a 400 μ droplet to fall 6 feet (182.8 cm) is approximately 0.4 seconds,

$$182.8 \div (2.87 \cdot 10^5 (0.04)^2).$$

For most applications, the wind velocity will be no more than 5 miles/hour, which is equivalent to approximately 7.5 feet/second (1 mile/hour = 1.467 feet/second). Assuming a wind direction perpendicular to the line of application, 100 μ particles falling from 3 feet above the surface could drift as far as 23 feet (3 seconds \cdot 7.5 feet/second). A raindrop or 400 μ particle applied at 6 feet above the surface could drift about 3 feet (0.4 seconds \cdot 7.5 feet/second).

For backpack applications, wind speeds of up to 15 miles/hour are allowed in Forest Service programs. At this wind speed, a 100 μ droplet can drift as far as 68 feet (3 seconds \cdot 15 \cdot 1.5 feet/second). Smaller droplets will of course drift further, and the proportion of these particles in the spray as well as the wind speed and turbulence will affect the proportion of the applied herbicide that drifts off-site.

4.2.3.3. Runoff – Picloram or any other herbicide may be transported to off-site soil by runoff or percolation. Both runoff and percolation are considered in estimating contamination of ambient water. For assessing off-site soil contamination, however, only runoff is considered. This approach is reasonable because off-site runoff will contaminate the off-site soil surface and could impact non-target plants. Percolation, on the other hand, represents the amount of the herbicide that is transported below the root zone and thus may impact water quality but should not affect off-site vegetation.

Based on the results of the GLEAMS modeling (Section 3.2.3.4.2), the proportion of the applied picloram lost by runoff was estimated for clay, loam, and sand at rainfall rates ranging from 5 inches to 250 inches per year. These results are summarized in Worksheet G04 and indicate that runoff will be negligible in relatively arid environments as well as sandy or loam soils. In clay soils, which have the highest runoff potential, off-site loss may reach up to about 55% of the applied amount in regions with very high rainfall rates.

4.2.3.4. Contaminated Irrigation Water – Unintended direct exposures of nontarget plant species may occur through the use of contaminated ambient water for irrigation. As discussed

further in Section 4.4.2.2, Bovey and Scifres (1971) have discussed the potential effects on irrigation water contaminated with picloram on alfalfa and effects on non-target vegetation have been observed irrigation water contaminated with other herbicides (e.g., Bhandary et al. 1997; Gomez de Barreda et al. 1993).

The levels of exposure associated with this scenario will depend on the concentration of picloram in the ambient water used for irrigation and the amount of irrigation water that is applied. As detailed in Section 3.2.3.4, picloram is relatively mobile and contamination of ambient water may be anticipated and can be quantified (i.e., 0.05 [0.01 to 0.2] mg a.e./L at an application rate of 1 lb a.e./acre [Worksheet B07]).

The amount of irrigation water that may be applied will be highly dependent on the climate, soil type, topography, and plant species under cultivation. Thus, the selection of an irrigation rate is somewhat arbitrary. Typically, plants require 0.1 to 0.3 inch of water per day (Delaware Cooperative Extension Service 1999). In the absence of any general approach of determining and expressing the variability of irrigation rates, the application of one inch of irrigation water will be used in this risk assessment. This is somewhat higher than the maximum daily irrigation rate for sandy soil (0.75 inches/day) and substantially higher than the maximum daily irrigation rate for clay (0.15 inches/day) (Delaware Cooperative Extension Service 1999). This variability is addressed further in the risk characterization (Section 4.4.2.2).

Based on the estimated concentrations of picloram in ambient water and an irrigation rate of 1 inch per day, the estimated functional application rate of picloram to the irrigated area is 8×10^{-6} ($8 \times 10^{-7} - 3 \times 10^{-5}$) lb a.e./acre (see Worksheet F15 for details of these calculations). As discussed in the risk characterization, this level of exposure is inconsequential relative to off-site drift and runoff.

4.2.3.5. *Wind Erosion* – Wind erosion is a major transport mechanism for soil (e.g., Winegardner 1996). Although no specific incidents of nontarget damage from wind erosion have been encountered in the literature for picloram, this mechanism has been associated with the environmental transport of other herbicides (Buser 1990). Numerous models have been developed for wind erosion (e.g., Strek and Spaan 1997; Strek and Stein 1997) and the quantitative aspects of soil erosion by wind are extremely complex and site specific. Field studies conducted on agricultural sites found that wind erosion may account for annual soil losses ranging from 2 to 6.5 metric tons/ha (Allen and Fryrear 1977). The upper range reported by Allen and Fryrear (1977) is nearly the same as the rate of 2.2 tons/acre (5.4 tons/ha) recently reported by the USDA (1998). The temporal sequence of soil loss (i.e., the amount lost after a specific storm event involving high winds) depends heavily on soil characteristics as well as meteorological and topographical conditions.

To estimate the potential transport of picloram by wind erosion, this risk assessment uses average soil losses ranging from 1 to 10 tons/ha year, with a typical value of 5 tons/ha year. The value of 5 tons/ha year is equivalent to 500 g/m² (1 ton=1000 kg and 1 ha = 10,000 m²) or 0.05 g/cm²

 $(1m^2=10,000 \text{ cm}^2)$. Using a soil density of 2 g/cm³, the depth of soil removed from the surface per year would be 0.025 cm [(0.05 g/cm²)÷ (2 g/cm³)]. The average amount per day would be about 0.00007 cm/day (0.025 cm per year ÷ 365 days/year). This central estimate is based on a typical soil loss rate of 5 tons/ha·year. Since the range of plausible rates of annual soil loss is 1 to 10 tons/ha·year, the range of soil loss per day may be calculated as 0.00001 cm/day (0.00007÷5 = 0.000014) to 0.0001 cm/day (0.00007×2 = 0.00014).

The amount of picloram that might be transported by wind erosion depends on several factors, including the application, the depth of incorporation into the soil, the persistence in the soil, the wind speed, and the topographical and surface conditions of the soil. Under desirable conditions, like relatively deep (10 cm) soil incorporation, low wind speed, and surface conditions that inhibit wind erosion, it is likely that wind transport of picloram would be neither substantial or nor significant. For this risk assessment, it will be assumed that picloram is incorporated into the top 1 cm of soil. Thus, daily soil losses expressed as a proportion of applied amount would be 0.00007 with a range of 0.00001 to 0.001.

As with the deposition of picloram in runoff, the deposition of the picloram contaminated soil from wind erosion will vary substantially with local conditions and, for this risk assessment, neither concentration nor dispersion is considered quantitatively. Nonetheless, these factors together with the general and substantial uncertainties in the exposure assessment are considered in the risk characterization (see Section 4.4).

4.2.4. Aquatic Organisms. The potential for effects on aquatic species are based on estimated concentrations of picloram in water that are identical to those used in the human health risk assessment (Worksheet B06). As summarized in Worksheet B06, the peak estimated rate of contamination of ambient water associated with the normal application of picloram is 0.05 (0.01 to 0.2) mg a.e./L at an application rate of 1 lb a.e./acre. For longer-term exposures, average estimated rate of contamination of ambient water associated with the normal application of picloram is 0.001 (0.0001 to 0.004) mg a.e./L at an application rate of 1 lb a.e./acre. For longer-term exposures, average estimated rate of contamination of ambient water associated with the normal application of picloram is 0.001 (0.0001 to 0.004) mg a.e./L at an application rate of 1 lb a.e./acre. For the assessment of potential hazards, these contamination rates are adjusted based on the application considered in this risk assessment – i.e., 0.35 lb a.e./acre.

4.3. DOSE-RESPONSE ASSESSMENT

4.3.1. Overview. For terrestrial mammals, the dose-response assessment for chronic exposure is based on a NOAEL of 7 mg/kg/day from a 6-month dog feeding study. For acute exposures, a NOAEL of 34 mg/kg/day is used based on a teratogenicity study in rabbits. For birds, short term feeding studies are used to estimate an acute NOAEL of 1,500 mg/kg body weight. No adequate data are available on chronic toxicity in birds and the chronic NOAEL of 7 mg/kg/day derived for mammals is used to characterize risk. Relatively little data is available on terrestrial invertebrates and a standard LD₅₀ value of >1000 mg/kg in bees is used to characterize risk in terrestrial invertebrates.

For assessing the potential consequences of exposures to nontarget plants via runoff, a NOEC of 0.000012 lb a.e./acre is used for sensitive species and a NOEC of 0.4 lb a.e./acre is used for tolerant species. For assessing the impact of drift, bioassays on vegetative vigor are used with a NOEC of 0.00021 lb a.e./acre for sensitive species and a NOEC of 0.062 lb a.e./acre for tolerant species.

Soil microorganisms may display detectable responses to picloram at relatively low concentrations and true NOEC values for effects on microorganisms are not available. For this risk assessment, a soil concentration of 1 ppm is used as benchmark dose and the potential consequences of soil contamination by picloram is considered further in the risk characterization.

The general dose-response assessment for aquatic species is characterized by substantial variability within different groups (fish, invertebrates, and plants) but few substantial difference among the different groups. In general, sensitive species have NOECs or LC_{50} values in the 0.2 to 4 mg/L range and tolerant species have NOECs or LC_{50} values in the 10 to over 100 mg/L range. Trout appear to be the most sensitive animal species, with acute LC_{50} values as low as 0.8 mg/L. The dose response assessment for aquatic species is complicated by a very low reported LOEC, 0.035 mg/L in lake trout. This is an older study that was not designed to meet current standards. Nonetheless, this study is well documented and is not discounted. A relatively low LOEC of 0.1 mg/L has also been reported in on species of macrophyte. The observed effect was a transient delay in flowering with no inhibition of growth.

4.3.2. Toxicity to Terrestrial Organisms.

4.3.2.1. *Mammals* – As summarized in Section 4.1.2.1, data are available on experimental mammals that permit the derivation of both acute and chronic NOAEL values that can be used for risk characterization. Most acute exposure scenarios developed in this risk assessment involve an exposure period of one day. The U.S. EPA (1995b) used an acute LD_{50} value to characterize effects in mammalian wildlife. In the current risk assessment, an attempt is made to avoid the use of LD_{50} and LC_{50} values for risk characterization and rely instead on NOAEL values. The available teratology studies on picloram (Section 3.1.9) all involve gavage exposures conducted over a period of about 10 days. The NOAEL for either material toxicity or toxicity to offspring is 34 mg a.e./kg/day in rabbits (John et al. 1984) and 150 mg a.e./kg/day in rats (Schroeder 1990). The lower NOAEL of 34 mg a.e./kg/day in rabbits from the study by John et al. (1984) is used to characterize risks of acute exposure.

For chronic exposures, NOAEL values are available in rats and dogs. As discussed in Section 4.1.2, a dietary NOAEL of 20 mg/kg/day and a corresponding LOAEL of 60 mg/kg/day were observed in a two-year feeding study in male and female Fischer rats using picloram acid (Landry et al. 1986). In dogs, a corresponding NOAEL of 7 mg/kg/day with a corresponding LOAEL of 35 mg/kg/day was noted in a 6-month feeding study (Barna-Lloyd et al. 1982). For this risk assessment, the NOAEL of 7 mg/kg/day is used to characterize risks from chronic exposures.

The selections of the acute and chronic NOAEL values is intended to be protective and it could be argued that the selections are overly protective. For example, the difference between the acute NOAEL is rats and rabbits is about a factor of 5. Since both rats and rabbits are wildlife species of concern, the application of separate NOAEL values for these and related species could be considered. In some cases, systematic and substantial differences across species can be well documented and related to body size (e.g., Boxenbaum and D'Souza 1990). For picloram, the larger species – i.e., rabbits and dogs – do appear to be somewhat more sensitive than the smaller species, the rats, in both acute and chronic exposures. Nonetheless, given the large number of species that may need to be considered in various site-specific assessments, it seems prudent to use the more protective approach of selecting the most sensitive species for both acute and chronic exposures. As noted in Section 4.4.2, this protective approach has no impact on the characterization of risk in this risk assessment. Thus, unlike the approach taken with fish and plants, sub-groupings based on apparently sensitive and tolerant species is not used in this risk assessment.

4.3.2.2. *Birds* – As noted in Section 4.1.2.2, the acute and subchronic dietary toxicity studies in birds are not substantially different from the studies in mammals and there are no chronic toxicity studies in birds that would support the development of a separate chronic NOAEL for avian species. Consequently, for longer-term exposures, the NOAEL of 7 mg/kg/day will be used and this value is included in Worksheet G02 for the risk characterization of birds for longer term exposure scenarios.

For short-term exposures, the use of the 20 mg/kg/day NOAEL would be grossly conservative and tend to distort rather than clarify risk. In acute LD_{50} studies involving gavage applications – i.e., the administration of the compound directly into the stomach using a tube – doses of up to 2500 mg/kg body weight have not killed half of the animals – i.e., $LD_{50} > 2500$ mg/kg body weight. In 14-day feeding studies, the reported NOAEL is about 1500 mg/kg body weight (Section 4.1.2.2). These short term feeding studies are more closely analogous to environmental exposures than gavage studies and thus are more relevant to the risk assessment. Consequently, for risk characterization, the short-term NOAEL of 1,500 mg/kg body weight is used, as indicated in Worksheet G02.

4.3.2.3. Terrestrial Invertebrates – As discussed in Section 4.1.2.3, there is little information on the toxicity of picloram to terrestrial invertebrates compared to the more extensive studies available in mammals. The estimated acute contact LD_{50} value of greater than 1,000 mg/kg in bees is consistent with the available LD_{50} values on mammals and birds. Similarly, the 14-day dietary NOEL of 5000 mg/kg for the snail is similar to dietary NOAELs in rodents and birds. Thus, although the information is limited compared to the very large number of potential nontarget terrestrial invertebrates, there is no basis for suggesting that picloram is likely to be toxic to terrestrial invertebrates. For the purpose of characterizing risks, the reported LD_{50} of >1000 mg/kg/day will be used for risk characterization and is included in Worksheet G02.

4.3.2.4. Terrestrial Plants (Macrophytes) – As discussed in Section 4.1.2.4, picloram is more toxic to broadleaf plants than to grasses or grains. For assessing the potential consequences of exposures to nontarget plants via runoff, the EC₂₅ for seed emergence in soybeans of 0.000014 kg a.e./ha is used (equivalent to 0.000012 lb a.e./acre from U.S. EPA 1995b, p. 55). As discussed in Section 4.1.2.4, a more recent seedling emergence assay noted a much higher NOEC of 0.00027 kg a.i./hectare – the pinto bean from the study by Schwab (1995). The reason for the substantially lesser sensitivity of soybeans in the assay by Schwab (1995) relative to the earlier assay cited by U.S. EPA (1995b) – i.e., a factor of over 500 based on the EC₂₅ values – is not apparent. For this risk assessment, the NOEC of 0.000012 lb a.e./acre used by U.S. EPA (1995b) is maintained for assessing the impact of runoff and this value is used in Worksheet G04. The most tolerant plant species in seedling emergence assays appears to be corn, with an NOEC of 0.4 lb a.e./acre (Schwab 1995). This value is used in Worksheet G04 to characterize risks associated with runoff in tolerant plant species.

For assessing the impact of drift, bioassays on vegetative vigor will be used – i.e., studies in the herbicide is applied to the growing plant. In this type of assay, the most sensitive species appears to be the sunflower, with a NOEC of 0.27 g a.i./ha (Schwab 1996), which corresponds to an application rate of about 0.00021 lb a.e./acre [0.27 g a.i./ha \div 1000 g/kg \times 0.892 lb/acre per kg/ha \times 0.86 a.e./a.i.]. This value is used in Worksheets G05a and G05b for characterizing risks to sensitive plant species associated with drift. The highest reported NOEC from a vegetative vigor assay is 70 g a.e./ha for wheat (U.S. EPA 1995b, p. 55), which corresponds to an application rate of about 0.062 lb a.e./acre, and this value is used in Worksheets G05a and G05b for characterizing risks to tolerant plant species associated with drift.

As also indicated in Section 4.1.2.4, some plant species may develop resistence to picloram. While this may impact an assessment of the efficacy of picloram in some types of applications, the potential for resistence to picloram in nontarget plant species has not been documented and is not used to modify the risk characterization (Section 4.4.2.2).

4.3.2.5. *Terrestrial Microorganisms* – Picloram appears to be toxic to soil microorganisms under conditions of exposure that are analogous to applications used in Forest Service programs. Two different approaches may be used in the dose-response analysis and subsequent risk characterization for soil microorganisms – studies in which exposures are expressed as application rates or studies in which exposures are expressed as soil concentrations.

As summarized in Section 4.1.2.5, most of the data on the effects of picloram on soil microorganisms involve the latter type of studies and these studies will lead to a more conservative, and probably more realistic, assessment of risk. There does not appear to be a clear or defined threshold for the toxicity picloram to soil microorganisms. Concentrations of picloram in soil as low as 0.025 ppm appear to result in an increase in the persistence of picloram (USDA/ARD 1995) and this may be attributable to effects on microbial populations. It is less clear that this will result in remarkable changes in normal microbial functions in soil. Based on the study by Tu (1994), a concentration of 10 ppm results in only a transient change in only a transient decrease in nitrification – i.e., at 2 but not 3 weeks after incubation. Similarly, the study by Prado and Airoldi (2001) suggests that concentrations of 1 ppm in soil will only modestly inhibit microbial activity. There are not studies indicating that picloram will adversely effect mycorrhizal organisms resulting in secondary damage to plants. As noted in Section 4.1.2.4, the primary mechanism of action of picloram on terrestrial plants involves auxin activity leading to uncontrolled and abnormal growth.

For this risk assessment, a soil concentration of 1 ppm will be used as a benchmark to suggest concentrations that may result in a detectable effect on microbial populations in soil. While the toxicity of the metabolites of picloram may contribute to the effect of picloram on soil microorganisms (4.1.2.5), the effect of the metabolites should be encompassed by the available toxicity data on picloram in soil – i.e., metabolites are presumably formed in the available studies and it is the metabolites that may contribute most to the observed effects expressed in terms of initial concentrations of picloram.

4.3.3. Aquatic Organisms.

The toxicity values used in this risk assessment are summarized in Worksheet G03 based on the information presented in Section 4.1.3.

4.3.3.1. Fish – As summarized in Table 4-1 and detailed further in the review by Mayes and Oliver (1985), acute toxicity bioassays of picloram in fish are highly variable. Notwithstanding this variability, trout do generally appear to be more sensitive to picloram than other species. Thus, risk is characterized for both sensitive and tolerant fish species. For this risk assessment, the lower range of the lowest LC_{50} value for trout is used for the risk characterization – i.e., 0.8

mg/L in cutthroat trout reported in Mayes and Oliver (1985). Risk to more tolerant fish species is characterized using the highest reported LC_{50} value, 55.3 mg/L in fathead minnow (Mayes and Oliver 1985).

It should be noted that the use of LD_{50} and LC_{50} values for risk characterization is generally avoided because a common concern with this approach is that more subtle non-lethal effects, that may impact the stability of fish populations in the field, may not be properly assessed. In some respects, this concern is somewhat misguided. Most acute fish toxicity studies report the results as LC_{50} values and there are sound statistical reasons for this approach (e.g., Finney 1971). In addition, as used by the U.S. EPA/OPP, levels of concern for hazard quotients based on LC_{50} values may be as low as 0.05. In other words, if the expected exposure is equal to one-twentieth (0.05) of the LC_{50} , the Agency may judge that there is a cause for concern at least in sensitive or endangered species. This is essentially similar to the use of an uncertainty factor as in the human health risk assessments. For picloram, the sublethal effects associated with acute exposures must be assessed based on LC_{50} values because this is the only type of information that is available and this information is relevant for assessing risk, as discussed in Section 4.4.3.

Toxicity values for chronic toxicity may be based on the available egg-and-fry/early life stage studies. While these do not involve full life cycle testing, they encompass an exposure period of about 60-days during a sensitive period of development. This type of assay is used commonly by U.S. EPA (e.g., U.S. EPA 1995b) to characterize chronic toxicity in fish.

A major difficultly in the assessment of chronic effects in fish concerns the distinction between sensitive and tolerant species, particularly in the interpretation of the studies by Woodward (1976, 1979, 1982). The U.S. EPA (1995b) used the study by Mayes et al. (1984), with a NOEC of 0.55 mg/L in rainbow trout, to characterize chronic risks associated with exposure to the potassium salt of picloram. This is very similar to the chronic NOEC of 0.77 mg/L for the TIPA salt of picloram in fathead minnow. Taking this approach, no "sensitive" species would be identified for chronic effects in fish – i.e., the two NOEC values are virtually identical.

In the study by Woodward (1976), however, a different species of trout was tested, lake trout, and the reported LOEC is 0.035 mg/L, a factor of about 16 below the NOEC reported in rainbow trout by Mayes et al. (1984). As detailed in Section 4.1.3.1.2, the Woodward (1976) study may be criticized for not using a solvent control and not measuring the actual concentrations of picloram in the water. The failure to include a solvent control may be the most important criticism because a study is available on the toxicity and uptake of PCBs trout indicating that low concentrations of acetone may enhance absorption in trout during early life stages (Mac and Seelye (1981). Although the U.S. EPA (1995b) did not discuss the Woodward (1976) study, this study clearly does not meet current guidelines for fish toxicity bioassays.

Notwithstanding these reservations, the Woodward (1976) study cannot be discounted. The study appears to have been well conducted and is from a credible source - i.e., U.S. Fish and Wildlife Service. While flawed, one reasonable interpretation of the Woodward (1976) study is

that lake trout may be more sensitive than rainbow trout. In the absence of additional experimental data -i.e., a new matched bioassay in lake trout and rainbow trout - this interpretation cannot be discounted.

For the current risk assessment, both the study in rainbow trout Mayes et al. (1984) as well as the study in lake trout (Woodward 1976) are used to characterize the potential longer risks to salmonids.

4.3.3.2. *Invertebrates* – As with the acute toxicity data in fish, there appears to be substantial variation in the sensitivity of different invertebrate species to picloram. As detailed in Section 4.1.3.3, the lowest LC_{50} value is 26.8 mg a.e./L for pink shrimp (Heitmuller 1975). The highest LC_{50} value is 214 mg a.e./L for the fiddler crab (Heitmuller 1975). The LC_{50} value for *Daphnia magna*, a freshwater invertebrate often used in risk assessment, is 68.3 mg/L. For this risk assessment, the LC_{50} values for pink shrimp and fiddler crab are used to bracket a plausible range of sensitivities for invertebrates after acute exposures to picloram.

For chronic exposures, only two studies appear useful for assessing risk. The NOEC of 11.8 mg/L from standard life-cycle study in *Daphnia magna* by Gersich et al. (1985) was used by U.S. EPA (1995b) and will be adopted in this risk assessment for tolerant species. Heitmuller (1975) reports a 48 hour LC_{50} value in the range of 3.8-6.9 mg a.e./L for an embryo/larvae assay in oysters. While this is not a full life-cycle assay, the LC_{50} is below the NOEC for the life-cycle study in daphnids and suggests that oyster embryo/larvae may be more sensitive to picloram than daphnids. In the absence of any further information, the lower range of this LC_{50} , 3.8 mg a.e./L is used for assessing the effects of longer term exposures in sensitive invertebrates.

4.3.3.3. Aquatic Plants – As with both fish and invertebrates, there appear to be remarkable differences in the toxicity of picloram to various species of aquatic plants. Nonetheless, there appears to be relatively little difference in the toxicity of picloram to common test species of algae (microscopic plants) and macrophytes. For algae, the lowest NOEC is for a diatom, 0.23 mg a.e./L reported in *Anabaena flos-aquae* (Boeri et al. 1994c). The most tolerant algae appears to be a freshwater blue-green alga, *Anabaena flos-aquae*, with a NOEC of 94 mg a.e./L (Boeri et al. 1994b). These two NOEC values will be used in assessing potential effects in tolerant and sensitive species of algae. The highest reported NOEC in a macrophyte is 43.5 mg a.e./L reported in duckweed (U.S. EPA 1995b). The lowest reported effect level for any macrophyte is 0.1 mg/L for common water milfoil (Forsyth et al. 1997). This is classified as a LOEC because of a transient inhibition or delay in flowering but the exposure was not associated with any decrease in growth.

4.4. RISK CHARACTERIZATION

4.4.1. Overview. Picloram is an herbicide and the most likely damage to nontarget species will involve terrestrial plants. As is the case with any herbicide, the likelihood of damage to nontarget plant species is related directly to the difference between the sensitivity of target species—which dictates the application rate—and the sensitivity of the potential nontarget species. Sensitive plant species could be adversely affected by the off-site transport of picloram under a variety of different scenarios depending on local site-specific conditions that cannot be generically modeled. If picloram is applied in the proximity of sensitive crops or other desirable sensitive plant species, site-specific conditions and anticipated weather patterns will need to considered if unintended damage is to be avoided. More tolerant plant species are not likely to be affected unless they are directly sprayed or subject to substantial drift. A detectable inhibition of the activity of soil microorganisms is also likely at application rates used in Forest Service programs. These changes could lead to an increase in the persistence of picloram in soil and/or a more general decrease in microbial activity. That this inhibition would be associated with detectable changes in soil productivity or other undesirable gross effects is much less certain. The potential for adverse effects on other terrestrial nontarget animal species appears to be remote. The weight of evidence suggests that no adverse effects in terrestrial animals are plausible using typical or even very conservative worst case exposure assumptions.

There is substantial variability in the toxicity of picloram to aquatic species. While this variability adds uncertainty to the dose-response assessment, it has no substantial impact on the risk characterization. None of the hazard indices for fish, aquatic invertebrates, or aquatic plants exceed a level of concern.

The risk characterization for both terrestrial and aquatic species is limited by the relatively few animal and plant species on which data are available compared to the large number of species that could potentially be exposed. This limitation and consequent uncertainty is common to most if not all ecological risk assessments.

4.4.2. Terrestrial Organisms.

4.4.2.1. *Terrestrial Animals* – The quantitative risk characterization for terrestrial animals is summarized in Worksheet G02. The toxicity values used for each group of animals – mammals, birds, and insects – is summarized at the bottom of Worksheet G02 and refer to values derived in the dose-response assessment (Section 4.3).

As specified in Worksheet G02, both the central estimates as well as the upper range of the hazard quotients associated with the acute and longer-term exposure scenarios are below unity by at least a factor of 5, indicating that toxic effects attributable to picloram are not likely to occur. The highest hazard quotient is 0.2 - i.e., for the large bird consuming contaminated vegetation on-site. This scenario, as well as the similar exposure scenario for the large mammal consuming vegetation on-site, is essentially used in these risk assessments as a very conservative/extreme screening scenario. The scenarios assume that the vegetation is treated at the nominal application rate of 0.35 lb/acre and that the animal stays in the treated area consuming nothing but the

contaminated vegetation. Given that most forms of vegetation treated at the nominal application rate of 0.35 lb/acre would likely die or at least be substantially damaged (Section 4.4.2), this exposure scenario is implausible. It is, however, routinely used in Forest Service risk assessments as a very conservative upper estimate of potential exposures. Other more plausible exposure scenarios are below the level of concern by factors of about 16 (HQ=0.06) to 100,000 (HQ=1e-05 or 1×10^{-5}).

The simple verbal interpretation of this quantitative risk characterization is similar to that of the human health risk assessment: the weight of evidence suggests that no adverse effects in mammals are plausible using typical or even very conservative worst case exposure assumptions.

As with most ecological risk assessments, this characterization of risk must be qualified. Picloram has been tested in only a limited number of species and under conditions that may not well-represent populations of free-ranging nontarget animals. Notwithstanding this limitation, the available data are sufficient to assert that adverse effects in terrestrial animals from the use of this compound in Forest Service programs do not appear to be likely. Although based on somewhat different assessment methods and a much higher application rate (2.16 lb/acre), the conclusion reached in the current risk assessment is essentially identical to that presented by U.S. EPA (1995b) in the RED for picloram.

4.4.2.2. *Terrestrial Plants* – A quantitative summary of the risk characterization for terrestrial plants is presented in Worksheet G04 for runoff and Worksheets G05a and G05b for drift. Analogous to the approach taken for terrestrial animals, risk in these worksheets is characterized as a ratio of the estimated exposure to a benchmark exposure (i.e., exposure associated with a defined response). For both worksheets, the benchmark exposure is a NOEC, as derived in Section 4.3.2.2, for both sensitive and tolerant species.

Picloram is an effective herbicide, at least for a number of different broadleaf weeds, and adverse effects on some nontarget plant species due to drift are likely under certain application conditions and circumstances. As indicated in Worksheets G05a and G05b, off-site drift of picloram associated with ground and aerial applications may cause damage to sensitive plant species at distances of nearly 1000 feet from the application site. The closer that the non-target species is to the application site, the greater is the likelihood of damage. Whether or not damage due to drift would actually be observed after the application of picloram would depend on a several site-specific conditions, including wind speed and foliar interception. In other words, in some right-of-way applications conducted at low wind speeds and under conditions in which vegetation immediately adjacent to the application site would limit off-site drift, damage due to drift would probably be limited to the area immediately adjacent to the application site. Tolerant plant species would probably not be impacted by the drift of picloram and might show relatively little damage unless they were directly sprayed.

As summarized in Worksheet G04, runoff may present a significant risk to sensitive non-target plant species under conditions in which runoff is favored – i.e., clay soil over a very wide range

of rainfall rates. At sites in which runoff is less plausible -i.e., loam or sandy soils - effects on sensitive species are not likely.

The situational variability in the exposure assessments for runoff, wind erosion, and irrigation water does have a substantial impact on the characterization of risk for sensitive nontarget plant species. All of these scenarios may overestimate or underestimate risk under certain conditions. For example, the exposure conditions involving runoff and contaminated irrigation water are plausible for applications in which relatively substantial rainfall occurs shortly after application and in which local topographic and/or hydrological conditions favor either runoff or percolation.

Bovey and Scifres (1971) suggest that concentrations of picloram in water in the range of 0.001 to 0.004 mg/L are not likely to be associated with adverse effects but that multiple watering at concentrations of 0.004 mg/L could reduce growth and 0.01 mg/L could severely impact growth. This risk assessment supports this assertion. As indicated in Worksheet F15, peak concentrations of picloram in water are estimated at 0.0035 to 0.07 mg/L and functional application rates in the use of this water in irrigation would be in the range of 0.00008 to 0.002 lb/acre. Based on the NOEC values for seedling emergence in sensitive plants (0.000019 lb/acre) and tolerant plants (0.4 lb/acre), an inhibition of seedling emergence could be anticipated in some sensitive plant species. Based on the NOEC values for vegetative vigor in sensitive plants (0.00021 lb/acre) and tolerant plants (0.662 lb/acre), damage to growing plants could occur in sensitive but not tolerant plant species. The likelihood of observing these effects, however, may be remote. The highest concentrations of picloram in ambient water that might be used for irrigation are associated with high rainfall rates. In regions with high rainfall rates, the use of irrigation water would likely be less than that in arid regions.

As summarized in Section 4.2.3.5, daily soil losses due to wind erosion, expressed as a proportion of an application rate, could be in the range of 0.00001 to 0.001. As summarized in Worksheet G04, this is substantially less than off-site losses associated with runoff from clay but similar to off-site losses associated with drift in the range of about 200 feet to 900 feet. As with the drift scenarios, wind erosion could lead to adverse effects in sensitive plant species. Wind erosion of soil contaminated with picloram is most plausible in relatively arid environments and if local soil surface and topographic conditions favor wind erosion.

The simple verbal interpretation for this quantitative risk characterization is that sensitive plant species could be adversely affected by the off-site transport of picloram under a variety of different scenarios depending on local site-specific conditions that cannot be generically modeled. If picloram is applied in the proximity of sensitive crops or other desirable sensitive plant species, site-specific conditions and anticipated weather patterns will need to considered if unintended damage is to be avoided. More tolerant plant species are not likely to be affected unless they are directly sprayed or subject to substantial drift.

4.4.2.3. *Terrestrial Microorganisms* – Changes in soil microorganisms may be evident at very low concentrations in soil and 1 ppm is adopted in this risk assessment as a benchmark

associated with a modest inhibition of microbial activity, assayed as an apparent decrease and delay in glucose utilization. As summarized in Table 4-2, peak concentrations of picloram in soil at application rates used in Forest Service programs are likely to exceed this concentration. Longer term concentrations of picloram in soil will be much lower except in very arid climates – i.e., in the range of about 0.01 ppm in areas with substantial runoff or erosion. Even these lower concentrations, however, are associated with an increase in the persistence of picloram in soil, suggesting an inhibition of microbial metabolism and the persistence of picloram in soil has been shown to increase at application rates similar to those used in Forest Service programs – i.e., about 0.4 to 1.6 lbs per acre (Krzyszowska et al. 1994).

That such changes are likely to occur seems reasonably certain. The consequences of such effects, however, are far less certain. As noted in Section 4.1.3.1, field studies (Brooks et al. 1995; Nolte and Fulbright 1997) have not noted substantial adverse effects associated with the normal application of picloram that might be expected if soil microbial activity were substantially damaged. In addition, picloram has been used as an herbicide since 1964 (U.S. EPA 1995b). It does not seem plausible to assert that changes that might be anticipated in microbial populations after the application of picloram would have an adverse impact on soil productivity or other secondary changes that would lead to grossly detectable and significant effects in the environment.

4.4.3. Aquatic Organisms. There is substantial variability in the toxicity of picloram – i.e., within fish, aquatic invertebrates, and aquatic plants – that in some cases spans over an order of magnitude. An added complication in the risk assessment of picloram to aquatic species is the nature of the data on salmonids. As discussed at some length in Section 4.1.3.1, two sets of studies on the toxicity of picloram to trout are inconsistent and the study reporting the most sensitive response after longer term exposures, an LOEC of 0.035 mg/L (Woodward 1976), is a factor of about 16 below a NOEC reported in another species of trout (Mayes et al. 1984).

While this variability adds uncertainty to the dose-response assessment, it has no substantial impact on the risk characterization. As detailed in Worksheet G03, none of the hazard indices for fish or aquatic invertebrates exceed a hazard quotient of 0.09 at the upper range of plausible exposures and most hazard quotients are below the level of concern by factors of 100 to over 1000. For aquatic plants, the highest hazard quotient is 0.01 for longer-term exposures and 0.7 for short-term exposures. While there are always additional uncertainties associated with the use of general exposure assessments and while many different site specific conditions could impact exposure, the exposures assessments for aquatic species are based modeling data that are well-supported by monitoring data. While the fish-kill incident reported by Keys (1992) raises concern for the potential effects of a picloram on trout (Section 4.1.3.1.3), the very low hazard quotients for sensitive fish species (Worksheet G03) suggests that fish-kills associated with the normal application of picloram in Forest Service programs are implausible.

Thus, similar to the risk assessment of terrestrial animals, there seems no plausible basis for asserting that the use of picloram in Forest Service programs is likely to lead to adverse effects in

aquatic species. As with the risk characterization for terrestrial species, this risk characterization is limited by the relatively few animal and plant species on which data are available compared to the large number of species that could potentially be exposed. This limitation and consequent uncertainty is common to most if not all ecological risk assessments.

5. REFERENCES

Alexander HC; Batchelder TL. 1965. Results of a Study on the Acute Toxicity of Tordon^(R) Herbicide to Three Species of Fish.. (Unpublished study received Nov 6, 1967 under 0F0863; submitted by Dow Chemical U.S.A., Midland, Mich.; CDL:094524-H) MRID 00041475

Allen RR; Fryrear DW. 1977. Limited tillage saves soil, water, and energy. ASAE Annual Meeting, NC State Univ., Raleigh, NC. June 26-29, 1977. 14 pp.

ATSDR (Agency for Toxic Substances and Disease Registry). 2002. Toxicological Profile for Hexachlorobenzene. Available online at <u>http://www.atsdr.cdc.gov/toxprofiles/tp90.html</u>.

Baarschers WH; Donnelly JG; Heitland HS. 1988. Microbial toxicity of triclopyr and related herbicides. Toxic Assess. 3 (2):127-136.

Baker R. 1989a. Product Chemistry: Picloram. Unpublished study prepared by Dow Chemical U.S.A. 31 p. MRID 41094901.

Baker R. 1989b. Product Chemistry: Picloram. Unpublished study prepared by Dow Chemical U.S.A. 73 p. MRID 41094902.

Baker R. 1989c. Product Chemistry: Picloram. Unpublished study prepared by Dow Chemical U.S.A. 5 p. MRID 41094903.

Baker R. 1989d. Product Chemistry: Tordon K Salt Liquor. Unpublished study prepared by Dow Chemical U.S.A. 17 p. MRID 41094907.

Baker R. 1989e. Product Chemistry: Tordon K Salt Liquor. Unpublished study prepared by Dow Chemical U.S.A. 21 p. MRID 41094908.

Baker R. 1989f. Product Chemistry: Tordon K Salt Liquor. Unpublished study prepared by Dow Chemical U.S.A. 5 p. MRID 41094909.

Baker R. 1997. Discussion of the Formation of Impurities for Picloram Technical: Lab Project Number: GH-C 4404. Unpublished study prepared by DowElanco. 10 p. MRID 44278301

Barna-Lloyd T; Taylor H; Swaim L; et al. 1982. Results of a Six-month Dietary Toxicity Study of Picloram., Administered in the Diet to Male and Female Beagle Dogs: Study TXT:K-038323-(28). Final Report. (Unpublished study received Aug 23, 1982 under 464-502; submitted by Dow Chemical U.S.A., Midland, MI; CDL:248162-A). MRID 110534.

Beall ML Jr. 1976. Persistence of aerially applied hexachlorobenzene on grass and soil. J Environ Qual. 5:367-369.

Beavers J. 1983. An Acute Oral Toxicity Study in the Mallard with Picloram Technical: Final Report: Project No. 103-221. Unpublished study prepared by Wildlife International Ltd. 15 p. MRID 00157173.

Berdasco. 1990a. XRM-5179: Primary Eye Irritation Study in New Zealand White Rabbits: Final Report: Lab Project Number: M-005179-001C. Unpublished study prepared by the Dow Chemical Co. 14 p. MRID 4395606.

Berdasco. 1990b. XRM-5179: Acute Dermal Toxicity Study in New Zealand White Rabbits: Final Report: Lab Project Number: M-005179-001D. Unpublished study prepared by the Dow Chemical Co. 20 p. MRID 4395604.

Berdasco. 1990c. XRM-5179: Dermal Sensitization Potential in the Hartley Albino Guinea Pig: Final Report: Lab Project Number: M-005179-001E. Unpublished study prepared by the Dow Chemical Co. 14 p. MRID 4395609.

Bhandary RM; Whitwell T; Briggs J. 1997. Growth of containerized landscape plants is influenced by herbicide residues in irrigation water. Weed Technol. 11 (4):793-797.

Bidlack H. 1980a. Determination of the Bioconcentration Factor for Picloram in Bluegill Sunfish during Continuous Aqueous Exposure: GH-C 1384. Unpublished study received June 16, 1983 under 464-502; submitted by Dow Chemical U.S.A., Midland, MI; CDL:250517-A. MRID 00128947.

Bidlack H. 1980b. Kinetics of Aged Picloram in a Model Aquatic Microcosm: GH-C 1370. Unpublished study received June 16,1983 under 464-502; submitted by Dow Chemical U.S.A., Midland, MI; CDL:250517-B. MRID 00128948.

Bird SL. 1995. A compilation of aerial spray drift field study data for low-flight agricultural application of pesticides. Chapter 18 in Agrochemical Environmental Fate: State of the Art. ML Leng; EMK Leovey; PL Zubkoff (eds.). Lewis Publishers, Boca Raton, FL. pp. 195-207.

Blakley PM; Kim JS; Firneisz GD. 1989a. Effects of gestational exposure to Tordon 202c on fetal growth and development in CD-1 mice. J Toxicol Environ Health. 28 (3):309-316.

Blakley PM; Kim JS; Firneisz GD. 1989b. Effects of paternal subacute exposure to Tordon 202c on fetal growth and development in CD-1 mice. Teratology. 39 (3):237-241.

Blakley PM; Kim JS; Firneisz GD. 1989c. Effects of preconceptional and gestational exposure to Tordon 202c on fetal growth and development in CD-1 mice. Teratology. 39 (6):547-553.

Blakley BR. 1997. Effect of Roundup and Tordon 202C herbicides on antibody production in mice. Vet Hum Toxicol. 39 (4):204-206.

Boeri R; Kowalski P; Ward T. 1994a. Access Herbicide: Toxicity to the Freshwater Blue-Green Alga, Anabaena flos-aquae: Lab Project Number: 432/DO: ES/2768. Unpublished study prepared by T.R. Wilbury Laboratories, Inc. 25 p. MRID 43230307.

Boeri R; Kowalski P; Ward T. 1994b. Tordon K Herbicide: Toxicity to the Freshwater Blue-Green Alga, Anabaena flos-aquae: Lab Project Number: 438/DO: ES/2774. Unpublished study prepared by T.R. Wilbury Laboratories, Inc. 25 p. MRID 43230308.

Boeri R; Kowalski P; Ward T. 1994c. Tordon K Herbicide: Toxicity to the Freshwater Diatom, *Navicula pelliculosa*: Lab Project Number: 436/DO: ES/2772. Unpublished study prepared by T.R. Wilbury Laboratories, Inc. 24 p. MRID 43230302.

Boeri R; Kowalski P; Ward T. 1994d. Tordon K Herbicide: Toxicity to the Marine Diatom, Skeletonema costatum: Lab Project Number: 437/DO: ES/2773. Unpublished study prepared by T.R. Wilbury Laboratories, Inc. 24 p. MRID 43230305.

Boeri R; Magazu J; Ward T. 1994e. Tordon 101 Herbicide: Toxicity to the Freshwater Blue-Green Alga, Anabaena flos-aquae: Lab Project Number: 435/DO: ES/2771. Unpublished study prepared by T.R. Wilbury Laboratories, Inc. 24 p. MRID 43230309.

Boeri R; Magazu J; Ward T. 1994f. Tordon 101 Herbicide: Toxicity to the Freshwater Diatom, Navicula pelliculosa: Lab Project Number: 433/DO: ES/2769. Unpublished study prepared by T.R. Wilbury Laboratories, Inc. 25 p. MRID 43230303.

Boeri R; Magazu J; Ward T. 1995a. Tordon 101M Herbicide: Acute Toxicity to the Silverside, *Menidia beryllina*: Lab Project Number: DECO-ES-2927: 647-DO. Unpublished study prepared by T.R. Wilbury Labs, Inc. 26 p. MRID 43959503.

Boeri R; Magazu J; Ward T. 1995b. Tordon 22K Herbicide: Acute Toxicity to the Sheepshead Minnow, Cyprinodon variegatus: Lab Project Number: DECO-ES-2928: 651-DO. Unpublished study prepared by T.R. Wilbury Labs, Inc. 25 p. MRID 43959502.

Bovey RW; Scifres CJ. 1971. Residual characteristics of picloram in grassland ecosystems. Texas Agric Sation. B-1111. (Cited in Watson et al. 1989)

Boxenbaum J; D'Souza R. 1990. Interspecies pharmacokinetic scaling, biological design and neoteny. Adv Drug Res. 19:139-195.

Breslin W; Zielke G; Kociba R. 1991. Picloram: Two-Generation on Diet Reproduction Study in Sprague-Dawley Rats: Lab Project Number: K-038323-057F0: K-038323-057F1W: K-038323-057F1. Unpublished study prepared by The Toxicology Research Laboratory & The Dow Chemical Co. 1225 p. MRID 42078701.

Brooks JJ; Rodrigue JL; Cone MA; Miller KV; Chapman BR; Johnson AS. 1995. Small mammal and avian communities on chemically-prepared sites in the Georgia sandhills. General Technical Report - Southern Research Station, USDA Forest. (No. S) :21-23.

Brown A. 2002. Physical Properties of Picloram K Salt 2A: Final Report: Lab Project Number: MI-0223. Unpublished study prepared by Micro Flo Company. 11 p.. MRID 45675801

Budavari S; O'Neil MJ; Smith A; Heckelman PE (eds). 1989. The Merck Index : An Encyclopedia of Chemicals, Drugs, and Biologicals. 11th ed., Centennial ed. Rahway, N.J.

Buser HR. 1990. Atrazine and other s-triazine herbicides in lakes and in rain in Switzerland. Environ Sci Technol. 24 (7):1049-1058.

C&P Press (Chemical and Pharmaceutical Press). 1998. EPR II for Windows. Electronic Pesticide Reference.

Calabrese EJ; Baldwin LA. 1993. Performing Ecological Risk Assessments. Lewis Publishers, Boca Raton, LA. pp. 12-24.

Campbell MH; Nicol HI. 2000. Effect of split applications of herbicides on the control of Hypericum perforatum L. (St. John's wort) and regeneration of native grasses and annual clovers on non-arable land. Plant Prot Q. 15 (3):119-122.

Cantrell RL; Hyland JR. 1985. Application techniques. In: A Guide to Silvicultural Herbicide Use in the Southern United States. Auburn University School of Forestry, Alabama Agricultural Experiment Station. November 1985. 612 p.

Chaturvedi AK. 1993. Biochemical and toxicological studies on the mixtures of three commonly-used herbicides in mice. Arch Environ Contam Toxicol. 24 (4):449-454.

Close ME; Pang L; Watt JPC; Vincent KW. 1998. Leaching of picloram, atrazine and simazine through two New Zealand soils. Geoderma. 84 (1-3):45-63.

Close ME; Watt JPC; Vincent KW. 1999. Simulation of picloram, atrazine, and simazine transport through two New Zealand soils using LEACHM. Aust J Soil Res. 37 (1):53-74.

Cox C. 1998. Herbicide factsheet: Picloram. JPR. 18 (1):13-20.

Davis EA; Ingebo PA. 1973. Picloram movement from a chaparral watershed. Water Res. 9:1304-1312.

De Sapio R. 1976. Calculus for the Life Sciences. Freeman and Company, San Francisco, California. 740 p.

Delaware Cooperative Extension Service. 1999. Agronomy Facts Series: AF-03.

Diaz-Diaz R; Loague K. 2001. Assessing the potential for pesticide leaching for the pine forest areas of Tenerife. Environ Toxicol Chem. 20 (9):1958-1967.

Dinnage E. 1996. Series 62: Analysis and Certification of Product Ingredients of Tordon RTU, Forestry Tordon 101R and Pathway Herbicide: A Water Soluble Concentrate Formulation Containing Picloram Triisopropanolamine Salt and 2,4-D Triisopropanolamine Salt: Lab Project N. MRID 44021504

Donald DB; Syrgiannis J. 1995. Occurrence of pesticides in prairie lakes in Saskatchewan in relation to drought and salinity. J Environ Qual. 24 (2):266-270.

Dow Chemical Co. 1970. Comparison of Acute Oral Toxicities of Tordon Herbicide Formulations in Small Animals.. (Unpublished study including letter dated Mar 3, 1970 from M.L. Leng to Harold G. Alford, received Mar 6, 1970 under 0F0863; CDL: 093160-B) MRID 00045404

Dow Chemical Co. 1985. A Mortality and Cancer Incidence Survey of Employees Assigned to Tordon Operations with Cover Letters and Dated 02/12/87 and 02/23/87. EPA/OTS Document No. 86-870000229. OTS0513179.

Dow Chemical Co. No Year. An Aquatic Hazard Assessment: Picloram. EPP/OTS Document No. 40-84421611. OTS0519384.

DowElanco. 1996. Submission of Hazard to Wildlife and Aquatic Organisms, Hazard Evaluation: Non-Target Plants and Exposure/Risk to Non Target Organisms Data in Support of the Product Specific RED for Tordon 22K and Tordon 101M Containing Picloram. Transmittal of 5 Studies. MRID 43959500.

Durkin PR; Diamond G. 2002. Neurotoxicity, Immunotoxicity, and Endocrine Disruption with Specific Commentary on Glyphosate, Triclopyr, and Hexazinone: Final Report. SERA TR 01-43-08-04a dated January 30, 2002. Available at www.fs.fed.us/foresthealth/pesticide/risk.htm.

Durkin PR; Rubin L; Withey J; Meylan W. 1995. Methods of assessing dermal absorption with emphasis on uptake from contaminated vegetation. Toxicol Indust Health. 11 (1):63-79.

Durkin P; Rubin L; Withey J; Meylan W; Nakatsugawa T. 1998. A reevaluation of methods for assessing dermal absorption. Draft dated May 18, 1998. Available from: <u>http://www.sera-inc.com.</u>

Ecobichon DJ. 1998. Occupational Hazards of Pesticide Exposure – Sampling, Monitoring, Measuring. Taylor & Francis, Philadelphia, PA. 251 pp.

ExToxNet (Extension Toxicology Network). 1996a. Pesticide Information Profile on Hexachlorobenzene. Extension Toxicology Network: A Pesticide Information Project of Cooperative Extension Offices of Cornell University, Michigan State University, Oregon State University, and University of California at Davis. Revised June, 1996. 5 p. http://ace.orst.edu/info/extoxnet/pips/hexachlo.htm.

ExToxNet (Extension Toxicology Network). 1996b. Picloram Pesticide Information Profile. Extension Toxicology Network: A Pesticide Information Project of Cooperative Extension Offices of Cornell University, Michigan State University, Oregon State University, and University of California at Davis. Revised June, 1996. 5 p. <u>http://ace.orst.edu/cgi-bin/mfs/01/pips/picloram.htm</u>.

Finney DJ. 1971. Probit Analysis. New York: Cambridge University Press. 333 p.

Fletcher JS; Nellessen JE; Pfleeger TG. 1994. Literature review and evaluation of the EPA foodchain (Kenega) nomogram, an instrument for estimating pesticide residues on plants. Environ. Toxicol. Chem. 13(9):1383-1391.

Forsyth DJ; Martin PA; Shaw GG. 1997. Effects of herbicides on two submersed aquatic macrophytes, Potamogeton pectinatus L. and Myriophyllum sibiricum Komarov, in a prairie wetland. Environ Pollut. 95 (2):259-268.

Fuerst EP; Sterling TM; Norman MA; Prather TS; Irzyk GP; Wu Y; Lownds NK; Callihan RH. 1996. Physiological characterization of picloram resistance in yellow starthistle. Pestic Biochem Physiol. 56 :149-161.

Gaines TB. 1969. Acute toxicity of pesticides. Toxicol Appl Pharamacol. 14:515-534.

Gallina MA; Stephenson GR. 1992. Dissipation of (14C)glufosinate ammonium in two Ontario soils. J Agric Food Chem. 40 (1):165-168.

Gardner GR; Yevich PP; Hurst J; Thayer P; Benyi S; Harshbarger JC; Pruell RJ. 1991. Germinomas and teratoid siphon anomalies in softshell clams, *Mya arenaria*, environmentally exposed to herbicides. Environ Health Perspect. 90 :43-51.

Gersich FM; Hopkins DL; Milazzo DP. 1985. Acute and chronic toxicity of technical picloram (4-amino-3,5,6- trichloropicolinic acid) to *Daphnia magna* Straus. Bull Environ Contam Toxicol. 35 (1):121-126.

Gilbert K. 1996a. Tordon 101M: Acute Dermal Toxicity Study in New Zealand White Rabbits: Lab Project Number: M-003779-007D. Unpublished study prepared by The Dow Chemical Co. 29 p. MRID 43959602.

Gilbert K. 1996b. Tordon 101M: Dermal Sensitization Potential in the Hartley Albino Guinea Pig: Lab Project Number: M-003779-007E: GPIGS07/11/95. Unpublished study prepared by The Dow Chemical Co. 18 p. MRID 43959608.

Gilbert K. 1996c. Tordon 22K: Acute Dermal Toxicity Study in New Zealand White Rabbits: Lab Project Number: M-004713-005. Unpublished study prepared by The Dow Chemical Co. 28 p. MRID 43959603.

Goldstein RS; Schnellmann RG. 1996 Toxic Responses of the Kidney. In: Cassarett and Doull's Toxicology: The Basic Science of Poisons. 5th ed., CD Klaassen, MO Amdur, J Doull, Eds. McGraw-Hill, New York, NY. pp. 417-442.

Goldstein A; Aronow L; Kaman SM. 1974. Principles of Drug Action: The Basis of Pharmacology. 2nd ed. John Wiley and Sons, New York, NY. 854 p.

Gomez de Barreda D; Carbonell EA; Cases B; Munoz N. 1993. Use of tomato *(Lycopersicon esculentum)* seedlings to detect bensulfuron and quinclorac residues in water. Weed Technol. 7 (2):376-381.

Grossmann K; Kwiatkowski J; Tresch S. 2001. Auxin herbicides induce H(2)O(2) overproduction and tissue damage in cleavers (Galium aparine L.). J Exp Bot. 52 (362):1811-1816.

Grover R; Gear JR; Zerr R. 1975. Relative phytotoxicity of some picloram derivatives. Bull Environ Contam Toxicol. 14 (6):721-725.

Hansen H; Grossmann K. 2000. Auxin-induced ethylene triggers abscisic acid biosynthesis and growth inhibition. Plant Physiol. 124 (3):1437-1448.

Harrington TB; Minogue PJ; Lauer DK; Ezell AW. 1998. Two-year development of southern pine seedlings and associated vegetation following spray-and-burn site preparation with imazapyr alone or in mixture with other herbicides. New Forests. 15 (1):89-106.

Harris SA; Solomon KR. 1992. Human exposure to 2,4-D following controlled activities on recently sprayed turf. J Environ Sci Health. B27(1): 9-22.

Haut K; Bell T. 1997. Tordon 22K: Dermal Sensitization Potential in Hartley Albino Guinea Pigs. (Final Report): Lab Project Number: 971122: T1.05-003-000-025. Unpublished study prepared by The Dow Chemical Co. 37 p. MRID 44389101.

Havens P; Peacock A. 1995. Picloram: Refinement of Ecological Exposure Estimates Due to Drift, Runoff, and Leaching: Lab Project Number: GH-C 3726. Unpublished study prepared by DowElanco North American Environmental Chemistry Lab. 107 p. MRID 43959501

Havens P; Murrow D; Pottinger M. 2001. State Groundwater Surveillance Monitoring Program for Picloram Herbicide: Field Progress Report: Lab Project Number: 980054: 96-330: ENV960714. Unpublished study prepared by Stone Environmental, Inc. and Agvise Laboratories. 861 p. MRID 45400701

Hayes JR; Condie LW; Borzelleca JF. 1986. Acute 14-day repeated dosing, and 90-day subchronic toxicity studies of potassium picloram. Fund Appl Toxicol 7:464-470.

Heitmuller T. 1975. Acute Toxicity of Tordon 22K to Larvae of the Eastern Oyster (Crassostrea virginica), Pink Shrimp (Penaeus duorarum), and Fiddler Crabs (Uca pugilator).. (Unpublished study received Jun 24, 1983 under 464-502; prepared by Bionomics- EG & G, Inc., submitted by Dow Chemical U.S.A., Midland, MI; CDL:250605-L) MRID 00129073

Hoerger F; Kenaga EE. 1972. Pesticide residues on plants: Correlation of representative data as a basis for estimation of their magnitude in the environment. In: Environmental Quality and Safety, Volume I: Global Aspects of Toxicology and Technology as Applied to the Environment. F Coulston and F Kerte (eds.). Academic Press, New York, NY. pp. 9-28.

IARC (International Agency for Research on Cancer). 1991. IARC Monographs on the Evaluation of Carcinogenic Risks to Humans. Occupational Exposures in Insecticide Application, and Some Pesticides. Vol. 53. Lyon, France. pp. 481-493.

Ismail BS; Kalithasan K. 1997. Effects of repeated application on persistence and downward movement of four herbicides in soil. Aust J Soil Res. 35 (3):503-513.

Jacobs JS; Sheley RL; Story JM. 2000. Use of picloram to enhance establishment of Cyphocleonus achates (Coleoptera: Curculionidae). Environ Entomol. 29 (2):349-354.

Jacoby PW; Meadors CH; Clark LE. 1990. Effects of triclopyr, clopyralid, and picloram on growth and production of cotton. J Prod Agric. 3 (3):297-299.

Jeffrey M. 1987a. Picloram Acid (Picloram Technical): Acute Oral Toxicity Study in Fischer 344 Rats: Final Report.: K-038323-042A. Unpublished study prepared by The Dow Chemical Co., Mammalian and Environmental Toxicology Research Laboratory. 26 p. MRID 40479413.

Jeffrey M. 1987b. Picloram Acid (Picloram Technical): Dermal Sensitization Potential in the Hartley Albino Guinea Pig: Final Report: K-038323-042E. Unpublished study prepared by The Dow Chemical Co., Mammalian and Environmental Toxicology Research Laboratory. 13 p. MRID 40479418.

Jeffrey M. 1987c. Picloram Acid (Picloram Technical): Primary Dermal Irritation Study in New Zealand White Rabbits: Final Report: K-038323-042B. Unpublished study prepared by The Dow Chemical Co., Mammalian and Environmental Toxicology Research Laboratory. 10 p. MRID 40479417.

Jeffrey M. 1987d. Picloram Acid (Picloram Technical): Primary Eye Irritation Study in New Zealand White Rabbits: Final Report: K- 038323-042C. Unpublished study prepared by The Dow Chemical Co., Mammalian and Environmental Toxicology Research Laboratory. 11 p. MRID 40479416.

Jeffrey M; Battjes J; Yano B. 1987. Tordon K Salt Liquor: Acute Oral Toxicity Study in Fischer 344 Rats: Final Report: K 050731-004A. Unpublished study prepared by The Dow Chemical Co. Mammalian and Environmental Toxicology Research Laboratory. 26 p. MRID 40479401

Jeffrey M; Phillips J; Lomax L. 1987. Tordon K Salt Liquor: Acute Dermal Toxicity Study in New Zealand White Rabbits: Final Rept.: K-050731-004D. Unpublished study prepared by The Dow Chemical Co., Mammalian and Environmental Toxicology Research Laboratory. 18 p.. MRID 40479402

Jeffrey M; Battjes J; Yano B. 1987. Tordon 22K Weed Killer: Acute Oral Toxicity Study in Fischer 344 Rats: Project ID: M-004713-002A. Unpublished study prepared by Dow Chemical Co. 16 p. MRID 40677401

Jeffrey M; Battjes J; Johnson K. 1987. Tordon 22K Weed Kill er: Acute Dermal Toxicity Study in New Zealand White Rabbits: Project ID: M-004713-002D. Unpublished study prepared by Dow Chemical Co. 18 p. MRID 40677402

John J; Ouellette J; Johnson K. 1984. Picloram Potassium Salt: Oral Teratology study in Rabbits: HET K-050731-003. (Un published study received Jan 30, 1984 under 464-502; submitted by Dow Chemical U.S.A., Midland, MI; CDL:252493-A) MRID 00138703. Summarized in U.S. EPA 1995b.

Keshava N; Ong T-M. 1999. Occupational exposure to genotoxic agents. Mutat Res. 437 (2):175-194.

Keys B. 1992. Goin' fishin with Tordon 22K: Roadside spraying leaves 15,000 pounds of trout dead. JPR 12 (2):18-19.

Kirk H; Milazzo D; Humbert L; et al. 1994. The Toxicity of Tordon K Herbicide Formulation to the Aquatic Plant, Duckweed, Lemna gibba L.G-3: Lab Project Number: DECO/ES/2764. Unpublished study prepared by The Environmental Toxicology & Chemistry Research Lab. 35 p. MRID 43230311.

Knisel WG; Davis FM. 2000. GLEAMS (Groundwater Loading Effects of Agricultural Management Systems), Version 3.0, User Manual. U.S. Department of Agriculture, Agricultural Research Service, Southeast Watershed Research Laboratory, Tifton, GA. Pub. No.: SEWRL-WGK/FMD-050199. Report Dated May 1, 1999 and revised August 15, 2000. 194pp.

Koizumi A. 1991. Experimental evidence for the possible exposure of workers to hexachlorobenzene by skin contamination. Br J Indus Med. 48 :622-628.

Kookana RS; Baskaran S; Naidu R. 1998. Pesticide fate and behaviour in Australian soils in relation to contamination and management of soil and water: A review. Aust J Soil Res. 36 (5):715-764.

Krzyszowska AJ; Allen RD; Vance GF. 1994. Assessment of the fate of two herbicides in a Wyoming rangeland soil: column studies. J Environ Qual. 23 (5):1051-1058.

Krzywszowska-Waitkus AJ; Allen RD; Vance GF; Zhang R; Legg DE. 1999. A field lysimeter study to evaluate herbicide transport in a Wyoming irrigated pasture. Commun Soil Sci Plant Anal. 30 (1-2):245-263.

Landry T; Johnson K; Cieszlak F; et al. 1986. Picloram: A Two-year Dietary Chronic Toxicity-oncogenicity Study in Fischer 344 Rats. Unpublished study prepared by Dow Chemical U.S.A. 243 p. MRID 00132705. Summarized in U.S. EPA 1992a, 1994, and 1999.

Lanman B. 1996a. Series 61: Product Identity and Composition of Forestry Tordon 101 Mixture and Grazon P+D; Water Soluble Concentrate Formulations Containing Picloram Triisopropanolamine Salt and 2,4-D Triisopropanolamine Salt: Lab Project Number: GH-C 3519. Unpublished. MRID 44021501

Lanman B. 1996b. Series 61: Product Identity and Composition of Tordon 22K, Tordon K, and Grazon PC; Water Soluble Concentrate Formulations Containing Picloram Potassium Salt: Lab Project Number: GH-C 3520. Unpublished study prepared by DowElanco. 56 p.. MRID 44021502

Lanman B. 1996c. Series 61: Product Identity and Composition of Tordon RTU, Forestry Tordon 101R, and Pathway; Water Soluble Concentrate Formulations Containing Picloram Triisopropanolamine Salt and 2,4-D Triisopropanolamine Salt: Lab Project Number: GH-C 4025. Unpublish. MRID 44021503

Lavy TL; Norris LA; Mattice JD; Marx DB. 1987. Exposure of forestry ground workers to 2,4-D, picloram and dichlorprop. Environ Toxicol Chem. 6 (3):209-224.

Levine, TE. 1996. The regulation of inert ingredients in the United States. Chapter 1 in Pesticide Formulation and Adjuvant Technology. CL Foy and DW Pritchard (eds). CRC Press. Boca Raton, FL. pp. 1-11.

Libich S; To JC; Frank R; Sirons GJ. 1984. Occupational exposure of herbicide applications to herbicides used along electric power transmission line right-of-way. J Am Indus Hyg Assoc. 45 (1):56-62.

Linscombe V; Gollapudi B. 1996. Picloram: Evaluation in the Chinese Hamster Ovary Cell/Hypoxanthine-Guanine Phosphoribosyl Transferase (CHO/HGPRT) Forward Mutation Assay: Lab Project Number: K-038323-049. Unpublished study prepared by The Dow Chemical Co. 25 p. MRID 44164801.

Lorz H; Glenn S; Williams R; et al. 1979. Effects of Selected Herbicides on Smolting of Coho Salmon. By Oregon, Departnent of Fish and Wildlife, Research and Development Section and U.S. Forest Service, Pacific Northwest Forest and Range Experiment Station. Corvallis, OR: US EPA. (EPA-600/3-79-071; Grant #R 804283; pages i,iv-x,1,6-14,40-50,83-85,92 only. Also in unpublished submission received June 24, 1983 under 464-502; submitted by Dow Chemical U.S.A., Midland, MI; CDL:250605-N. MRID 00129075.

Mabury SA; Crosby DG. 1996. Pesticide reactivity toward hydroxyl and its relationship to field persistence. J Agric Food Chem. 44 :1920-1924.

Mac MJ; Seelye JG. 1981. Potential influence of acetone in aquatic bioassays testing the dynamics of and effects of PCBs. Bull Environ Contam Toxicol. 27: 359-367.

Mayes MA; Dill DC. 1984. The acute toxicity of picloram, picloram potassium salt, and picloram triisopropanolamine salt to aquatic organisms. Environ Toxicol Chem. 3 (4):263-269.

Mayes MA; Oliver GR. 1985. An Aquatic Hazard Assessment: Picloram. In: Bahner & Hansen. Aquatic Toxicology and Hazard Assessment. 8th Symposium. ASTM STP 891. American Society of Testing Materials. Philadelphia, PA. pp. 253-269.

Mayes M; Dill D; Hopkins D. 1984. The Toxicity of Technical Picloram to the Embryo, Larval, and Juvenile Stages of the Rainbow Trout. (Salmo gairdneri Richardson): ES-703. Unpublished study prepared by Dow Chemical USA. 17 p. MRID 00151784.

Mayes MA; Hopkins DL; Dill DC. 1987. Toxicity of picloram (4-amino-3,5,6-trichloropicolinic acid) to life stages of the rainbow trout. Bull Environ Comtam Toxicol 38:653-660.

McGuirk R; Cieszlak F. 1996. Tordon 22K Weed Killer: Acute Aerosol Inhalation Toxicity Study with Fischer 344 Rats: Lab Project Number: M-004713-004. Unpublished study prepared by The Dow Chemical Co. 38 p. MRID 43959605.
McMaster S. 1999. Regulatory Manager, Dow AgroSciences LLC, 9330 Zionsville Road, Indianapolis, IN. 46268-1054. Phone: (317) 337-4670. Personal communication with P. Durkin, SERA Inc. (Cited in SERA 1999)

Michael JL; Neary DG. 1993. Herbicide dissipation studies in southern forest ecosystems. Environ Toxicol Chem. 12 (3):405-410.

Michael JL; Fowler WP; Gibbs HL; Fischer JB. 1994. Water Chemistry of Ephemeral Streams. Symposium on Ecosystem Management Research in the Ouachita Mountains: Pretreatment Conditions and Preliminary Findings, Hot Springs, AZ, October 26-27, 1993. pp. 186-190.

Micro Flo Company. 2002. Submission of Product Chemistry Data in Support of the Application for Registration of Picloram Acid Technical. Transmittal of 3 Studies.. MRID 45645500

Milazzo D; Kirk H; Humbert L; et al. 1994a. The Toxicity of Access Herbicide Formulation to the Aquatic Plant, Duckweed, *Lemna gibba* L.G-3: Lab Project Number: DECO/ES/2762. Unpublished study prepared by The Environmental Toxicology & Chemistry Research Lab. 33 p. MRID 43230310.

Milazzo D; Kirk H; Humbert L; et al. 1994b. The Toxicity of Tordon 101 Herbicide Formulation to the Aquatic Plant Duckweed, *Lemna gibba* L.G-3: Lab Project Number: DECO-ES-2763. Unpublished study prepared by The Environmental Toxicology & Chemistry Research Lab. 35 p. MRID 43230312.

Miller K; Meredith C. 2000. Immunotoxicology. In: General and Applied Toxicology, 2nd edition. B Ballantyne; TC Marrs; T Syversen (eds). Grove's Dictionaries Inc., New York, NY. pp. 997-1015.

Miller JH; Boyd RS; Edwards MB. 1999. Floristic diversity, stand structure, and composition 11 years after herbicide site preparation. Can J Forest Res. 29 (7):1073-1083.

Mizzell M; Atkin L; Haut K; et al. 1990. Tordon RTU: 21-Day Dermal Toxicity Study in New Zealand White Rabbits: Lab Project Number: M-005179-003. Unpublished study prepared by the Dow Chemical Co. 166 p. MRID 4395610.

Mohammed KB; Ma TH. 1999. Tradescantia-micronucleus and -stamen hair mutation assays on genotoxicity of the gaseous and liquid forms of pesticides. Mutat Res. 426 (2):193-199.

Moore JA. 1964. Physiology of the Amphibia. Academic Press, New York. 654 p. (Cited in USDA 1993)

Moslen MT. 1996. Toxic Responses of the Liver. In: Casarett and Doull's Toxicology: The Basic Science of Poisons. 5th Edition. McGraw-Hill, Health Professions Division, New York, NY. pp. 403-415.

Moyer J. 2002. Picloram Acid Technical: Product Identity and Composition, Description of Materials Used to Produce the Product, Description of Production Process, Discussion of Formation of Impurities and Certified Limits: Lab Project Number: MFC2002-02. Unpublished st. MRID 45658501

Murphy G. 1993. Response to Letter Written by Walter I. Waldrop (Received 12/3/93): Subject: Picloram Registration Standard: Lab Project Number: GM122293A. Unpublished study prepared by DowElanco, Analytical and Product Chemistry. 8 p. MRID 43065001.

NCI (National Cancer Institute). 1978. Bioassay of Picloram for Possible Carcinogenicity. Technical Report Series No. 23. National Institute of Health, Bethesda, MD. NTIS# PB 276 471. pp. 1-93.

Neary DG; Bush PB; Michael JL. 1993. Fate, dissipation and environmental effects of pesticides in southern forests: A review of a decade of research progress. Environ Toxicol Chem. 12 (3):411-428.

Nolan RJ; Freshour NL; Kastl PE; Saunders JH. 1984. Pharmacokinetics of picloram in male volunteers. Toxicol Appl Pharmacol. 76 (2):264-269.

Nolte KR; Fulbright TE. 1997. Plant, small mammal, and avian diversity following control of honey mesquite. J Range Manage. 50 (2):205-212.

NRC (National Research Council). 1983. Risk Assessment in the Federal Government: Managing the Process. Washington, DC: National Academy Press. 176 p. + App.

NTP (National Toxicology Program). 1978. NTP (NCI) Technical Report on the Bioassay of Picloram for Possible Carcinogenicity (CAS No. 1918-02-1).TR-23. NTIS# PB276471/AS. Online at <u>http://ntp-server.niehs.nih.gov/htdocs/LT-Studies/TR023.html</u>.

O'Flaherty EJ. 1981. Toxicants and Drugs: Kinetics and Dynamics. John Wiley and Sons, New York, NY. 398 p.

Oakes DJ; Pollak JK. 1999. Effects of a herbicide formulation, Tordon 75D, and its individual components on the oxidative functions of mitochondria. Toxicology. 136 (1):41-52.

Oakes DJ; Webster WS; Brown-Woodman PD; Ritchie HE. 2002a. A study of the potential for a herbicide formulation containing 2,4-D and picloram to cause male-mediated developmental toxicity in rats. Toxicol Sci. 68 (1):200-206.

Oakes DJ; Webster WS; Brown-Woodman PD; Ritchie HE. 2002b. Testicular changes induced by chronic exposure to the herbicide formulation, Tordon 75D((R)) (2,4-dichlorophenoxyacetic acid and picloram) in rats. Reprod Toxicol. 16 (3):281-289.

Pang L; Close ME. 1999. Attenuation and transport of atrazine and picloram in an alluvial gravel aquifer: A tracer test and batch study. N Z J Mar Freshwater Res. 33 (2):279-291.

Peterson HG; Boutin C; Martin PA; Freemark KE; Ruecker NJ; Moody MJ. 1994. Aquatic phytotoxicity of 23 pesticides applied at expected environmental concentrations. Aquat Toxicol. 28 (3/4):275-292.

Prado AG; Airoldi C. 2001. Toxic effect caused on microflora of soil by pesticide picloram application. J Environ Monit. 3 (4):394-397.

Saito H; Nakano M. 2001. Partial synchronization of cell division and micronucleation in suspension-cultured cells of Hemerocallis hybrida: The effects of hydroxyurea and various spindle toxins. Breeding Sci. 51 (4):285-291.

Schroeder R. 1990. A Teratogenicity Study in Rats with Picloram K Salt: Final Report: Lab Project Number: 89-3459. Unpublished study prepared by Bio/dynamics, Inc. 434 p. MRID 41382502. Summarized in U.S. EPA 1995b.

Schulz CO; LaGoy PK; Paxton MB. 1986. Reviewing the literature on the health effects of phenoxy herbicides and associated dioxins. Chemosphere. 15 (9-12):2099-2102.

Schuytema GS; Nebeker AV; Griffis WL. 1994. Effects of dietary exposure to forest pesticides on the brown garden snail Helix aspersa Muller. Arch Environ Contam Toxicol. 26 (1):23-28.

Schwab D. 1995. Evaluating the Effects of Picloram (K-Salt) on the Seedling Emergence of Non-Target Terrestrial Plants: Lab Project Number: 42488: RES95069. Unpublished study prepared by ABC Labs, Inc. 141 p. MRID 43959505.

Schwab D. 1996. Evaluating the Effects of Picloram (K-Salt) on the Vegetative Vigor of Non-Target Terrestrial Plants: Lab Project Number: 43214: RES96067. Unpublished study prepared by ABC Labs., Inc. 156 p. MRID 44156701.

Scifres CJ; McCall HG; Mazey R; Tai H. 1977. Residual properties of 2,4,5-T and picloram in sandy rangeland soils. J Environ Qual. 6 (1):36-42. (Summarized in U.S. EPA 1992b)

SERA (Syracuse Environmental Research Associates, Inc.). 1998. Reevaluation of Methods for Assessing Dermal Exposure, SERA TR 98-21-08-01d1, dated May 18, 1998. Prepared under USDA/FS Contract 53-3187-5-12, Order No. 43-3187-7-0240. Syracuse Environmental Research Associates, Inc., Fayetteville, NY.

SERA (Syracuse Environmental Research Associates, Inc.). 1998. Methodology for Human Health Risk Assessments, SERA TR 96-21-07-03e, draft dated March 26, 1998. Prepared under USDA/FS Contract 53-3187-5-12, Order No. 43-3187-7-0407. Syracuse Environmental Research Associates, Inc., Fayetteville, NY.

SERA (Syracuse Environmental Research Associates, Inc.). 1999. Picloram (Tordon K and Tordon 22K) - Final Report, SERA TR 99-21-15- 01e, July, 1999. Prepared under USDA/FS Contract No. 53-3187-5-12. Order No. 43-3187-8-0502. Syracuse Environmental Research Associates, Inc., Fayetteville, NY.

SERA (Syracuse Environmental Research Associates, Inc.). 2001. Preparation of Environmental Documentation and Risk Assessments, SERA MD 2001-01a, draft dated July 2001. Syracuse Environmental Research Associates, Inc., Fayetteville, NY. Available at www.sera-inc.com

SERA (Syracuse Environmental Research Associates, Inc.). 2003. Documentation for the Use of GLEAMS (Version 3) and Auxiliary Programs in Forest Service Risk Assessments (Version 2.01), SERA TD 2003-02e. Syracuse Environmental Research Associates, Inc., Fayetteville, NY. Available at <u>www.sera-inc.com</u>.

Sheets TJ; Harrell BJ. 1986. Effects of low levels of soil-applied picloram on flue-cured tobacco. Tech. Bull. 280. Raleigh, NC: North Carolina State University. North Carolina Agricultural Research Service.

Somers JD; Moran ET Jr; Reinhart BS. 1978. Reproductive success of hens and cockerels originating from eggs sprayed with 2,4-D, 2,4-5-T and picloram followed by early performance of their progency after a comparable in ovo exposure. Bull Environ Contam Toxicol. 20 (1):111-119.

Sonnenschein C; Soto AM. 1998. An updated review of environmental estrogen and androgen mimics and antagonists. J Steroid Biochem Mol Biol. 65 (1-6):143-150.

Sorensen PB; Mogensen BB; Gyldenkaerne S; Rasmussen AG. 1998. Pesticide leaching assessment method for ranking both single substances and scenarios of multiple substance use. Chemosphere. 36 (10):2251-2276.

Sparkes EC; Grace S; Panetta FD. 2002. The effects of various herbicides on Bryophyllum pinnatum (Lam.) Pers in Nudgee Wetlands Reserve, Queensland. Plant Prot Q. 17 (2):77-80.

SRC (Syracuse Research Corporation). 1998. Estimation Program for Log K_{o/w}. <u>http://www.syrres.com/</u>.

St-Laurent D; Blaise C; MacQuarrie P; Scroggins R; Trottier B. 1992. Comparative assessment of herbicide phytotoxicity to Selenastrum capricornutum using microplate and flask bioassay procedures. Environ Toxicol Water Qual. 7 (1):35-48.

Stevenson GT. 1965. A Gamebird Toxicology Study--Acute Dietary Feeding of Tordon^(R) to Wild Type Mallard Ducklings.. (Unpub lished study received Nov 21, 1967 under 8F0660; submitted by Dow Chemical Co., Indianapolis, Ind.; CDL:091155-H) MRID 00075781

Streeter C; Battjes J; Johnson K. 1987. Picloram Acid: An Acute Aerosol Inhalation Study in Fischer 344 Rats: Final Report.: K-038323-043. Unpublished study prepared by The Dow Chemical Co., Mammalian and Environmental Toxicology Research Laboratory. 21 p. MRID 40479415.

Streeter C; Battjes J; Yano B. 1987. Tordon K Salt Liquor: An Acute Aerosol Inhalation Study in Fischer 344 Rats: Final Rept.: K-050731-005. Unpublished study prepared by The Dow Chemical Co., Mammalian and Environmental Toxicology Research Laboratory. 20 p. MRID 40479403

Streeter C; Eisenbrandt D; Battjes J. 1988. Tordon 22K Weed Killer: An Acute Aerosol Inhalation Study in Fischer 344 Rats: Project ID: M-004713-003. Unpublished study prepared by Dow Chemical Co. 20 p. MRID 40677403

Strek G; Spaan WP. 1997. Wind erosion control with crop residues in the Sahel. Soil Sci Soc Am J. 61 (3):911-917.

Strek G; Stein A. 1997. Mapping wind-blown mass transport by modeling variability in space and time. Soil Sci Soc Am J. 61 (1):232-239.

Sun Y; Pignatello JJ. 1993. Activation of hydrogen peroxide by iron(III) chelates for abotic degradation of herbicides and insecticides in water. J Agric Food Chem. 41 (2):308-312.

Tan LK; Humphries D; Yeung PYP; Florence LZ. 1996. Determinations of clopyralid, picloram, and silvex at low concentration in soils by calcium hydroxide-water extraction and gas chromatography measurement. J Agric Food Chem. 44 (4):1135-1143.

Teske ME; Curbishley TB. 1990. Forest Service Aerial Spray Computer Model, FSCBF 4.0, User Manual. Continuum Dynamics, Inc, Princeton, NJ. CDI Report No. 90-06.

Thomas JA; Thomas MJ. 2000. Endocrine Toxicology. In: General and Applied Toxicology, 2nd edition. B Ballantyne; TC Marrs; T Syversen (eds). Grove's Dictionaries Inc., New York, NY. pp. 979-995.

Thompson OJ; Emerson JL; Strebing RJ; Gerbig CG; Robinson VB. 1972. Teratology and postnatal studies on 4-amino-3,5,6-trichloropicolinic acid (picloram) in the rat. Food Cosmet Toxicol. 10 :797-803.

Tomkins DJ; Grant WF. 1976. Monitoring natural vegetation for hebicide-induced chromosome aberrations. Mutat Res. 36:73-84.

Tout NL; Yau KY; Trevors JT; Lee H; Hall JC. 2001. Synthesis of ligand-specific phage-display ScFv against the herbicide picloram by direct cloning from hyperimmunized mouse. J Agric Food Chem. 49 (8):3628-3637.

Trevors JT. 1998. Bacteria biodiversity in soil with an emphasis on chemically-contaminated soils. Water Air Soil Pollut. 101 (1-4):45-67.

Tu CM. 1994. Effects of herbicides and fumigants on microbial activities in soil. Bull Environ Contam Toxicol. 53 (1):12-17.

U.S. EPA (U.S. Environmental Protection Agency). 1989. Recommendations for and Documentation of Biological Values for use in Risk Assessment. U.S. EPA, Environmental Criteria and Assessment Office, Office of Health and Environmental Assessment, Cincinnati, OH. ECAO-CIN-554.

U.S. EPA (U.S. Environmental Protection Agency). 1992a. Dermal Exposure Assessment: Principles and Applications. EPA/600/8-91/011B. Interim Report. Exposure Assessment Group, Office of Health and Environmental Assessment, Washington, DC.

U.S. EPA (United States Environmental Protection Agency). 1992b. Drinking water criteria document for picloram. United States Environmental Protection Agency. Health and Ecological Criteria Division.

U.S. EPA (United States Environmental Protection Agency). 1992c. Integrated Risk Information System, Picloram. RfD revised 5/1/92. http://www.epa.gov/ngispgm3/iris/subst/0256.htm.

U.S. EPA (United States Environmental Protection Agency). 1994. RfD Peer Review Report of Picloram. United States Environmental Protection Agency, Office of Prevention, Pesticides, and Toxic Substances, memo from George Ghali to Robert Taylor dated April 1, 1994.

U.S. EPA (United States Environmental Protection Agency). 1995a. R.E.D. Facts: Picloram. Prevention, Pesticides, and Toxic Substances, EPA 738–F-95-018. August 1995.

U.S. EPA (United States Environmental Protection Agency). 1995b. Reregistration eligibility decision (RED) for picloram. EPA 738-R95-019. Office of Prevention, Pesticides and Toxic Substances. Washington, DC. <u>http://www.epa.gov/oppsrt1/REDs/0096.pdf</u>.

U.S. EPA (U.S. Environmental Protection Agency). 1996. Exposure Factors Handbook. Office of Research and Development, National Center for Environmental Assessment, U.S. EPA, Washington, DC. EPA/600/P-95/002Ba-c. Avail. NTIS: PB97-117683, 97-117691, PB97-117709.

U.S. EPA (United States Environmental Protection Agency). 1997. Hexachlorobenzene: IRIS Entry. Last update 3/1/97. <u>http://www.epa.gov/iris/subst/0374.htm</u>.

U.S. EPA (United States Environmental Protection Agency). 1998a. Drinking Water and Health Technical Fact Sheet on Hexachlorobenzene. Last update 1/27/98. http://www.epa.gov/OGWDW/dwh/t-soc/hcb.html.

U.S. EPA (United States Environmental Protection Agency). 1999. Picloram; Time-Limited Pesticide Tolerances. Federal Register. 64 (2):418-425. <u>http://frwebgate.access.gpo.gov/cgi-bin/getdoc.cgi?dbname=1999_register&docid=98-34830-file d.pdf</u>.

U.S. EPA (United States Environmental Protection Agency). 2002. U.S. National Primary Drinking Water Regulations; Announcement of the Results of EPA's Review of Existing Drinking Water Standards and Request for Public Comment. Federal Register. 67 (74):19030-19090.

U.S. EPA (U.S. Environmental Protection Agency). 2003. Lists of Inert Pesticide Ingredients. <u>Http://www.epa.gov/opprd001/inerts/lists.html</u>.

U.S. EPA/OPPTS (U.S. Environmental Protection Agency/Office of Prevention, Pesticides, and Toxic Substances). 2003. Harmonized Test Guidelines. Available at www.epa.gov/OPPTS_Harmonized/.

U.S. EPA/ORD (U.S. Environmental Protection Agency/Office of Research and Development). 1993. Wildlife Exposure Factors Handbook. Volumes 1 and 2. EPA/600/R-93/187a,b. Pagination not continuous. Available NTIS: PB94-174778 and PB94-174779.

U.S. Weather Service. 1998. NWS Climate Tables, 1961-1990. http://www.nws.noaa.gov/.

University of Florida. 1999. Grouping of selected pesticides by persistence and sorption in soils. Institute of Food and Agricultural Sciences.

USDA (U.S. Department of Agriculture/Forest Service). 1989a. Draft Environmental Impact Statement: Vegetation Management in the Ozark/Ouachita Mountains, Management Bulletin R8-MB-23, dated June, 1989. 499 p.

USDA (U.S. Department of Agriculture/Forest Service). 1989b. Final Environmental Impact Statement: Vegetation Management in the Appalachian Mountains, Management Bulletin R8-MB-38, dated July, 1989. 1104 p.

USDA (U.S. Department of Agriculture/Forest Service). 1989c. Final Environmental Impact Statement: Vegetation Management in the Coastal Plain/Piedmont, Management Bulletin R8-MB-23, dated January, 1989. 1213 p.

USDA (U.S. Department of Agriculture/Forest Service). 1989d. Pesticide Background Statements: Volume 1. Herbicides. Available from NTIS. PB89-226682.

USDA (U.S. Department of Agriculture/Animal and Plant Health Inspection Service). 1993. Nontarget Risk Assessment for the MEDFLY Cooperative Eradication Program. February 1993.

USDA (U.S. Department of Agriculture/Forest Service). 1995. Picloram: Pesticide Fact Sheet. Prepared by Information Ventures, Inc. under U.S. Forest Service Contract. November 1995.

USDA (U.S. Department of Agriculture). 1998a. Cropland acreage, soil erosion, and installation of conservation buffer strips: preliminary estimates of the 1997 National Resource Inventory.

USDA/APHIS (U.S. Department of Agriculture Animal and Plant Health Inspection Service). 1993. Nontarget Risk Assessment for the MEDFLY Cooperative Eradication Program. USDA Animal and Plant Health Inspection Service. February 1993.

USDA/ARS (U.S. Department of Agriculture Agricultural Research Station). 1995. ARS Pesticide Properties Database. <u>Http://wizard.arsusda.gov/rsml/testfiles.</u> Last updated May 1995

USGS (U.S. Geological Survey). 1998. Data on Pesticides in Surface and Ground Water of the United States., Results of the National Water Quality Assessment Program (NAWQA). Revised Oct. 23, 1998. <u>Http://wwwdwatem.wr.usgs.gov/cppt/pns_data/data.html</u>.

USGS (U.S. Geological Survey). 2000. Data on Pesticides in Surface and Ground Water of the United States., Results of the National Water Quality Assessment Program (NAWQA). Updated July 17, 2000.

Van Beneden RJ. 1993. Implications for the presence of transforming genes in gonadal tumors in two bivalve mollusk species. Cancer Res. 53 :2976-2979.

van Hemmen JJ. 1992. Agricultural pesticide exposure data bases for risk assessment. Rev. Environ Contam Toxicol. 126 :1-85.

Ward T; Kowalski P; Boeri R. 1994a. Access Herbicide: Toxicity to the Freshwater Diatom, Navicula pelliculosa: Lab Project Number: 430/DO: ES/2766. Unpublished study prepared by T.R. Wilbury Laboratories, Inc. 25 p. MRID 43230301.

Ward T; Kowalski P; Boeri R. 1994b. Access Herbicide: Toxicity to the Marine Diatom, Skeletonema costatum: Lab Project Number: 431/DO: ES/2767. Unpublished study prepared by T.R. Wilbury Laboratories, Inc. 25 p. MRID 43230304.

Ward T; Magazu J; Boeri R. 1994c. Tordon 101 Herbicide: Toxicity to the Marine Diatom, Skeletonema costatum: Lab Project Number: 434/DO: ES/2770. Unpublished study prepared by T.R. Wilbury Laboratories, Inc. 25 p. MRID 43230306.

Warisnoicharoen W; Lansley AB; Lawrence MJ. 2003. Toxicological evaluation of mixtures of nonionic surfactants, alone and in combination with oil. J Pharm Sci. 92(4): 859-868.

Washburn K. 2002. Product Chemistry: Technical Grade Product: Picloram Technical: Final Report: Lab Project Number: 6811-01. Unpublished study prepared by StillMeadow, Inc. 31 p.

Waters MD; Stack HF; Jackson MA. 1999. Genetic toxicology data in the evaluation of potential human environmental carcinogens. Mutat Res. 437 (1):21-49.

Watson VJ; Rice PM; Monnig EC. 1989. Environmental fate of picloram used for roadside weed control. J Environ Qual. 18 :198-205.

Webb SR; Hall JC. 1995. Auxinic herbicide-resistant and -susceptible wild mustard (Sinapis arvenis L.) biotypes: Effect of auxinic herbicides on seedling growth and auxin-binding activity. Pest Biochem Physiol. 52:137-148.

Weinberg J; Hugo J; Landre A. 1996. Evaluation of the Toxicity of Picloram Triisopropanolamine (TIPA) Salt to the Early Life-Stages of the Fathead Minnow, Pimephales promelas Rafinesque: Lab Project Number: DECO-ES-3037. Unpublished study prepared by The Dow Chemical Co. 39 p. MRID 43959504.

Welp G; Bruemmer GW. 1999. Effects of organic pollutants on soil microbial activity: The influence of sorption, solubility, and speciation. Ecotoxicol Environ Saf. 43 (1):83-90.

Winegardner DL. 1996. An Introduction to Soils for Environmental Professionals. Lewis Publishers, New York, NY.

Woodburn KB; Fontaine DD; Bjerke EL. 1989. Photolysis of picloram in dilute aqueous solution. Environ Toxicol Chem. 8 :769-775.

Woodburn K; Hansen S; Ball T. 1991. Picloram 2-EthylHexyl Ester: Bioconcentration in Rainbow Trout, Oncorhynchus mykiss: Lab Project Number: ES-DR-0044-1725-1. Unpublished study pre pared by Dow Chemical Co. 71 p. MRID 42121108.

Woodward DF. 1976. Toxicity of the berbicides dinoseb and picloram to cutthroat (*Salmo clarki*) and lake trout (*Salvelinu namaycush*). J Fish Res Board Can 33:1671-1676.

Woodward DF. 1979. Assessing the hazard of picloram to cutthroat trout. J Range Manage 32 (3):230-232.

Woodward DF. 1982. Acute toxicity of mixtures of range management herbicides to cutthroat trout. J Range Manage 35 (40):539-540.

Wright J. 2000. Phytotoxicity and Yield Effects on Soybeans Following a Soil Surface Application of Picloram K Salt: Lab Project Number: RES97047. Unpublished study prepared by Dow AgroSciences LLC. 94 p. MRID 45289601.

Zahm SH; Ward MH. 1998. Pesticides and childhood cancer. Environ Health Perspect. 106 (Suppl 3):893-908.



Figure 2-1. Use of picloram by the USDA Forest Service in various regions of the United States.



Figure 2-2. Agricultural use of picloram in the United States for 1992 (USGS 1998).

Property	Value	Reference
Synonyms	Formulations: Tordon 22K, Tordon K	C&P Press 1998
CAS Number	1918-02-1 (acid) 081510-83-0 (salt)	Budavari et al. 1989 C&P Press 1998
U.S. EPA Registration Number	62719-6	C&P Press 1998
MW C ₆ H ₃ Cl ₃ N ₂ O ₂	241.48 (acid ⁺) 280.6 (potassium salt)	Budavari et al. 1989 U.S. EPA (1995b)
a.i. (K-salt) to a.e. conversion	0.8606	241.48÷280.6
Henry's Law Constant (atm m ³ /mole)	3.3×10 ⁻¹⁰	Mabury and Crosby 1996
pK _a	3.6 2.3 (22°C) 1.9	Budavari et al. 1989 Baker 1989c USDA/ARS 1995
Vapor pressure	6.16×10 ⁻⁷ mm Hg (35°C) 6.0×10 ⁻¹⁰ mm Hg (25°C)	Budavari et al. 1989 Baker 1989c
Water solubility	430 mg/L (acid, pH 2.5) ∞ (salt) 2×10 ⁵ mg/L (salt) 4.3×10 ⁵ mg/L (K salt) 7.2 mg/L (technical in distilled water)	USDA/ARS 1995 C&P Press 1998 Knisel et al. 1992 Neary et al. 1993 Washburn 2002
K _{o/w} (acid)	22.9 84 79 [log K _{o/w} = 1.9] 1.8 0.89 [pH 5-9] {log Ko/w = -0.05}	SRC 1998 Baker 1989c Washburn 2002 Bidlack (1980) USDA/ARS 1995
K _{o/c} (acid, ml/g)	16 16 (2.2 to 92.9) 29 (7-48) 23(14-33) [Silt Loam, 2.9% OM] 47(22-71) [Sandy Loam, 3.3% OM] 29.9(23.7-36.1) [Sandy Loam, 3.3% OM] 45.3(9-82) [Silt Loam, 2.9% OM]	Knisel et al. 1992 Havens et al. 2001 USDA/ARS 1995 Close et al. 1998 Close et al. 1998 Close et al. 1999 Close et al. 1999
Foliar $t_{1/2}$ (field dissipation)	8 days	Knisel and Davis 2000
Water $t_{1/2}$ (field dissipation)	15 days [0.046 day ⁻¹] 14 [0.048 day ⁻¹]	USDA 1989c,d Scifres et al. 1977
Water $t_{1/2}$ (surface water with degradation via photolysis.)	2.6 days	Woodburn et al. 1989
Soil, aerobic $t_{1/2}$ in days	90 (24 to 272) 18 to 300	Havens et al. 2001 USDA/ARS 1995
Soil $t_{1/2}$ (field dissipation)	90 days 131 days 108 (31 to 206) 203 (160-246) [Silt Loam] 244 (181-299) [Sandy Loam]	Knisel et al. 1992 Micheal and Neary 1993 USDA/ARS 1995 Close et al. 1998 Close et al. 1998

Table 2-1. Identification and Physical/Chemical Properties of Picloram and the Potassium Salt of Picloram.

Use Classification	Total	Total	Pounds per acre	Proporti	on of Use
	Pounds	Acres	average	by Pounds	by Acres
Noxious Weed Control	12,865.90	62,897.12	0.20	0.9989	0.9992
Recreation Improvement	3.50	7.00	0.50	0.0003	0.0001
Rights-of-Way	11.10	44.80	0.25	0.0009	0.0007
Grand Total	12,880.50	62,948.92	0.20	1	1

Table2-2: Use of picloram by USDA Forest Service in 2001 by Type of Use (USDA/FS 2002)

Region		Pounds	Acres	lbs/acre	Proportion of Total Pounds	Proportion of Total Acres
Northern (R1)		4450.06	22147.83	0.20	0.345	0.35
Rocky Mountain (R2)		2759.65	15422.42	0.18	0.214	0.25
Southwestern (R3)		5.00	30.00	0.17	0.000	0.00
Intermountain (R4)		4015.10	14776.19	0.27	0.312	0.23
Pacific Southwest (R5)		0.00	0.00	N/A	0.000	0.00
Pacific Northwest (R6)		1565.69	10402.18	0.15	0.122	0.17
Southern (R8)		85.00	170.00	0.50	0.007	0.003
Eastern (R9)		0.0004	0.30	0.001	3e-08	5e-06
	Total	12880.50	62948.92	0.20	1	1

Table 2-3: Use of picloram by USDA Forest Service in 2001 by Region (USDA/FS 2002)

Application Method	Application Rate (kg/ha) ^a	Concentration in water (ppb) ^a	Application Rate (lb/acre) ^b	Water Contamination Rate (ppb per lb/acre applied)
Aerial	5.6	241	5.00	48.25
Ground broadcast	5.6	77	5.00	15.41
Injection	5	10	4.46	2.24
Injection	0.3	6	0.27	22.42
Injection	1.3	21	1.16	18.11
Injection	0.3	10	0.27	37.37
Injection	0.6	4	0.54	7.47

Table 3-1: Concentrations of picloram in streams after the application of known amounts

^a Taken from data from Michael and Neary 1993, Table 3, p. 407. ^b 1 kg/ha = 0.892 lb/acre

^c Concentration in ppb divided by application rate in lb/acre.

Chemical Specific Parameters							
Parameter	Clay	Loam	Sand	Comment/ Reference			
Halftimes (days)							
Aquatic Sediment	2000	2000	2000	Note 1			
Foliar	8	8	8	Knisel and Davis 2000			
Soil	150 days to 1', 320 2000 days	at 0-1", 200 days at 1' t s at 2' to 5' 1) days at 1" o 2', and layer.	Note 2			
Water	14	14	14	Scifres et al. (1977)			
Ko/c	48	29	7	Note 3			
K _d	1.44	0.43	0.021	Note 4			
Water Solubility, mg/L	200000	200000	200000	Knisel and Davis 2000			
Foliar wash-off fraction	0.6	0.6	0.6	Knisel and Davis 2000			

. ... ъ

Table 3-2: Chemical and site parameters used in GLEAMS Modeling for picloram. 10

Note 1 Aquatic sediment halftimes not encountered. Value of 2000 days based on halftimes for deep soil layers (Close et al. 1999, Table 5, p. 70).

Soil halftimes for picloram are complex and depend on application rate and soil depth. Aerobic soil Note 2 halftimes vary with soil concentration: 18 days at 0.0025 ppm ,29 days at 0.025 ppm , 150 at 0.25 ppm, and 300 days at 2.5 ppm (ARS 1995). The effect of soil depth has been demonstrated by Close et al. (1998, 1999), with soil halftimes over a range of 160 to 324 days in the upper soil layer (about 1 foot) and 318 to 2146 days at depths of 1 foot to about 5 feet (Close et al. 1998, Table 4, p. 57). An application rate of 1 lb/acre is equivalent to $11.21 \,\mu g/cm^2$. Assuming a soil incorporation depth of 1 cm and a density of 1 g/cm³, the initial soil concentration would be about 11.2 ppm in the top 1 cm and 0.37 ppm in the top one foot of soil. For the GLEAMS modeling, a soil halftime of 150 days is used for the upper one inch layer, 200 days for the 1" to 1 foot layer and 320 days for the 1' to 2' layer, and 2000 days for the 2' to 5' layer.

Central value and range taken from USDA/ARS (1995) assuming highest value in clay, followed by Note 3 loam and then sand. Value for loam supported by values from Close et al. (1998, 1999).

Note 4 Based on K_{o/c} and estimates of proportion (P)of organic carbon in clay (0.03), loam (0.015), and sand (0.003), where $K_d = K_{o/c} * P$.

Site Parameters

(see SERA 2003, SERA AT 2003-02d dated for details)

- 1 acre pond, 2 meters deep, with a 0.01 sediment fraction. 10 acre square field (660' by 660') Pond with a root zone of 60 inches and four soil layers. Stream Base flow rate of 4,420,000 L/day with a flow velocity of 0.08 m/second or 6912 meters/day.
- Stream width of 2 meters (about 6.6 feet') and depth of about 1 foot. 10 acre square field (660' by 660') with a root zone of 60 inches and four soil layers.

Annual	Clay		Lo	oam	S	Sand	
(inches)	Average	Maximum	Average	Maximum	Average	Maximum	
5	0.00	0.00	0.00	0.00	0.00	0.00	
10	0.00	0.00	0.00	0.00	0.00	0.00	
15	0.10	10.21	0.00	0.00	0.62	19.36	
20	0.18	22.28	0.06	2.47	1.18	16.55	
25	0.24	35.66	0.26	6.81	1.37	26.33	
50	0.44	98.00	0.87	11.57	1.59	48.24	
100	0.57	184.02	0.93	16.45	1.29	67.98	
150	0.57	187.13	0.85	17.86	1.05	74.53	
200	0.54	178.01	0.75	17.64	0.89	72.87	
250	0.51	166.37	0.67	16.92	0.77	75.56	

Table 3-3: Summary of modeled concentrations of picloram in streams (all units are $\mu g/L$ or ppb per lb/acre)

,							
Annual	Clay		Lo	am	Sa	Sand	
(inches)	Average	Maximum	Average	Maximum	Average	Maximum	
5	0.00	0.00	0.00	0.00	0.00	0.00	
10	0.00	0.00	0.00	0.00	0.00	0.00	
15	0.69	6.98	1.0e-05	1.5e-04	3.25	9.43	
20	0.84	14.32	0.15	0.61	4.09	9.96	
25	0.96	21.73	0.45	1.39	3.95	16.97	
50	1.31	59.10	1.01	2.42	3.16	32.04	
100	1.57	125.79	0.90	5.19	2.37	48.15	
150	1.56	139.78	0.75	5.19	1.99	54.74	
200	1.49	141.84	0.64	5.19	1.75	58.48	
250	1.41	139.22	0.56	5.08	1.58	61.62	

Table 3-4: Summary of modeled concentrations of picloram in ponds (all units are $\mu g/L$ or ppb per lb/acre)

Chemical Specific Parameters								
Parame	Parameter Clay Loam Sand Comment/ Reference		Comment/ Reference					
Halftime	es (days)							
Aquat	ic Sediment	2190	2190	2190	Note 1			
Foliar		1	1	1	Note 2			
Soil, u	pper 1 cm	7.1	7.1	7.1	Note 3			
Soil, le	ower layers	1640	1640	1640	Note 3			
Water		1533	1533	1533	Note 4			
Ko/c (m	L/g)	50000	50000	50000	Note 5			
$K_d(mL/g$	g)	1500	750	150	Note 6			
Water Solubility, mg/L		0.006	0.006	0.006	ATSDR 2002			
Foliar washoff fraction0.10.10.1Note 2					Note 2			
Note 1	ATSDR (2002) giv aquatic sediment, th	es reported han e upper range	lftimes for he is used – i.e.,	xachlorobenze 3 years × 365	ene in soil ranging from 3 to 6 years. For 5 days/year = 2190 days			
Note 2	Volatilization will modeling, all hexactime and washoff d	rapidly remove hlorobenzene lo not impact t	e hexachlorob is assumed to he results of t	enzene from p be deposited he modeling.	olant surfaces. For the GLEAMS on soil. Thus, the values for foliar half-			
Note 3	For the top 1 cm, th (2002) gives report layers, the mid-poin	ne halftime is l ed halftimes fo nt, 4.5 years is	based on the s or hexachloro used – 4.5 ye	tudy by Beall benzene in soi ars×365 = 164	(1976), as discussed in the text. ATSDR l ranging from 3 to 6 years. For lower soil 42.5 ≈ 1640 days.			
Note 4	ATSDR (2002) giv 5.7 years. The aver	es reported ha rage, 4.2 years	lftimes for he or about 153	xachlorobenze 3 days, is usec	ene in surface water ranging from 2.7 to 1.			
Note 5	Knisel and Davis (2 462.8 in a soil with	2000) give a K 2.6% OM.	_{o/c} of 50,000.	ARS (1995) 1	reports a Ko/c of 30,649 based on a Kd of			
Note 6	Note 6 Calculated as $K_d = K_{o/c} \times OC$. The $K_{o/c}$ is taken as 50,000 from Knisel and Davis (2000) and the proportion of OC in sand, loam, and clay is estimated as 0.003 for sand, 0.015 for loam, and 0.030 for clay.							
	Site Parameters (see SERA 2003, SERA AT 2003-02d dated for details)							
Pond	1 acre pond,	2 meters deep	, with a 0.01	sediment frac	tion. 10 acre square field (660' by 660')			

Table 3-5: Chemical and site parameters used in GLEAMS Modeling for hexachlorobenzene.

Base flow rate of 4,420,000 L/day with a flow velocity of 0.08 m/second or 6912 meters/day.

Stream width of 2 meters (about 6.6 feet') and depth of about 1 foot. 10 acre square field (660'

with a root zone of 12 inches.

by 660') with a root zone of 12 inches.

Stream

ppin per 10/ u	(10)						
Annual	Annual Clay		Lo	oam	S	Sand	
(inches)	Average	Maximum	Average	Maximum	Average	Maximum	
5	0.027	0.665	0.023	0.588	0.023	0.588	
10	0.030	0.665	0.026	0.588	0.026	0.588	
15	0.030	0.665	0.026	0.588	0.026	0.588	
20	0.030	0.665	0.026	0.588	0.027	0.588	
25	0.030	0.665	0.026	0.588	0.027	0.588	
50	0.028	0.665	0.026	0.588	0.027	0.588	
100	0.023	0.665	0.026	0.588	0.028	0.588	
150	0.019	0.665	0.024	0.588	0.030	0.588	
200	0.012	0.665	0.021	0.588	0.031	0.588	
250	0.007	0.665	0.017	0.588	0.031	0.588	

Table 3-6: Summary of modeled concentrations of hexachlorobenzene in soil (all units are mg/kg or ppm per lb/acre)

- FF F						
Annual	Clay		Lo	oam	Sand	
(inches)	Average	Maximum	Average	Maximum	Average	Maximum
5	0	0	0	0	0	0
10	0	0	0	0	0	0
15	0.0036	0.58	0	0	0	0
20	0.0140	2.29	0	0	0	0
25	0.0294	4.83	0	0	0	0
50	0.162	27.4	0	0	0	0
100	0.470	88.9	0.05	8.77	0	0
150	0.794	165	0.15	26.8	0	2.0e-05
200	0.905	221	0.27	52.0	0	4.0e-05
250	0.943	268	0.38	80.5	1.1e-03	0.18

Table 3-7: Summary of modeled concentrations of hexachlorobenzene in streams (all units are $\mu g/L$ or ppb per lb/acre)

Annual	Clay		L	oam	Sand	
(inches)	Average	Maximum	Average	Maximum	Average	Maximum
5	0	0	0	0	0	0
10	0	0	0	0	0	0
15	0.09	0.15	0	0	0	0
20	0.32	0.50	0	0	0	0
25	0.62	0.94	0	0	0	0
50	2.77	3.62	0	0	0	0
100	6.82	8.52	1.02	1.58	0	0
150	10.9	15.9	2.69	5.38	1.0e-05	2.0e-05
200	12.2	21.8	4.59	11.18	3.0e-05	4.0e-05
250	12.6	27.2	6.26	17.81	2.1e-02	1.0e-01

Table 3-8: Summary of modeled concentrations of hexachlorobenzene in a pond (all units are $\mu g/L$ or ppb per lb/acre)

Species	Endpoint	Value (mg a.e./L)	Form/Vehicle	Reference					
	Acute								
Bluegill Sunfish	Acute LC ₅₀ 's	14.5 to 45	Acid	Mayes and Oliver 1986					
Fathead minnow	Acute LC ₅₀	55.3 (47.2 - 69.6)	Acid	Mayes and Oliver 1986					
Cutthroat trout	Acute LC ₅₀	4.8 (3.8-6.2)	Acid	Mayes and Oliver 1986					
		3.5 (3.4-4.0)	Acid/Acetone	Woodward 1976					
		3.9 (3.2-4.8)	Acid/Acetone	Woodward 1982					
Rainbow trout	Acute LC ₅₀ 's	5.5 to 19.3	Acid	Mayes and Oliver 1986					
Lake trout	Acute LC ₅₀	1.6 (1.2 - 2.0)	Acid/Acetone	Woodward 1976					
Bluegill Sunfish	Acute LC ₅₀ 's	24 to 137	K-salt	Mayes and Oliver 1986					
Cutthroat trout	Acute LC ₅₀	1.5 (0.8-3.0)	K-salt	Mayes and Oliver 1986					
Rainbow trout	Acute LC ₅₀ 's	13 to 48	K-salt	Mayes and Oliver 1986					
		Chronic							
Rainbow trout	Chronic NOEC	0.55	K-salt/None	Mayes et al. 1987					
	Chronic LOEC	0.88	K-salt/None	Mayes et al. 1987					
Fathead minnow	Chronic NOEC	0.71	TIPA salt/None	Weinberg et al. 1996					
	Chronic LOEC	1.2	TIPA salt/None	Weinberg et al. 1996					
Lake trout	Chronic LOEC	0.035	acid/Acetone	Woodward 1976					
Cutthroat trout	Peak NOEC [TWA ¹]	0.29 [0.04]	acid/Acetone	Woodward 1979					
	Peak LOEC [TWA ¹]	0.61 [0.098]	acid/Acetone	Woodward 1979					

Table 4-1: Summar	v of key	v studies	on the	toxicity	of	picloram	to	fish.
	, or no	blaareb				protorum	ιU	mon.

¹ TWA calculated as the sum of exposure concentrations on various days reported in Table 1 of Woodward (1979) divided by the observation period of 25 days.

Annual Rainfall (inches)	Clay		Lo	Dam	Sand		
	Average	Maximum	Average	Maximum	Average	Maximum	
5	1.390	5.46	1.112	4.43	0.909	4.15	
10	0.715	4.61	0.441	3.64	0.426	3.79	
15	0.435	4.08	0.192	3.54	0.089	3.53	
20	0.307	4.00	0.114	3.53	0.048	3.53	
25	0.237	3.99	0.077	3.53	0.034	3.53	
50	0.102	3.99	0.026	3.53	0.018	3.53	
100	0.015	3.99	0.015	3.53	0.014	3.53	
150	0.013	3.99	0.014	3.53	0.013	3.53	
200	0.013	3.99	0.013	3.53	0.012	3.53	
250	0.013	3.99	0.013	3.53	0.012	3.53	

Table 4-2: Summary of modeled concentrations of picloram in soil (all units are mg/kg or ppm per lb/acre)